\begin{table}%t2 non ! %ta.2 !!
\caption{\label{tab_a2}GMCs as tracers of spiral arms. Columns~1 and~2 are the Galactic longitude and latitude; Cols.~3 to~6 list velocity, the luminosity of CO~emission line (in unit of $10^3$~K~km~s$^{-1}$~pc$^2$), distance and mass of molecular cloud, which were all taken from the reference given in Col.~8; Col.~7 is the note for distance ambiguity given in this reference, used as the same convention as in Table~A.1; Cols.~9 and~10 list the distances to the Sun and to the Galactic center, which are estimated by the velocity and the rotation curve with $R_0=8.5$~kpc
and $\Theta_0=220$~km~s$^{-1}$ when the stellar distance is unavailable; Col.~11 gives the mass of molecular clouds re-scaled by the newly estimated distance; Cols.~12$-$14 are the same as Cols.~9$-$11 but with $R_0=8.0$~kpc and $\Theta_0=220$~km~s$^{-1}$. Columns~15$-$17 are the same as Cols.~9$-$11 but with $R_0=8.4$~kpc and $\Theta_0=254$~km~s$^{-1}$.}
\par
%\centering
\small
\begin{tabular}{rrrrrrrrrrrrrrrrr}
\hline
\hline 
$l$ & $b$ & $V_{\rm lsr}$& $L_{\rm CO}$ & $D$ & $M_{\rm GMCs}$ &Mark& Ref.& $D_{8.5}$ &
$R_{8.5}$ & $M_{8.5}$ & $D_{8.0}$& $R_{8.0}$& $M_{8.0}$ & $D_{8.4}$& $R_{8.4}$& $M_{8.4}$ \\
$(^\circ)$& $(^\circ)$& km~s$^{-1}$& $\rm 10^3~K~km~s^{-1}~pc^2$ & kpc& $10^5~M_{\odot}$& & & kpc& kpc&
$10^5~M_{\odot}$& kpc& kpc &$10^5~M_{\odot}$ & kpc& kpc &$10^5~M_{\odot}$ \\
 (1) & (2) & (3)& (4) &(5) & (6)
&(7) & (8)& (9) & (10) & (11) & (12)& (13)& (14) & (15)& (16)& (17) \\ 
\hline
  8.00 & --0.50 & 128.00 &   72.70 &10.10 &   4.435 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
  8.20 &  0.20 &  20.00 & 1402.00 &15.90 &  85.522 &kfar  &srby87 &13.4 &  5.2 &  60.74 &12.7 &  4.9 &  54.56 & 13.4 & 5.2 & 60.74\\
  8.30 & --0.10 &  48.00 &   50.20 &13.20 &   3.062 &kfar  &srby87 &11.5 &  3.3 &   2.32 &10.9 &  3.2 &   2.09 & 11.6 & 3.4 &  2.36\\
  8.30 & --0.30 &  16.00 &   28.10 & 3.20 &   1.714 &knear &srby87 & 2.9 &  5.6 &   1.41 & 2.7 &  5.3 &   1.22 &  2.8 & 5.8 &  1.31\\
  8.40 & --0.30 &  37.00 &  233.00 & 5.70 &  14.213 &knear &srby87 & 4.7 &  3.9 &   9.66 & 4.4 &  3.7 &   8.47 &  4.4 & 4.1 &  8.47\\
  8.50 & --1.00 &  16.00 &  106.00 & 3.20 &   6.466 &knear &srby87 & 2.9 &  5.7 &   5.31 & 2.7 &  5.4 &   4.60 &  2.7 & 5.8 &  4.60\\
  8.70 &  0.60 &  22.00 &    8.60 & 4.00 &   0.525 &knear &srby87 & 3.5 &  5.1 &   0.40 & 3.2 &  4.8 &   0.34 &  3.3 & 5.3 &  0.36\\
  9.30 &  0.70 &  15.00 &    2.00 & 2.80 &   0.122 &knear &srby87 & 2.6 &  6.0 &   0.11 & 2.4 &  5.6 &   0.09 &  2.4 & 6.1 &  0.09\\
  9.30 &  0.00 &  31.00 &   15.60 & 4.80 &   0.952 &knear &srby87 & 4.1 &  4.5 &   0.69 & 3.8 &  4.3 &   0.60 &  3.9 & 4.7 &  0.63\\
  9.60 &  0.80 &  26.00 &   53.90 & 4.20 &   3.288 &knear &srby87 & 3.6 &  4.9 &   2.42 & 3.4 &  4.7 &   2.15 &  3.4 & 5.1 &  2.15\\
  9.80 & --0.75 &  28.00 &   18.30 & 4.40 &   1.116 &knear &srby87 & 3.8 &  4.8 &   0.83 & 3.5 &  4.6 &   0.71 &  3.6 & 5.0 &  0.75\\
 10.00 & --0.04 &  32.00 &   18.30 & 4.70 &   1.116 &knear &srby87 & 4.0 &  4.6 &   0.81 & 3.7 &  4.4 &   0.69 &  3.8 & 4.8 &  0.73\\
 10.80 & --0.80 &  29.00 &   38.00 & 4.20 &   2.318 &knear &srby87 & 3.7 &  5.0 &   1.80 & 3.4 &  4.7 &   1.52 &  3.5 & 5.1 &  1.61\\
 11.10 & --0.50 &  34.00 &  578.00 & 4.80 &  35.258 &kfar  &srby87 &12.7 &  4.7 & 246.82 &12.0 &  4.4 & 220.36 & 12.8 & 4.7 &250.72\\
 11.40 & --0.30 &  49.00 &   17.90 & 6.10 &   1.092 &knear &srby87 & 4.8 &  3.9 &   0.68 & 4.4 &  3.8 &   0.57 &  4.5 & 4.2 &  0.59\\
 11.60 & --0.40 &  32.00 &   16.70 & 4.30 &   1.019 &knear &srby87 & 3.7 &  4.9 &   0.75 & 3.5 &  4.6 &   0.68 &  3.5 & 5.1 &  0.68\\
 11.70 & --0.30 &  47.00 &   12.30 & 5.60 &   0.750 &knear &srby87 & 4.6 &  4.1 &   0.51 & 4.3 &  3.9 &   0.44 &  4.3 & 4.3 &  0.44\\
 11.90 &  0.80 &  26.00 &   22.40 & 3.60 &   1.366 &knear &srby87 & 3.2 &  5.4 &   1.08 & 3.0 &  5.1 &   0.95 &  3.0 & 5.6 &  0.95\\
 11.90 & --0.10 &  42.00 &   84.60 &14.40 &   5.161 &kfar  &srby87 &12.3 &  4.4 &   3.77 &11.6 &  4.2 &   3.35 & 12.4 & 4.4 &  3.83\\
 12.00 & --0.60 &  35.00 &   29.50 & 4.50 &   1.799 &knear &srby87 & 3.9 &  4.8 &   1.35 & 3.6 &  4.5 &   1.15 &  3.7 & 5.0 &  1.22\\
 12.20 & --0.10 &  23.00 &   83.00 &16.30 &   5.063 &kfar  &srby87 &13.7 &  5.7 &   3.58 &12.9 &  5.4 &   3.17 & 13.6 & 5.6 &  3.52\\
 12.40 & --0.70 &  25.00 &    3.20 & 3.40 &   0.195 &knear &srby87 & 3.1 &  5.5 &   0.16 & 2.9 &  5.2 &   0.14 &  2.9 & 5.7 &  0.14\\
 12.70 &  0.70 &  18.00 &   28.10 & 2.50 &   1.714 &knear &srby87 & 2.4 &  6.2 &   1.58 & 2.2 &  5.8 &   1.33 &  2.3 & 6.3 &  1.45\\
 12.70 & --0.10 &  56.00 &   45.70 &13.60 &   2.788 &kfar  &srby87 &11.7 &  3.9 &   2.06 &11.1 &  3.7 &   1.86 & 11.8 & 4.0 &  2.10\\
 12.80 & --0.20 &  32.00 &  278.00 & 4.10 &  16.958 &knear &srby87 & 3.5 &  5.1 &  12.36 & 3.3 &  4.8 &  10.99 &  3.3 & 5.3 & 10.99\\
 12.90 &  0.50 &  35.00 &   10.50 & 4.30 &   0.640 &knear &srby87 & 3.7 &  4.9 &   0.47 & 3.5 &  4.7 &   0.42 &  3.5 & 5.1 &  0.42\\
 13.00 &  0.40 &  32.00 &   35.30 & 4.00 &   2.153 &knear &srby87 & 3.5 &  5.1 &   1.65 & 3.3 &  4.9 &   1.47 &  3.3 & 5.3 &  1.47\\
 13.00 & --0.10 &  47.00 &  140.00 & 5.30 &   8.540 &knear &srby87 & 4.4 &  4.3 &   5.89 & 4.1 &  4.1 &   5.11 &  4.2 & 4.5 &  5.36\\
 13.70 & --0.10 &  48.00 &   31.60 & 5.20 &   1.928 &knear &srby87 & 4.4 &  4.4 &   1.38 & 4.1 &  4.2 &   1.20 &  4.1 & 4.6 &  1.20\\
 13.90 &  0.90 &  27.00 &   13.20 & 3.30 &   0.805 &knear &srby87 & 3.0 &  5.6 &   0.67 & 2.8 &  5.3 &   0.58 &  2.9 & 5.8 &  0.62\\
 13.90 &  0.30 &  49.00 &   47.20 & 5.20 &   2.879 &knear &srby87 & 4.4 &  4.4 &   2.06 & 4.1 &  4.2 &   1.79 &  4.1 & 4.6 &  1.79\\
 14.00 & --0.10 &  26.00 &    7.60 & 3.20 &   0.464 &knear &srby87 & 2.9 &  5.7 &   0.38 & 2.7 &  5.4 &   0.33 &  2.8 & 5.8 &  0.36\\
 14.00 & --0.60 &  18.00 &  195.00 & 2.40 &  11.895 &knear &srby87 & 2.2 &  6.4 &  10.00 & 2.1 &  6.0 &   9.11 &  2.2 & 6.4 & 10.00\\
 14.20 & --0.20 &  39.00 &  297.00 & 4.40 &  18.117 &knear &srby87 & 3.8 &  4.9 &  13.51 & 3.5 &  4.6 &  11.46 &  3.6 & 5.1 & 12.13\\
 14.60 & --0.60 &  37.00 &   17.90 & 4.20 &   1.092 &knear &srby87 & 3.6 &  5.1 &   0.80 & 3.4 &  4.8 &   0.72 &  3.4 & 5.2 &  0.72\\
 15.00 & --0.70 &  20.00 &   25.10 & 2.50 &   1.531 &knear &srby87 & 2.3 &  6.3 &   1.30 & 2.2 &  5.9 &   1.19 &  2.2 & 6.4 &  1.19\\
 15.00 &  0.00 &  25.00 &    3.50 & 3.00 &   0.213 &knear &srby87 & 2.7 &  5.9 &   0.17 & 2.6 &  5.6 &   0.16 &  2.6 & 6.0 &  0.16\\
 15.10 &  0.90 &  43.00 &    4.80 & 4.60 &   0.293 &knear &srby87 & 3.9 &  4.8 &   0.21 & 3.7 &  4.6 &   0.19 &  3.7 & 5.0 &  0.19\\
 15.30 &  0.00 &  32.00 &   18.10 & 3.60 &   1.104 &knear &srby87 & 3.2 &  5.5 &   0.87 & 3.0 &  5.2 &   0.77 &  3.1 & 5.6 &  0.82\\
 15.70 & --0.20 &  57.00 &   13.20 & 5.40 &   0.805 &knear &srby87 & 4.6 &  4.3 &   0.58 & 4.2 &  4.1 &   0.49 &  4.3 & 4.5 &  0.51\\
 15.80 & --0.60 &  48.00 &   18.30 & 4.80 &   1.116 &knear &srby87 & 4.1 &  4.7 &   0.81 & 3.8 &  4.4 &   0.70 &  3.9 & 4.9 &  0.74\\
 15.90 & --0.60 &  19.00 &   16.00 & 2.20 &   0.976 &knear &srby87 & 2.1 &  6.5 &   0.89 & 2.0 &  6.1 &   0.81 &  2.1 & 6.5 &  0.89\\
 15.90 &  0.30 &  19.00 &    4.20 & 2.20 &   0.256 &knear &srby87 & 2.1 &  6.5 &   0.23 & 2.0 &  6.1 &   0.21 &  2.1 & 6.5 &  0.23\\
 15.90 & --1.00 &  58.00 &    7.30 & 5.50 &   0.445 &knear &srby87 & 4.6 &  4.3 &   0.31 & 4.3 &  4.1 &   0.27 &  4.3 & 4.5 &  0.27\\
 16.20 & --0.80 &  56.00 &    3.20 & 5.30 &   0.195 &knear &srby87 & 4.5 &  4.4 &   0.14 & 4.1 &  4.2 &   0.12 &  4.2 & 4.6 &  0.12\\
 16.25 & --0.05 &  48.00 &  143.00 &14.50 &   8.723 &kfar  &srby87 &12.3 &  4.7 &   6.28 &11.6 &  4.5 &   5.58 & 12.3 & 4.8 &  6.28\\
 16.30 &  0.40 &  28.00 &   16.10 & 3.10 &   0.982 &knear &srby87 & 2.8 &  5.8 &   0.80 & 2.7 &  5.5 &   0.74 &  2.7 & 6.0 &  0.74\\
 16.50 &  0.40 &  38.00 &   53.60 & 3.90 &   3.270 &knear &srby87 & 3.5 &  5.3 &   2.63 & 3.2 &  5.0 &   2.20 &  3.3 & 5.4 &  2.34\\
 16.65 & --0.05 &  59.00 &   49.80 & 5.40 &   3.038 &knear &srby87 & 4.6 &  4.3 &   2.20 & 4.2 &  4.1 &   1.84 &  4.3 & 4.6 &  1.93\\
 16.70 & --0.45 &  43.00 &   66.00 & 4.30 &   4.026 &knear &srby87 & 3.7 &  5.0 &   2.98 & 3.5 &  4.8 &   2.67 &  3.5 & 5.2 &  2.67\\
 16.80 &  0.40 &  22.00 &         & 2.00 &   6.166 &stel  &dect86 &     &      &   6.17 &     &      &   6.17 &      &     &  6.17\\
 16.80 &  0.10 &  30.00 &  284.00 &15.90 &  17.324 &kfar  &srby87 &13.4 &  5.8 &  12.30 &12.6 &  5.4 &  10.88 & 13.3 & 5.7 & 12.12\\
 16.90 &  0.30 &  24.00 &   97.30 & 2.60 &   5.935 &knear &srby87 & 2.5 &  6.2 &   5.49 & 2.3 &  5.8 &   4.64 &  2.4 & 6.3 &  5.06\\
 17.20 & --0.20 &  43.00 &  910.00 &14.90 &  55.510 &kfar  &srby87 &12.5 &  5.1 &  39.07 &11.8 &  4.8 &  34.81 & 12.6 & 5.1 & 39.70\\
 17.30 & --0.90 &  46.00 &    5.30 & 4.40 &   0.323 &knear &srby87 & 3.8 &  5.0 &   0.24 & 3.6 &  4.7 &   0.22 &  3.6 & 5.1 &  0.22\\
 17.70 &  0.30 &  22.00 &    9.90 & 2.40 &   0.604 &knear &srby87 & 2.2 &  6.4 &   0.51 & 2.1 &  6.0 &   0.46 &  2.2 & 6.5 &  0.51\\
 17.70 &  0.10 &  41.00 &   51.00 &15.10 &   3.111 &kfar  &srby87 &12.7 &  5.2 &   2.20 &11.9 &  5.0 &   1.93 & 12.7 & 5.2 &  2.20\\
 18.15 & --0.30 &  54.00 &  245.00 & 4.90 &  14.945 &knear &srby87 & 4.2 &  4.7 &  10.98 & 3.9 &  4.5 &   9.47 &  3.9 & 4.9 &  9.47\\
 18.65 &  0.30 &  21.00 &   63.50 &16.80 &   3.873 &kfar  &srby87 &14.0 &  6.6 &   2.69 &13.2 &  6.2 &   2.39 & 13.9 & 6.4 &  2.65\\
\hline
 18.85 &  0.05 &  50.00 &  330.00 &14.40 &  20.130 &kfar  &srby87 &12.2 &  5.0 &  14.45 &11.5 &  4.7 &  12.84 & 12.2 & 5.0 & 14.45\\
 18.85 & --0.50 &  66.00 &  576.00 & 5.60 &  35.136 &knear &srby87 & 4.7 &  4.4 &  24.75 & 4.3 &  4.2 &  20.72 &  4.4 & 4.6 & 21.69\\
 19.20 & --0.30 &  33.00 &  272.00 &15.70 &  16.592 &kfar  &srby87 &13.2 &  5.8 &  11.73 &12.4 &  5.5 &  10.35 & 13.1 & 5.8 & 11.55\\
 19.25 &  0.05 &  26.00 &    7.70 & 2.60 &   0.470 &knear &srby87 & 2.4 &  6.3 &   0.40 & 2.3 &  5.9 &   0.37 &  2.3 & 6.3 &  0.37\\
 19.50 &  0.00 &  26.00 &         & 2.20 &   0.891 &stel  &dect86 &     &      &   0.89 &     &      &   0.89 &      &     &  0.89\\
 19.60 & --0.05 &  58.00 &   13.70 & 4.90 &   0.836 &knear &srby87 & 4.2 &  4.7 &   0.61 & 3.9 &  4.5 &   0.53 &  4.0 & 4.9 &  0.56\\
 19.65 & --0.65 &  55.00 &   14.60 & 4.80 &   0.891 &knear &srby87 & 4.1 &  4.8 &   0.65 & 3.8 &  4.6 &   0.56 &  3.9 & 5.0 &  0.59\\
 19.70 &  0.10 &  26.00 &  218.00 &16.30 &  13.298 &kfar  &srby87 &13.6 &  6.3 &   9.26 &12.8 &  5.9 &   8.20 & 13.5 & 6.2 &  9.12\\
 19.80 & --0.45 &  69.00 &  153.00 & 5.60 &   9.333 &knear &srby87 & 4.7 &  4.4 &   6.57 & 4.4 &  4.2 &   5.76 &  4.4 & 4.6 &  5.76\\
 19.90 & --0.55 &  44.00 &  111.00 & 4.00 &   6.771 &knear &srby87 & 3.5 &  5.3 &   5.18 & 3.3 &  5.0 &   4.61 &  3.3 & 5.5 &  4.61\\
 20.55 & --0.45 &  67.00 &  108.00 &13.30 &   6.588 &kfar  &srby87 &11.3 &  4.5 &   4.76 &10.7 &  4.3 &   4.26 & 11.4 & 4.6 &  4.84\\
 20.70 & --0.30 &  63.00 &   72.70 &13.60 &   4.435 &kfar  &srby87 &11.5 &  4.7 &   3.17 &10.9 &  4.4 &   2.85 & 11.6 & 4.7 &  3.23\\
 20.75 & --0.10 &  59.00 &  168.00 &13.80 &  10.248 &kfar  &srby87 &11.7 &  4.8 &   7.37 &11.0 &  4.6 &   6.51 & 11.8 & 4.8 &  7.49\\
 20.75 &  0.10 &  77.00 &   31.60 & 6.00 &   1.928 &knear &srby87 & 5.0 &  4.2 &   1.34 & 4.6 &  4.0 &   1.13 &  4.7 & 4.5 &  1.18\\
 20.85 &  0.00 &  30.00 &  262.00 &15.90 &  15.982 &kfar  &srby87 &13.3 &  6.2 &  11.18 &12.5 &  5.8 &   9.88 & 13.2 & 6.1 & 11.02\\
 20.90 & --0.30 &  68.00 &   31.60 & 5.40 &   1.928 &knear &srby87 & 4.6 &  4.5 &   1.40 & 4.3 &  4.3 &   1.22 &  4.3 & 4.7 &  1.22\\
 21.35 &  0.35 &   8.00 &    3.00 & 0.80 &   0.183 &knear &srby87 & 0.7 &  7.8 &   0.14 & 0.8 &  7.3 &   0.18 &  0.9 & 7.7 &  0.23\\
 21.40 & --0.65 &  55.00 &   87.50 & 4.60 &   5.337 &knear &srby87 & 4.0 &  5.0 &   4.04 & 3.7 &  4.7 &   3.45 &  3.8 & 5.2 &  3.64\\
 21.40 &  0.00 &  75.00 &  153.00 &12.80 &   9.333 &kfar  &srby87 &11.0 &  4.3 &   6.89 &10.4 &  4.1 &   6.16 & 11.1 & 4.4 &  7.02\\
 21.50 &  0.25 &  77.00 &    5.50 & 5.90 &   0.335 &knear &srby87 & 5.0 &  4.3 &   0.24 & 4.6 &  4.1 &   0.20 &  4.6 & 4.5 &  0.20\\
 21.75 &  0.00 &  67.00 &   46.20 & 5.30 &   2.818 &knear &srby87 & 4.5 &  4.6 &   2.03 & 4.2 &  4.4 &   1.77 &  4.2 & 4.8 &  1.77\\
 21.90 & --0.35 &  82.00 &  409.00 &12.40 &  24.949 &kfar  &srby87 &10.6 &  4.2 &  18.23 &10.1 &  4.0 &  16.55 & 10.8 & 4.3 & 18.93\\
 22.05 &  0.20 &  50.00 &   70.90 & 4.20 &   4.325 &knear &srby87 & 3.7 &  5.3 &   3.36 & 3.4 &  5.0 &   2.83 &  3.5 & 5.4 &  3.00\\
 22.10 & --0.20 &  53.00 &         & 4.40 &  11.482 &knear &dect86 & 3.8 &  5.2 &   8.56 & 3.6 &  4.9 &   7.69 &  3.6 & 5.3 &  7.69\\
 22.35 &  0.10 &  85.00 &   71.70 &12.20 &   4.374 &kfar  &srby87 &10.5 &  4.2 &   3.24 &10.0 &  4.0 &   2.94 & 10.6 & 4.3 &  3.30\\
 22.40 &  0.30 &  84.00 &   53.00 &12.20 &   3.233 &kfar  &srby87 &10.5 &  4.2 &   2.39 &10.0 &  4.0 &   2.17 & 10.7 & 4.3 &  2.49\\
 22.55 & --0.05 & 115.00 &   23.20 & 8.50 &   1.415 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 22.55 & --0.20 &  77.00 &   64.80 &12.60 &   3.953 &kfar  &srby87 &10.8 &  4.4 &   2.90 &10.2 &  4.2 &   2.59 & 10.9 & 4.5 &  2.96\\
 22.75 & --0.25 & 109.00 &   40.30 & 9.20 &   2.458 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 22.80 &  0.40 &  92.00 &   90.50 &11.70 &   5.520 &kfar  &srby87 &10.2 &  4.0 &   4.20 & 9.7 &  3.9 &   3.79 & 10.4 & 4.1 &  4.36\\
 22.85 &  0.40 & 114.00 &    8.10 & 8.40 &   0.494 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 23.00 & --0.10 &  78.00 &         & 5.90 &  11.220 &knear &dect86 & 4.9 &  4.4 &   7.74 & 4.6 &  4.2 &   6.82 &  4.6 & 4.6 &  6.82\\
 23.00 & --0.10 &  78.00 &         &12.60 &  51.286 &kfar  &dect86 &10.7 &  4.4 &  36.98 &10.2 &  4.2 &  33.61 & 10.9 & 4.5 & 38.38\\
 23.00 & --0.40 &  74.00 & 2901.00 &12.80 & 176.961 &kfar  &srby87 &10.9 &  4.5 & 128.32 &10.3 &  4.3 & 114.59 & 11.0 & 4.6 &130.69\\
 23.00 &  0.00 &  81.00 &  202.00 &12.40 &  12.322 &kfar  &srby87 &10.6 &  4.3 &   9.00 &10.1 &  4.1 &   8.17 & 10.8 & 4.4 &  9.35\\
 23.10 &  0.60 &  38.00 &    3.30 & 3.20 &   0.201 &knear &srby87 & 2.9 &  5.9 &   0.17 & 2.8 &  5.6 &   0.15 &  2.8 & 6.0 &  0.15\\
 23.30 & --0.30 &  64.00 &         & 4.00  &   0.450 &knear &tllw07 &4.3  & 4.9  &   0.52 & 4.0 &  4.6 &   0.45 & 4.0  & 5.0 &  0.45\\
 23.40 & --0.25 & 102.00 &  258.00 &11.00 &  15.738 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 23.50 & --0.40 &  73.00 &   21.80 & 5.50 &   1.330 &knear &srby87 & 4.7 &  4.6 &   0.97 & 4.3 &  4.4 &   0.81 &  4.4 & 4.8 &  0.85\\
 23.55 &  0.20 &  82.00 &  280.00 &12.30 &  17.080 &kfar  &srby87 &10.5 &  4.3 &  12.45 &10.0 &  4.1 &  11.29 & 10.7 & 4.4 & 12.93\\
 23.70 &  0.50 &  82.00 &    9.70 & 6.10 &   0.592 &knear &srby87 & 5.1 &  4.4 &   0.41 & 4.7 &  4.2 &   0.35 &  4.7 & 4.6 &  0.35\\
 23.80 &  0.10 &  98.00 &         & 7.10 &  28.840 &knear &dect86 & 5.8 &  4.0 &  19.25 & 5.3 &  3.8 &  16.07 &  5.3 & 4.2 & 16.07\\
 23.95 &  0.15 &  79.00 &   15.00 & 5.80 &   0.915 &knear &srby87 & 4.9 &  4.5 &   0.65 & 4.6 &  4.2 &   0.58 &  4.6 & 4.7 &  0.58\\
 24.20 & --0.05 &  88.00 &   90.70 &11.90 &   5.533 &kfar  &srby87 &10.2 &  4.2 &   4.07 & 9.7 &  4.1 &   3.68 & 10.4 & 4.4 &  4.23\\
 24.40 & --0.25 &  60.00 &  111.00 &13.60 &   6.771 &kfar  &srby87 &11.4 &  5.1 &   4.76 &10.8 &  4.8 &   4.27 & 11.5 & 5.1 &  4.84\\
 24.45 & --0.80 &  56.00 &    4.80 & 4.30 &   0.293 &knear &srby87 & 3.9 &  5.2 &   0.24 & 3.6 &  5.0 &   0.21 &  3.7 & 5.4 &  0.22\\
 24.45 &  0.25 & 120.00 & 1095.00 & 9.10 &  66.795 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 24.45 & --0.50 &  44.00 &   98.50 & 3.50 &   6.008 &knear &srby87 & 3.2 &  5.7 &   5.02 & 3.0 &  5.4 &   4.41 &  3.1 & 5.8 &  4.71\\
 24.50 &  0.20 &  38.00 &   78.20 &15.10 &   4.770 &kfar  &srby87 &12.6 &  6.0 &   3.32 &11.9 &  5.7 &   2.96 & 12.5 & 5.9 &  3.27\\
 24.50 & --0.15 &  44.00 &   21.50 & 3.50 &   1.311 &knear &srby87 & 3.2 &  5.7 &   1.10 & 3.0 &  5.4 &   0.96 &  3.1 & 5.8 &  1.03\\
 24.50 & --0.70 &  47.00 &   11.80 & 3.70 &   0.720 &knear &srby87 & 3.4 &  5.6 &   0.61 & 3.2 &  5.3 &   0.54 &  3.2 & 5.7 &  0.54\\
 24.50 & --0.25 & 101.00 &  157.00 &10.90 &   9.577 &kfar  &srby87 & 9.6 &  4.0 &   7.43 & 9.1 &  3.8 &   6.68 &  9.9 & 4.1 &  7.90\\
 24.55 & --0.50 &  61.00 &   35.00 & 4.80 &   2.135 &knear &srby87 & 4.1 &  5.1 &   1.56 & 3.8 &  4.8 &   1.34 &  3.9 & 5.2 &  1.41\\
 24.60 & --0.15 &  83.00 &  212.00 &12.10 &  12.932 &kfar  &srby87 &10.4 &  4.4 &   9.55 & 9.8 &  4.2 &   8.48 & 10.5 & 4.5 &  9.74\\
 24.65 & --0.10 & 114.00 &   63.40 & 9.10 &   3.867 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 25.20 &  0.15 & 105.00 &   85.90 & 9.10 &   5.240 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 25.25 &  0.30 &  47.00 &  161.00 &14.40 &   9.821 &kfar  &srby87 &12.0 &  5.7 &   6.82 &11.3 &  5.3 &   6.05 & 12.0 & 5.6 &  6.82\\
 25.30 & --0.50 &  55.00 &         &13.90 &  47.863 &kfar  &dect86 &11.6 &  5.3 &  33.33 &10.9 &  5.0 &  29.43 & 11.6 & 5.4 & 33.33\\
 25.45 & --0.20 &  65.00 &  620.00 &13.20 &  37.820 &kfar  &srby87 &11.1 &  5.0 &  26.74 &10.5 &  4.7 &  23.93 & 11.2 & 5.1 & 27.23\\
 25.45 & --0.20 & 120.00 &   32.20 & 9.00 &   1.964 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 25.55 & --0.40 & 116.00 &   10.40 & 9.00 &   0.634 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 25.65 & --0.10 &  94.00 &  793.00 &11.50 &  48.373 &kfar  &srby87 & 9.7 &  4.2 &  34.42 & 9.3 &  4.0 &  31.64 & 10.0 & 4.3 & 36.58\\
 25.70 & --0.15 & 109.00 &  133.00 & 9.00 &   8.113 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 25.80 &  0.45 &  48.00 &   34.70 & 3.70 &   2.117 &knear &srby87 & 3.4 &  5.7 &   1.79 & 3.2 &  5.3 &   1.58 &  3.2 & 5.8 &  1.58\\
\hline
 25.80 &  0.25 & 109.00 &  347.00 & 9.00 &  21.167 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 25.90 &  0.20 &  69.00 &    4.50 & 5.10 &   0.274 &knear &srby87 & 4.4 &  4.9 &   0.20 & 4.1 &  4.7 &   0.18 &  4.2 & 5.1 &  0.19\\
 25.90 & --0.15 & 107.00 &  273.00 & 9.00 &  16.653 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 25.95 & --0.60 &  63.00 &    9.40 & 4.70 &   0.573 &knear &srby87 & 4.1 &  5.1 &   0.44 & 3.9 &  4.8 &   0.39 &  3.9 & 5.3 &  0.39\\
 26.20 &  0.10 &  71.00 &   17.30 & 5.20 &   1.055 &knear &srby87 & 4.5 &  4.9 &   0.79 & 4.2 &  4.6 &   0.69 &  4.2 & 5.1 &  0.69\\
 26.35 &  0.80 &  47.00 &    5.30 & 3.60 &   0.323 &knear &srby87 & 3.3 &  5.7 &   0.27 & 3.1 &  5.4 &   0.24 &  3.2 & 5.8 &  0.26\\
 26.50 & --0.60 &  67.00 &   93.50 & 4.90 &   5.703 &knear &srby87 & 4.3 &  5.0 &   4.39 & 4.0 &  4.8 &   3.80 &  4.1 & 5.2 &  3.99\\
 26.55 & --0.30 & 108.00 &   39.50 & 8.90 &   2.409 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 26.60 &  0.00 &  26.00 &  186.00 &15.80 &  11.346 &kfar  &srby87 &13.2 &  6.8 &   7.92 &12.4 &  6.3 &   6.99 & 13.0 & 6.6 &  7.68\\
 26.65 &  0.00 & 100.00 &  222.00 & 9.60 &  13.542 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 26.65 &  0.00 & 112.00 &   23.90 & 8.90 &   1.458 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 26.70 &  0.50 &  86.00 &   13.40 & 6.20 &   0.817 &knear &srby87 & 5.2 &  4.5 &   0.57 & 4.8 &  4.3 &   0.49 &  4.8 & 4.7 &  0.49\\
 26.90 &  0.10 &  91.00 &   44.40 & 6.50 &   2.708 &knear &srby87 & 5.5 &  4.4 &   1.94 & 5.0 &  4.2 &   1.60 &  5.1 & 4.6 &  1.67\\
 26.95 & --0.40 &  69.00 &    5.70 & 5.00 &   0.348 &knear &srby87 & 4.4 &  5.0 &   0.27 & 4.1 &  4.7 &   0.23 &  4.1 & 5.2 &  0.23\\
 26.95 & --0.10 &  81.00 &   22.10 & 5.80 &   1.348 &knear &srby87 & 5.0 &  4.6 &   1.00 & 4.6 &  4.4 &   0.85 &  4.6 & 4.8 &  0.85\\
 27.05 & --0.15 & 101.00 &   21.50 & 8.90 &   1.311 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 27.25 &  0.15 &  33.00 &  148.00 &15.20 &   9.028 &kfar  &srby87 &12.7 &  6.4 &   6.30 &11.9 &  6.0 &   5.53 & 12.5 & 6.3 &  6.11\\
 27.30 & --0.30 &  72.00 &   14.50 & 5.20 &   0.884 &knear &srby87 & 4.5 &  4.9 &   0.66 & 4.2 &  4.7 &   0.58 &  4.3 & 5.1 &  0.60\\
 27.35 & --0.15 &  93.00 &   18.00 & 6.60 &   1.098 &knear &srby87 & 5.6 &  4.4 &   0.79 & 5.1 &  4.2 &   0.66 &  5.1 & 4.6 &  0.66\\
 27.50 &  0.20 &  36.00 &   46.10 &15.00 &   2.812 &kfar  &srby87 &12.5 &  6.3 &   1.95 &11.7 &  5.9 &   1.71 & 12.4 & 6.2 &  1.92\\
 27.50 &  0.15 &  95.00 &  167.00 &10.90 &  10.187 &kfar  &srby87 & 9.4 &  4.3 &   7.58 & 9.0 &  4.1 &   6.95 &  9.7 & 4.5 &  8.07\\
 27.65 &  0.10 &  83.00 &   32.10 & 5.90 &   1.958 &knear &srby87 & 5.1 &  4.6 &   1.46 & 4.7 &  4.4 &   1.24 &  4.7 & 4.8 &  1.24\\
 27.75 &  0.10 & 101.00 &  141.00 & 8.90 &   8.601 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 28.20 & --0.05 &  95.00 &  150.00 &10.80 &   9.150 &kfar  &srby87 & 9.3 &  4.4 &   6.78 & 8.8 &  4.2 &   6.07 &  9.6 & 4.5 &  7.23\\
 28.30 & --0.35 &  47.00 &   18.60 & 3.50 &   1.135 &knear &srby87 & 3.2 &  5.9 &   0.95 & 3.0 &  5.5 &   0.83 &  3.1 & 6.0 &  0.89\\
 28.30 & --0.10 &  81.00 &  186.00 & 5.80 &  11.346 &knear &srby87 & 5.0 &  4.7 &   8.43 & 4.6 &  4.5 &   7.14 &  4.6 & 4.9 &  7.14\\
 28.60 &  0.05 & 100.00 &  141.00 & 8.80 &   8.601 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 28.60 &  0.20 &  79.00 &   68.10 &11.90 &   4.154 &kfar  &srby87 &10.0 &  4.8 &   2.93 & 9.5 &  4.6 &   2.65 & 10.2 & 4.9 &  3.05\\
 28.80 & --0.10 &  80.00 &         & 5.70 &  21.380 &knear &dect86 & 4.9 &  4.8 &  15.80 & 4.6 &  4.6 &  13.92 &  4.6 & 5.0 & 13.92\\
 28.85 & --0.25 &  88.00 &   94.40 & 6.30 &   5.758 &knear &srby87 & 5.4 &  4.6 &   4.23 & 4.9 &  4.4 &   3.48 &  4.9 & 4.8 &  3.48\\
 28.95 & --0.65 &  51.00 &    7.60 & 3.80 &   0.464 &knear &srby87 & 3.4 &  5.7 &   0.37 & 3.2 &  5.4 &   0.33 &  3.3 & 5.8 &  0.35\\
 29.00 &  0.05 &  98.00 &  255.00 & 8.80 &  15.555 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 29.10 & --0.30 &  94.00 &  223.00 & 7.10 &  13.603 &knear &srby87 & 5.7 &  4.5 &   8.77 & 5.2 &  4.3 &   7.30 &  5.2 & 4.7 &  7.30\\
 29.35 & --0.45 &  77.00 &  141.00 & 5.40 &   8.601 &knear &srby87 & 4.8 &  4.9 &   6.80 & 4.4 &  4.7 &   5.71 &  4.5 & 5.1 &  5.97\\
 29.35 & --0.55 &  64.00 &   25.70 & 4.60 &   1.568 &knear &srby87 & 4.1 &  5.3 &   1.25 & 3.8 &  5.0 &   1.07 &  3.9 & 5.5 &  1.13\\
 29.55 &  0.20 &  79.00 &   87.40 & 5.60 &   5.331 &knear &srby87 & 4.9 &  4.9 &   4.08 & 4.5 &  4.6 &   3.44 &  4.6 & 5.1 &  3.60\\
 29.60 & --0.60 &  75.00 &  274.00 &12.10 &  16.714 &kfar  &srby87 &10.1 &  5.0 &  11.65 & 9.6 &  4.7 &  10.52 & 10.2 & 5.1 & 11.88\\
 29.85 & --0.05 & 100.00 &  350.00 & 8.70 &  21.350 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 29.90 &  0.10 &  40.00 &   44.00 &14.40 &   2.684 &kfar  &srby87 &12.0 &  6.2 &   1.86 &11.2 &  5.9 &   1.62 & 11.9 & 6.2 &  1.83\\
 29.90 & --0.80 &  83.00 &   17.50 & 5.90 &   1.067 &knear &srby87 & 5.1 &  4.8 &   0.80 & 4.7 &  4.6 &   0.68 &  4.8 & 5.0 &  0.71\\
 30.30 & --0.25 & 105.00 &  265.00 & 8.60 &  16.165 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 30.40 &  0.45 &  45.00 &    2.00 & 3.30 &   0.122 &knear &srby87 & 3.1 &  6.1 &   0.11 & 2.9 &  5.7 &   0.09 &  3.0 & 6.1 &  0.10\\
 30.55 &  0.35 &  95.00 &   79.40 & 8.60 &   4.843 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 30.60 & --0.05 &  43.00 &  321.00 &14.10 &  19.581 &kfar  &srby87 &11.7 &  6.2 &  13.48 &11.0 &  5.8 &  11.92 & 11.6 & 6.1 & 13.25\\
 30.60 & --0.45 &  94.00 &   75.70 &10.30 &   4.618 &kfar  &srby87 & 8.8 &  4.6 &   3.37 & 8.4 &  4.4 &   3.07 &  9.1 & 4.7 &  3.60\\
 30.60 & --0.10 & 115.00 &   56.90 & 8.60 &   3.471 &ktan  &srby87 & 7.3 &  4.3 &   2.50 & 6.9 &  4.1 &   2.23 &      &     &      \\
 30.80 & --0.05 &  92.00 &  766.00 & 6.80 &  46.726 &knear &srby87 & 5.7 &  4.6 &  32.83 & 5.2 &  4.4 &  27.32 &  5.2 & 4.8 & 27.32\\
 30.85 & --0.15 &  53.00 &    8.20 & 3.90 &   0.500 &knear &srby87 & 3.5 &  5.8 &   0.40 & 3.3 &  5.4 &   0.36 &  3.3 & 5.9 &  0.36\\
 30.90 & --0.60 & 102.00 &   13.10 & 8.60 &   0.799 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 30.95 &  0.10 &  40.00 &  237.00 &14.20 &  14.457 &kfar  &srby87 &11.8 &  6.3 &   9.98 &11.1 &  5.9 &   8.83 & 11.7 & 6.2 &  9.81\\
 30.95 &  0.40 &  77.00 &   29.70 & 5.50 &   1.812 &knear &srby87 & 4.8 &  5.0 &   1.38 & 4.5 &  4.8 &   1.21 &  4.5 & 5.2 &  1.21\\
 31.00 &  0.00 &  48.00 &         &13.70 &  28.184 &kfar  &dect86 &11.4 &  6.0 &  19.52 &10.7 &  5.6 &  17.19 & 11.3 & 5.9 & 19.17\\
 31.30 &  0.00 &  79.00 &   41.90 & 5.60 &   2.556 &knear &srby87 & 4.9 &  5.0 &   1.96 & 4.6 &  4.7 &   1.72 &  4.6 & 5.2 &  1.72\\
 31.40 &  0.00 &  39.00 &  514.00 &14.20 &  31.354 &kfar  &srby87 &11.8 &  6.4 &  21.65 &11.1 &  6.0 &  19.16 & 11.7 & 6.3 & 21.29\\
 31.40 & --0.25 &  88.00 &   30.20 &10.70 &   1.842 &kfar  &srby87 & 9.0 &  4.8 &   1.30 & 8.6 &  4.5 &   1.19 &  9.3 & 4.9 &  1.39\\
 31.45 &  0.05 & 104.00 &  102.00 & 8.50 &   6.222 &ktan  &srby87 & 7.3 &  4.4 &   4.59 &     &      &        &      &     &      \\
 31.95 & --0.30 &  97.00 &   49.30 & 8.50 &   3.007 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 32.00 &  0.00 &  98.00 &  166.00 & 8.50 &  10.126 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 32.45 &  0.20 &  51.00 &   34.30 &13.30 &   2.092 &kfar  &srby87 &11.0 &  5.9 &   1.43 &10.3 &  5.6 &   1.25 & 10.9 & 5.9 &  1.41\\
 32.60 & --0.25 &  90.00 &  565.00 &10.30 &  34.465 &kfar  &srby87 & 8.6 &  4.8 &  24.03 & 8.2 &  4.6 &  21.84 &  8.9 & 4.9 & 25.73\\
 33.35 & --0.55 &  92.00 &   36.50 & 8.40 &   2.226 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 33.40 &  0.00 &  75.00 &  167.00 &11.40 &  10.187 &kfar  &srby87 & 9.4 &  5.2 &   6.93 & 8.9 &  4.9 &   6.21 &  9.6 & 5.3 &  7.22\\
 33.45 & --0.10 &  87.00 &  191.00 &10.30 &  11.651 &kfar  &srby87 & 8.6 &  4.9 &   8.12 & 8.2 &  4.7 &   7.38 &  8.9 & 5.0 &  8.70\\
\hline
 33.65 &  0.20 &  42.00 &   48.90 &13.60 &   2.983 &kfar  &srby87 &11.3 &  6.3 &   2.06 &10.6 &  6.0 &   1.81 & 11.2 & 6.3 &  2.02\\
 33.80 & --0.20 &  48.00 &  138.00 &13.20 &   8.418 &kfar  &srby87 &10.9 &  6.1 &   5.74 &10.3 &  5.7 &   5.13 & 10.9 & 6.1 &  5.74\\
 33.85 &  0.00 &  89.00 &   57.30 &10.00 &   3.495 &kfar  &srby87 & 8.3 &  4.9 &   2.41 & 7.9 &  4.6 &   2.18 &  8.7 & 5.0 &  2.65\\
 33.90 &  0.10 & 106.00 &  182.00 & 8.30 &  11.102 &ktan  &srby87 & 7.1 &  4.7 &   8.12 & 6.6 &  4.5 &   7.02 &  7.0 & 4.7 &  7.90\\
 34.15 & --0.10 &  89.00 &   40.40 & 8.30 &   2.464 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 34.25 &  0.10 &  53.00 &   50.90 & 3.70 &   3.105 &knear &srby87 & 3.5 &  6.0 &   2.78 & 3.3 &  5.6 &   2.47 &  3.3 & 6.0 &  2.47\\
 34.35 & --0.85 &  13.00 &    2.50 & 0.90 &   0.152 &knear &srby87 & 0.9 &  7.8 &   0.15 & 0.9 &  7.2 &   0.15 &  1.1 & 7.6 &  0.23\\
 34.40 & --0.20 &  53.00 &    6.70 & 3.70 &   0.409 &knear &srby87 & 3.5 &  6.0 &   0.37 & 3.3 &  5.6 &   0.33 &  3.3 & 6.0 &  0.33\\
 34.65 & --0.80 &  55.00 &   37.50 & 3.90 &   2.287 &knear &srby87 & 3.6 &  5.9 &   1.95 & 3.4 &  5.6 &   1.74 &  3.5 & 6.0 &  1.84\\
 34.70 & --0.70 &  46.00 &  230.00 & 3.30 &  14.030 &knear &srby87 & 3.1 &  6.2 &  12.38 & 2.9 &  5.9 &  10.83 &  3.0 & 6.3 & 11.60\\
 34.80 & --0.15 &  76.00 &   38.90 & 5.60 &   2.373 &knear &srby87 & 4.9 &  5.3 &   1.82 & 4.6 &  5.0 &   1.60 &  4.6 & 5.4 &  1.60\\
 35.05 &  0.30 &  51.00 &   44.10 & 3.60 &   2.690 &knear &srby87 & 3.4 &  6.1 &   2.40 & 3.2 &  5.7 &   2.13 &  3.2 & 6.1 &  2.13\\
 35.15 & --0.75 &  35.00 &   30.80 & 3.30 &   1.879 &knear &srby87 & 2.4 &  6.7 &   0.99 & 2.3 &  6.3 &   0.91 &  2.4 & 6.7 &  0.99\\
 35.20 & --0.10 &  13.00 &    3.80 & 0.90 &   0.232 &knear &srby87 & 0.9 &  7.8 &   0.23 & 0.9 &  7.3 &   0.23 &  1.1 & 7.6 &  0.35\\
 35.75 &  0.15 &  82.00 &  273.00 &10.20 &  16.653 &kfar  &srby87 & 8.3 &  5.2 &  11.03 & 7.9 &  4.9 &   9.99 &  8.6 & 5.3 & 11.84\\
 35.80 & --0.20 &  29.00 &    7.70 & 2.00 &   0.470 &knear &srby87 & 2.0 &  7.0 &   0.47 & 1.9 &  6.5 &   0.42 &  2.0 & 6.9 &  0.47\\
 35.95 & --0.50 &  58.00 &    6.20 & 4.10 &   0.378 &knear &srby87 & 3.8 &  5.9 &   0.32 & 3.6 &  5.5 &   0.29 &  3.6 & 6.0 &  0.29\\
 36.10 &  0.65 &  77.00 &  101.00 & 5.60 &   6.161 &knear &srby87 & 5.1 &  5.3 &   5.11 & 4.7 &  5.0 &   4.34 &  4.7 & 5.5 &  4.34\\
 36.30 & --0.10 &  57.00 &         &12.10 &  42.658 &kfar  &dect86 &10.0 &  5.9 &  29.14 & 9.4 &  5.6 &  25.74 & 10.0 & 5.9 & 29.14\\
 36.40 & --0.10 &  52.00 &   60.60 & 3.60 &   3.697 &knear &srby87 & 3.4 &  6.1 &   3.30 & 3.2 &  5.7 &   2.92 &  3.3 & 6.2 &  3.11\\
 36.50 & --0.15 &  77.00 &  107.00 & 5.90 &   6.527 &knear &srby87 & 5.1 &  5.3 &   4.88 & 4.8 &  5.0 &   4.32 &  4.8 & 5.5 &  4.32\\
 36.90 & --0.05 &  80.00 &   36.50 & 5.90 &   2.226 &knear &srby87 & 5.5 &  5.3 &   1.93 & 5.1 &  5.0 &   1.66 &  5.0 & 5.4 &  1.60\\
 37.35 &  0.25 &  88.00 &  199.00 & 8.60 &  12.139 &ktan  &srby87 & 6.8 &  5.2 &   7.59 & 6.4 &  4.9 &   6.72 &      &     &      \\
 37.45 &  0.10 &  41.00 &    4.00 & 3.00 &   0.244 &knear &srby87 & 2.7 &  6.5 &   0.20 & 2.6 &  6.1 &   0.18 &  2.7 & 6.6 &  0.20\\
 37.75 & --0.20 &  60.00 &  137.00 &11.60 &   8.357 &kfar  &srby87 & 9.5 &  5.9 &   5.61 & 8.9 &  5.5 &   4.92 &  9.5 & 5.9 &  5.61\\
 38.25 & --0.15 &  65.00 &   32.70 &11.10 &   1.995 &kfar  &srby87 & 9.0 &  5.8 &   1.31 & 8.5 &  5.4 &   1.17 &  9.1 & 5.8 &  1.34\\
 38.90 &  0.40 &  32.00 &         & 2.20 &   5.754 &knear &dect86 & 2.2 &  6.9 &   5.75 & 2.1 &  6.5 &   5.24 &  2.2 & 6.9 &  5.75\\
 38.95 & --0.45 &  42.00 &   16.30 & 2.90 &   0.994 &knear &srby87 & 2.8 &  6.5 &   0.93 & 2.7 &  6.1 &   0.86 &  2.8 & 6.6 &  0.93\\
 39.85 & --0.20 &  57.00 & 1718.00 &11.30 & 104.798 &kfar  &srby87 & 9.2 &  6.1 &  69.47 & 8.6 &  5.7 &  60.70 &  9.2 & 6.1 & 69.47\\
 40.30 & --0.45 &  74.00 &   14.70 & 5.70 &   0.897 &knear &srby87 & 5.5 &  5.6 &   0.84 & 5.1 &  5.3 &   0.72 &  5.0 & 5.7 &  0.69\\
 41.00 & --0.20 &  38.00 &   16.10 & 2.70 &   0.982 &knear &srby87 & 2.6 &  6.8 &   0.91 & 2.5 &  6.3 &   0.84 &  2.6 & 6.8 &  0.91\\
 41.15 & --0.20 &  60.00 &  464.00 &10.70 &  28.304 &kfar  &srby87 & 8.6 &  6.0 &  18.28 & 8.1 &  5.7 &  16.22 &  8.6 & 6.0 & 18.28\\
 41.90 & --0.40 &  59.00 &   84.30 &10.60 &   5.142 &kfar  &srby87 & 8.5 &  6.1 &   3.31 & 8.0 &  5.7 &   2.93 &  8.5 & 6.1 &  3.31\\
 42.15 & --0.60 &  67.00 &   50.30 & 5.40 &   3.068 &knear &srby87 & 5.0 &  5.8 &   2.63 & 4.7 &  5.5 &   2.32 &  4.7 & 5.9 &  2.32\\
 42.35 & --0.05 &  58.00 &   80.20 &10.50 &   4.892 &kfar  &srby87 & 8.4 &  6.1 &   3.13 & 7.9 &  5.8 &   2.77 &  8.5 & 6.1 &  3.21\\
 42.75 & --0.35 &  59.00 &   39.80 & 4.60 &   2.428 &knear &srby87 & 4.3 &  6.1 &   2.12 & 4.0 &  5.7 &   1.84 &  4.1 & 6.2 &  1.93\\
 43.10 &  0.05 &  12.00 &  102.00 &13.80 &   6.222 &kfar  &srby87 &11.7 &  8.0 &   4.47 &10.8 &  7.4 &   3.81 & 11.2 & 7.7 &  4.10\\
 43.20 & --0.50 &  57.00 &    7.50 & 4.40 &   0.457 &knear &srby87 & 4.1 &  6.2 &   0.40 & 3.9 &  5.8 &   0.36 &  4.0 & 6.2 &  0.38\\
 44.35 & --0.20 &  64.00 &  426.00 & 9.20 &  25.986 &kfar  &srby87 & 7.0 &  6.0 &  15.04 & 6.5 &  5.7 &  12.97 &  7.2 & 6.0 & 15.92\\
 45.45 &  0.05 &  58.00 &  289.00 & 9.50 &  17.629 &kfar  &srby87 & 7.4 &  6.2 &  10.70 & 6.9 &  5.8 &   9.30 &  7.4 & 6.2 & 10.70\\
 46.00 &  0.20 &  25.00 &         & 1.80 &   0.513 &knear &dect86 & 1.7 &  7.4 &   0.46 & 1.7 &  6.9 &   0.46 &  1.9 & 7.3 &  0.57\\
 46.30 & --0.20 &  59.00 &   46.50 & 9.10 &   2.836 &kfar  &srby87 & 6.8 &  6.2 &   1.58 & 6.3 &  5.8 &   1.36 &  6.9 & 6.2 &  1.63\\
 47.05 &  0.25 &  55.00 &   41.80 & 9.30 &   2.550 &kfar  &srby87 & 7.1 &  6.4 &   1.49 & 6.6 &  6.0 &   1.28 &  7.1 & 6.4 &  1.49\\
 48.60 &  0.25 &  10.00 &   36.50 &12.50 &   2.226 &kfar  &srby87 &10.6 &  8.1 &   1.60 & 9.9 &  7.5 &   1.40 & 10.2 & 7.8 &  1.48\\
 48.60 &  0.00 &  20.00 &   63.40 &11.80 &   3.867 &kfar  &srby87 & 9.9 &  7.7 &   2.72 & 9.2 &  7.1 &   2.35 &  9.5 & 7.5 &  2.51\\
 48.85 &  0.15 &  51.00 &   38.70 & 6.60 &   2.361 &ktan  &srby87 &     &      &        &     &      &        &      &     &      \\
 49.50 & --0.40 &  57.00 &  665.00 & 6.60 &  40.565 &ktan  &srby87 & 5.5 &  6.5 &  28.17 & 5.2 &  6.1 &  25.18 &  5.5 & 6.4 & 28.17\\
 49.75 & --0.55 &  68.00 &   15.40 & 6.50 &   0.939 &ktan  &srby87 & 5.5 &  6.5 &   0.67 & 5.2 &  6.1 &   0.60 &  5.4 & 6.4 &  0.65\\
 50.05 &  0.60 &  --2.00 &  112.00 &13.00 &   6.832 &kout  &srby87 &11.3 &  8.7 &   5.16 &10.4 &  8.1 &   4.37 & 10.7 & 8.4 &  4.63\\
 50.20 & --0.50 &  45.00 &         & 3.60 &   2.512 &kfar  &dect86 & 7.2 &  6.8 &  10.05 & 6.6 &  6.3 &   8.44 &  6.9 & 6.7 &  9.23\\
 50.85 &  0.25 &  43.00 &    5.70 & 3.60 &   0.348 &knear &srby87 & 3.6 &  6.8 &   0.35 & 3.5 &  6.4 &   0.33 &  3.7 & 6.8 &  0.37\\
 51.35 & --0.05 &  54.00 &   71.30 & 6.30 &   4.349 &ktan  &srby87 & 5.3 &  6.6 &   3.08 & 5.0 &  6.2 &   2.74 &  5.3 & 6.6 &  3.08\\
 52.10 & --0.30 &  59.00 &         & 6.10 &  18.197 &ktan  &dect86 & 5.2 &  6.7 &  13.22 & 4.9 &  6.3 &  11.74 &  5.2 & 6.6 & 13.22\\
 52.30 & --0.05 &  52.00 &    7.30 & 6.10 &   0.445 &ktan  &srby87 & 5.2 &  6.7 &   0.32 & 4.9 &  6.3 &   0.29 &  5.1 & 6.7 &  0.31\\
 52.55 & --0.95 &  64.00 &   97.20 & 6.10 &   5.929 &ktan  &srby87 & 5.2 &  6.7 &   4.31 & 4.9 &  6.4 &   3.83 &  5.1 & 6.7 &  4.14\\
 53.15 & --0.25 &  61.00 &   11.70 & 6.00 &   0.714 &ktan  &srby87 & 5.1 &  6.8 &   0.52 & 4.8 &  6.4 &   0.46 &  5.0 & 6.7 &  0.50\\
 53.55 &  0.05 &  23.00 &   11.10 & 1.80 &   0.677 &knear &srby87 & 1.7 &  7.6 &   0.60 & 1.8 &  7.1 &   0.68 &  2.0 & 7.5 &  0.84\\
 54.10 & --0.05 &  40.00 &   21.90 & 8.20 &   1.336 &kfar  &srby87 & 6.2 &  7.0 &   0.76 & 5.5 &  6.5 &   0.60 &  5.7 & 6.9 &  0.65\\
 54.65 &  0.80 &  31.00 &   67.10 & 9.00 &   4.093 &kfar  &srby87 & 7.2 &  7.3 &   2.62 & 6.6 &  6.8 &   2.20 &  6.8 & 7.2 &  2.34\\
 55.30 &  0.20 &  30.00 &   72.60 & 9.00 &   4.429 &kfar  &srby87 & 7.1 &  7.4 &   2.76 & 6.5 &  6.9 &   2.31 &  6.7 & 7.2 &  2.45\\
 55.70 &  0.00 &  36.00 &         & 5.60 &   8.318 &ktan  &dect86 &     &      &        &     &      &        &      &     &      \\
 56.80 &  1.90 & --54.60 &         &14.40 &   1.400 &kout  &  mk88 &14.9 & 12.5 &   1.50 &13.6 & 11.4 &   1.25 & 13.4 &11.3 &  1.21\\
 59.30 & --0.20 &  28.00 &    6.10 & 2.70 &   0.372 &knear &srby87 & 2.8 &  7.5 &   0.40 & 3.0 &  7.0 &   0.46 &  3.5 & 7.4 &  0.63\\
 59.90 &  1.50 & --65.00 &         &14.90 &   0.580 &kout  &  mk88 &15.4 & 13.3 &   0.62 &14.0 & 12.1 &   0.51 & 13.7 &11.9 &  0.49\\
 60.00 &  0.10 &  22.00 &  110.00 & 8.00 &   6.710 &kfar  &srby87 & 6.5 &  7.7 &   4.43 & 5.9 &  7.2 &   3.65 &  5.8 & 7.5 &  3.53\\
\hline
 60.90 & --0.10 &  23.00 &   13.80 & 7.50 &   0.842 &kfar  &srby87 & 6.1 &  7.7 &   0.56 & 5.4 &  7.1 &   0.44 &  5.2 & 7.5 &  0.40\\
 61.50 &  0.10 &  22.00 &   32.90 & 7.40 &   2.007 &kfar  &srby87 & 6.0 &  7.7 &   1.32 & 5.3 &  7.2 &   1.03 &  5.1 & 7.5 &  0.95\\
 63.10 &  0.40 &  20.00 &   21.20 & 7.00 &   1.293 &kfar  &srby87 & 5.7 &  7.8 &   0.86 & 5.0 &  7.3 &   0.66 &  4.7 & 7.6 &  0.58\\
 63.50 &  0.00 &  --9.70 &         & 8.49 &  38.019 &kout  & sod91 & 8.8 &  9.1 &  40.85 & 8.0 &  8.4 &  33.76 &  8.0 & 8.7 & 33.76\\
 65.50 &  1.40 & --72.20 &   29.00 &12.00 &   5.800 &kout  & dbt90 &14.9 & 13.7 &   8.94 &13.5 & 12.5 &   7.34 & 13.0 &12.2 &  6.81\\
 65.95 &  0.85 & --63.00 &    4.40 &12.00 &   0.900 &kout  & dbt90 &13.6 & 12.8 &   1.16 &12.3 & 11.6 &   0.95 & 12.0 &11.5 &  0.90\\
 66.45 &  1.75 & --60.90 &    7.30 &12.00 &   1.400 &kout  & dbt90 &13.2 & 12.5 &   1.69 &12.0 & 11.5 &   1.40 & 11.7 &11.4 &  1.33\\
 66.70 &  1.45 & --69.10 &    4.50 &12.00 &   0.900 &kout  & dbt90 &14.2 & 13.3 &   1.26 &12.8 & 12.2 &   1.02 & 12.4 &12.0 &  0.96\\
 67.00 &  1.15 & --68.00 &    3.30 &12.00 &   0.700 &kout  & dbt90 &14.0 & 13.2 &   0.95 &12.6 & 12.0 &   0.77 & 12.2 &11.9 &  0.72\\
 67.50 &  1.00 & --51.10 &    8.00 &12.00 &   1.600 &kout  & dbt90 &11.9 & 11.7 &   1.57 &10.8 & 10.7 &   1.30 & 10.5 &10.7 &  1.23\\
 68.16 &  0.90 & --63.90 &    4.90 &12.00 &   1.000 &kout  & dbt90 &13.2 & 12.8 &   1.21 &11.9 & 11.6 &   0.98 & 11.6 &11.5 &  0.93\\
 68.19 &  1.24 & --60.90 &   14.00 &12.00 &   2.900 &kout  & dbt90 &12.8 & 12.5 &   3.30 &11.6 & 11.4 &   2.71 & 11.3 &11.3 &  2.57\\
 68.20 &  0.90 & --63.40 &    2.50 &12.00 &   0.500 &kout  & dbt90 &13.1 & 12.7 &   0.60 &11.9 & 11.6 &   0.49 & 11.5 &11.5 &  0.46\\
 68.30 &  1.25 & --62.60 &   26.00 &12.00 &   5.200 &kout  & dbt90 &13.0 & 12.6 &   6.10 &11.8 & 11.5 &   5.03 & 11.4 &11.4 &  4.69\\
 68.37 &  1.26 & --64.00 &    9.30 &12.00 &   1.800 &kout  & dbt90 &13.1 & 12.8 &   2.15 &11.9 & 11.6 &   1.77 & 11.5 &11.5 &  1.65\\
 68.56 &  1.22 & --61.90 &    0.90 &12.00 &   0.200 &kout  & dbt90 &12.9 & 12.6 &   0.23 &11.6 & 11.5 &   0.19 & 11.3 &11.4 &  0.18\\
 69.00 &  0.80 & --62.30 &    6.10 &12.00 &   1.200 &kout  & dbt90 &12.8 & 12.6 &   1.37 &11.6 & 11.5 &   1.12 & 11.2 &11.4 &  1.05\\
 69.30 &  0.85 & --63.00 &    4.40 &12.00 &   0.900 &kout  & dbt90 &12.8 & 12.6 &   1.02 &11.6 & 11.5 &   0.84 & 11.2 &11.4 &  0.78\\
 69.44 &  1.39 & --67.70 &    2.20 &12.00 &   0.400 &kout  & dbt90 &13.4 & 13.1 &   0.50 &12.1 & 11.9 &   0.41 & 11.6 &11.8 &  0.37\\
 69.74 &  1.46 & --64.20 &   12.00 &12.00 &   2.300 &kout  & dbt90 &12.9 & 12.7 &   2.66 &11.6 & 11.6 &   2.15 & 11.2 &11.5 &  2.00\\
 69.83 &  1.27 & --60.60 &    4.20 &12.00 &   0.800 &kout  & dbt90 &12.4 & 12.4 &   0.85 &11.2 & 11.3 &   0.70 & 10.9 &11.2 &  0.66\\
 69.85 &  1.50 & --63.60 &   22.00 &12.00 &   4.400 &kout  & dbt90 &12.8 & 12.7 &   5.01 &11.5 & 11.6 &   4.04 & 11.1 &11.5 &  3.76\\
 69.85 &  1.01 & --57.30 &    1.00 &12.00 &   0.200 &kout  & dbt90 &12.0 & 12.1 &   0.20 &10.9 & 11.1 &   0.17 & 10.6 &11.0 &  0.16\\
 69.90 &  0.95 & --65.30 &   19.00 &12.00 &   3.800 &kout  & dbt90 &12.9 & 12.8 &   4.39 &11.7 & 11.7 &   3.61 & 11.3 &11.6 &  3.37\\
 69.92 &  1.50 & --63.80 &    4.80 &12.00 &   1.000 &kout  & dbt90 &12.8 & 12.7 &   1.14 &11.5 & 11.6 &   0.92 & 11.1 &11.5 &  0.86\\
 70.01 &  1.04 & --63.60 &    6.30 &12.00 &   1.200 &kout  & dbt90 &12.7 & 12.7 &   1.34 &11.5 & 11.6 &   1.10 & 11.1 &11.4 &  1.03\\
 70.02 &  1.17 & --67.10 &   11.00 &12.00 &   2.200 &kout  & dbt90 &13.1 & 13.0 &   2.62 &11.9 & 11.8 &   2.16 & 11.4 &11.7 &  1.99\\
 70.05 &  1.15 & --66.50 &   19.00 &12.00 &   3.800 &kout  & dbt90 &13.1 & 12.9 &   4.53 &11.8 & 11.8 &   3.67 & 11.4 &11.7 &  3.43\\
 70.16 &  1.17 & --66.90 &    6.10 &12.00 &   1.200 &kout  & dbt90 &13.1 & 13.0 &   1.43 &11.8 & 11.8 &   1.16 & 11.4 &11.7 &  1.08\\
 70.16 &  1.29 & --64.80 &    2.60 &12.00 &   0.500 &kout  & dbt90 &12.8 & 12.8 &   0.57 &11.6 & 11.6 &   0.47 & 11.2 &11.5 &  0.44\\
 70.30 &  1.12 & --65.70 &    7.10 &12.00 &   1.400 &kout  & dbt90 &12.9 & 12.8 &   1.62 &11.7 & 11.7 &   1.33 & 11.2 &11.6 &  1.22\\
 70.33 &  0.96 & --62.90 &    5.60 &12.00 &   1.100 &kout  & dbt90 &12.6 & 12.6 &   1.21 &11.4 & 11.5 &   0.99 & 11.0 &11.4 &  0.92\\
 70.40 &  0.50 & --70.80 &    6.80 &12.00 &   1.400 &kout  & dbt90 &13.5 & 13.3 &   1.77 &12.2 & 12.1 &   1.45 & 11.7 &11.9 &  1.33\\
 70.40 &  0.90 & --62.60 &   14.00 &12.00 &   2.700 &kout  & dbt90 &12.5 & 12.5 &   2.93 &11.3 & 11.5 &   2.39 & 10.9 &11.4 &  2.23\\
 70.53 &  1.15 & --59.20 &    4.80 &12.00 &   1.000 &kout  & dbt90 &12.1 & 12.2 &   1.02 &10.9 & 11.2 &   0.83 & 10.6 &11.1 &  0.78\\
 70.90 &  0.55 & --71.00 &   15.00 &12.00 &   2.900 &kout  & dbt90 &13.4 & 13.3 &   3.62 &12.1 & 12.1 &   2.95 & 11.6 &11.9 &  2.71\\
 71.10 & --0.40 &  10.00 &    5.20 & 1.40 &   0.317 &knear &srby87 & 1.1 &  8.2 &   0.20 & 1.6 &  7.6 &   0.41 &  2.7 & 8.0 &  1.18\\
 74.10 &  1.50 & --36.20 &         & 8.40 &   0.400 &kout  &  mk88 & 8.9 & 10.5 &   0.45 & 8.0 &  9.6 &   0.36 &  7.8 & 9.8 &  0.34\\
 75.00 &  1.90 & --79.20 &         &13.10 &   0.150 &kout  &  mk88 &13.6 & 14.0 &   0.16 &12.2 & 12.7 &   0.13 & 11.6 &12.5 &  0.12\\
 76.20 &  0.10 &  --2.00 &   72.00 & 5.10 &   4.392 &kout  &srby87 & 4.8 &  8.7 &   3.89 & 4.1 &  8.1 &   2.84 &  3.5 & 8.4 &  2.07\\
 76.30 & --0.70 &  --1.00 &   43.20 & 4.90 &   2.635 &kout  &srby87 & 4.7 &  8.7 &   2.42 & 3.9 &  8.0 &   1.67 &  3.3 & 8.4 &  1.20\\
 77.20 &  0.70 &   2.00 &   75.70 & 4.10 &   4.618 &kout  &srby87 & 4.0 &  8.5 &   4.40 &     &      &        &  1.9 & 8.3 &  0.99\\
 78.00 & --0.30 &  --4.00 &   19.80 & 4.80 &   1.208 &kout  &srby87 & 4.7 &  8.8 &   1.16 & 3.9 &  8.2 &   0.80 &  3.3 & 8.5 &  0.57\\
 78.10 & --0.80 &   0.00 &   29.70 & 4.10 &   1.812 &kout  &srby87 & 4.1 &  8.6 &   1.81 &     &      &        &  2.1 & 8.3 &  0.48\\
 78.20 &  0.10 &  11.00 &   11.30 & 2.00 &   0.689 &ktan  &srby87 & 1.7 &  8.3 &   0.50 & 1.6 &  7.8 &   0.44 &  1.7 & 8.2 &  0.50\\
 78.80 &  1.70 & --76.00 &         &11.80 &   0.700 &kout  &  mk88 &12.3 & 13.6 &   0.76 &11.1 & 12.3 &   0.62 & 10.5 &12.1 &  0.55\\
 78.80 & --0.50 &  --1.00 &   42.10 & 4.10 &   2.568 &kout  &srby87 & 4.0 &  8.7 &   2.44 & 3.3 &  8.0 &   1.66 &  2.3 & 8.4 &  0.81\\
 79.20 &  0.10 &   8.00 &    6.40 & 1.90 &   0.390 &ktan  &srby87 & 1.6 &  8.3 &   0.28 & 1.5 &  7.9 &   0.24 &  1.6 & 8.3 &  0.28\\
 79.60 & --0.60 &   4.00 &   12.10 & 2.50 &   0.738 &kfar  &srby87 & 2.9 &  8.5 &   0.99 & 1.9 &  7.9 &   0.43 &  1.5 & 8.4 &  0.27\\
 80.10 &  1.90 & --72.10 &         &11.20 &   0.200 &kout  &  mk88 &11.6 & 13.2 &   0.21 &10.4 & 12.0 &   0.17 &  9.8 &11.8 &  0.15\\
 80.70 &  0.70 &  --1.00 &   14.10 & 3.50 &   0.860 &kout  &srby87 & 3.6 &  8.7 &   0.91 & 2.8 &  8.0 &   0.55 &  1.4 & 8.4 &  0.14\\
 80.80 & --0.50 &  --3.00 &    3.60 & 3.80 &   0.220 &kout  &srby87 & 3.9 &  8.8 &   0.23 & 3.1 &  8.1 &   0.15 &  2.2 & 8.4 &  0.07\\
 81.20 &  1.00 &  13.00 &    9.00 & 1.50 &   0.549 &ktan  &srby87 & 1.3 &  8.4 &   0.41 & 1.2 &  7.9 &   0.35 &  1.3 & 8.3 &  0.41\\
 81.40 &  0.70 &  --2.00 &   10.30 & 3.50 &   0.628 &kout  &srby87 & 3.6 &  8.7 &   0.66 & 2.8 &  8.1 &   0.40 &  1.3 & 8.4 &  0.09\\
 81.40 &  0.00 &  --4.00 &    3.60 & 2.00 &   0.220 &kout  &srby87 & 3.9 &  8.8 &   0.84 & 3.2 &  8.1 &   0.56 &  2.3 & 8.5 &  0.29\\
 81.70 &  0.60 &  --2.00 &   42.90 & 2.00 &   2.617 &kout  &srby87 & 3.5 &  8.7 &   8.01 & 2.7 &  8.1 &   4.77 &  1.2 & 8.4 &  0.94\\
 82.00 & --0.40 &   5.00 &    4.10 & 1.40 &   0.250 &ktan  &srby87 &     &      &        & 1.1 &  7.9 &   0.15 &  1.2 & 8.3 &  0.18\\
 83.30 &  1.90 & --72.30 &         &10.60 &   0.200 &kout  &  mk88 &11.0 & 13.1 &   0.22 & 9.9 & 12.0 &   0.17 &  9.2 &11.8 &  0.15\\
 83.50 & --1.00 &   2.00 &    2.00 & 1.10 &   0.122 &kout  &srby87 & 2.3 &  8.5 &   0.53 &     &      &        &  0.9 & 8.4 &  0.08\\
 84.60 &  0.20 &  --1.00 &   48.60 & 2.00 &   2.965 &kout  &srby87 & 2.7 &  8.7 &   5.40 & 1.8 &  8.0 &   2.40 &  0.8 & 8.5 &  0.47\\
 84.70 &  1.70 & --83.00 &         &11.40 &   1.800 &kout  &  mk88 &12.2 & 14.2 &   2.06 &10.9 & 12.9 &   1.65 & 10.1 &12.6 &  1.41\\
 85.00 &  2.40 & --51.00 &         & 7.80 &   0.200 &kout  &  mk88 & 8.3 & 11.4 &   0.23 & 7.4 & 10.4 &   0.18 &  7.0 &10.5 &  0.16\\
 85.00 &  2.40 & --82.80 &         &11.50 &   0.300 &kout  &  mk88 &12.1 & 14.2 &   0.33 &10.8 & 12.9 &   0.26 & 10.0 &12.6 &  0.23\\
 85.40 &  0.00 & --38.00 &    6.10 & 7.10 &   0.372 &kout  &srby87 & 6.9 & 10.5 &   0.35 & 6.1 &  9.7 &   0.27 &  5.7 & 9.8 &  0.24\\
 86.00 &  0.50 & --16.80 &         & 4.18 &   5.495 &kout  & sod91 & 4.6 &  9.4 &   6.65 & 3.9 &  8.7 &   4.78 &  3.4 & 8.9 &  3.64\\
\hline
 88.40 &  1.90 & --63.70 &         & 8.60 &   0.200 &kout  &  mk88 & 9.2 & 12.3 &   0.23 & 8.2 & 11.3 &   0.18 &  7.6 &11.2 &  0.16\\
 88.80 &  1.90 & --60.60 &         & 8.30 &   0.120 &kout  &  mk88 & 8.8 & 12.1 &   0.13 & 7.8 & 11.0 &   0.11 &  7.2 &11.0 &  0.09\\
 89.90 &  1.30 & --71.00 &         & 9.20 &   0.630 &kout  &  mk88 & 9.8 & 12.9 &   0.71 & 8.7 & 11.8 &   0.56 &  8.0 &11.7 &  0.48\\
 91.00 & --0.50 & --24.60 &         & 4.25 &   7.413 &kout  & sod91 & 4.7 &  9.8 &   9.07 & 4.0 &  9.0 &   6.57 &  3.5 & 9.2 &  5.03\\
 91.00 & --1.00 & --36.30 &         & 5.46 &   8.710 &kout  & sod91 & 5.9 & 10.4 &  10.17 & 5.1 &  9.6 &   7.60 &  4.6 & 9.7 &  6.18\\
 92.90 &  1.90 & --87.70 &         &10.40 &   0.200 &kout  &  mk88 &11.5 & 14.7 &   0.24 &10.2 & 13.3 &   0.19 &  9.4 &13.0 &  0.16\\
 93.20 &  1.70 & --71.20 &         & 9.00 &   0.200 &kout  &  mk88 & 9.3 & 13.0 &   0.21 & 8.3 & 11.8 &   0.17 &  7.6 &11.7 &  0.14\\
103.00 &  2.00 & --18.30 &         & 2.31 &   3.802 &kout  & sod91 & 2.7 &  9.5 &   5.19 & 2.2 &  8.7 &   3.45 &  1.7 & 9.0 &  2.06\\
106.16 &  0.99 & --60.70 &    0.98 & 6.33 &   0.119 &kout  & hcs01 & 6.8 & 12.3 &   0.14 & 5.9 & 11.2 &   0.10 &  5.3 &11.2 &  0.08\\
106.21 &  1.47 & --22.38 &    0.66 & 2.50 &   0.080 &kout  & hcs01 & 2.9 &  9.7 &   0.11 & 2.4 &  8.9 &   0.07 &  1.9 & 9.2 &  0.05\\
106.27 & --0.01 & --56.36 &    3.20 & 5.86 &   0.387 &kout  & hcs01 & 6.3 & 11.9 &   0.45 & 5.5 & 10.9 &   0.34 &  4.9 &10.9 &  0.27\\
106.36 &  1.34 & --25.49 &    0.79 & 2.80 &   0.096 &kout  & hcs01 & 3.2 &  9.9 &   0.13 & 2.6 &  9.1 &   0.08 &  2.2 & 9.3 &  0.06\\
106.44 &  0.41 & --55.63 &    1.20 & 5.77 &   0.145 &kout  & hcs01 & 6.2 & 11.9 &   0.17 & 5.4 & 10.9 &   0.13 &  4.8 &10.9 &  0.10\\
106.48 &  0.90 & --54.17 &    2.70 & 5.61 &   0.327 &kout  & hcs01 & 6.1 & 11.8 &   0.39 & 5.3 & 10.8 &   0.29 &  4.7 &10.8 &  0.23\\
106.51 &  1.09 & --60.92 &    1.70 & 6.33 &   0.206 &kout  & hcs01 & 6.8 & 12.3 &   0.24 & 5.9 & 11.2 &   0.18 &  5.3 &11.2 &  0.14\\
106.59 &  1.01 & --61.62 &    2.90 & 6.40 &   0.351 &kout  & hcs01 & 6.9 & 12.4 &   0.41 & 6.0 & 11.3 &   0.31 &  5.4 &11.3 &  0.25\\
106.61 &  1.02 & --11.89 &    1.10 & 1.42 &   0.133 &kout  & hcs01 & 1.8 &  9.2 &   0.21 & 1.3 &  8.5 &   0.11 &  0.8 & 8.8 &  0.04\\
106.89 & --1.03 & --37.89 &    0.76 & 3.96 &   0.092 &kout  & hcs01 & 4.3 & 10.6 &   0.11 & 3.7 &  9.8 &   0.08 &  3.2 & 9.9 &  0.06\\
106.98 &  5.26 &  --8.53 &    1.60 & 1.05 &   0.194 &kout  & hcs01 & 1.4 &  9.0 &   0.34 & 1.0 &  8.3 &   0.18 &  0.5 & 8.7 &  0.04\\
107.16 &  0.67 & --52.78 &    1.40 & 5.42 &   0.169 &kout  & hcs01 & 5.9 & 11.7 &   0.20 & 5.1 & 10.7 &   0.15 &  4.5 &10.7 &  0.12\\
107.17 &  0.01 & --50.58 &    2.50 & 5.19 &   0.303 &kout  & hcs01 & 5.6 & 11.5 &   0.35 & 4.9 & 10.5 &   0.27 &  4.3 &10.6 &  0.21\\
107.17 &  4.52 &  --6.33 &    1.00 & 0.79 &   0.121 &kout  & hcs01 & 1.2 &  8.9 &   0.28 & 0.7 &  8.2 &   0.10 &  0.3 & 8.6 &  0.02\\
107.23 & --0.93 & --41.58 &    1.60 & 4.29 &   0.194 &kout  & hcs01 & 4.7 & 10.9 &   0.23 & 4.0 & 10.0 &   0.17 &  3.5 &10.1 &  0.13\\
107.36 &  0.15 & --50.66 &    0.81 & 5.19 &   0.098 &kout  & hcs01 & 5.6 & 11.5 &   0.11 & 4.9 & 10.5 &   0.09 &  4.3 &10.6 &  0.07\\
107.40 & --0.14 & --50.69 &    2.10 & 5.19 &   0.254 &kout  & hcs01 & 5.6 & 11.5 &   0.30 & 4.9 & 10.5 &   0.23 &  4.3 &10.6 &  0.17\\
107.44 &  0.50 & --50.76 &    0.72 & 5.19 &   0.087 &kout  & hcs01 & 5.6 & 11.5 &   0.10 & 4.9 & 10.5 &   0.08 &  4.3 &10.6 &  0.06\\
107.66 & --2.27 & --55.19 &    1.80 & 5.65 &   0.218 &kout  & hcs01 & 6.1 & 11.9 &   0.25 & 5.3 & 10.9 &   0.19 &  4.7 &10.9 &  0.15\\
107.81 &  0.74 & --52.20 &    0.75 & 5.32 &   0.091 &kout  & hcs01 & 5.8 & 11.6 &   0.11 & 5.0 & 10.7 &   0.08 &  4.4 &10.7 &  0.06\\
108.03 &  1.04 & --57.45 &    0.73 & 5.86 &   0.088 &kout  & hcs01 & 6.3 & 12.1 &   0.10 & 5.5 & 11.0 &   0.08 &  4.9 &11.0 &  0.06\\
108.36 &  0.43 & --47.77 &    0.76 & 4.83 &   0.092 &kout  & hcs01 & 5.3 & 11.3 &   0.11 & 4.5 & 10.4 &   0.08 &  4.0 &10.5 &  0.06\\
108.47 & --2.85 & --53.70 &    0.84 & 5.44 &   0.102 &kout  & hcs01 & 5.9 & 11.8 &   0.12 & 5.1 & 10.8 &   0.09 &  4.5 &10.8 &  0.07\\
108.54 & --1.02 & --51.98 &   31.00 & 5.25 &   3.751 &kout  & hcs01 & 5.7 & 11.6 &   4.42 & 4.9 & 10.7 &   3.27 &  4.3 &10.7 &  2.52\\
108.63 & --0.17 & --50.32 &    0.71 & 5.07 &   0.086 &kout  & hcs01 & 5.5 & 11.5 &   0.10 & 4.8 & 10.5 &   0.08 &  4.2 &10.6 &  0.06\\
108.64 &  0.73 & --54.88 &    3.40 & 5.54 &   0.411 &kout  & hcs01 & 6.0 & 11.9 &   0.48 & 5.2 & 10.9 &   0.36 &  4.6 &10.9 &  0.28\\
108.71 & --1.13 & --50.13 &    2.30 & 5.05 &   0.278 &kout  & hcs01 & 5.5 & 11.5 &   0.33 & 4.7 & 10.5 &   0.24 &  4.2 &10.6 &  0.19\\
108.73 & --2.63 & --55.66 &    3.40 & 5.63 &   0.411 &kout  & hcs01 & 6.1 & 11.9 &   0.48 & 5.3 & 10.9 &   0.36 &  4.6 &10.9 &  0.27\\
108.76 &  2.70 & --11.11 &    1.60 & 1.25 &   0.194 &kout  & hcs01 & 1.6 &  9.1 &   0.32 & 1.2 &  8.5 &   0.18 &  0.7 & 8.8 &  0.06\\
108.77 &  1.34 &  --7.58 &    0.66 & 0.88 &   0.080 &kout  & hcs01 & 1.2 &  9.0 &   0.15 & 0.8 &  8.3 &   0.07 &  0.3 & 8.6 &  0.01\\
108.78 & --0.83 & --49.71 &    1.50 & 5.00 &   0.182 &kout  & hcs01 & 5.4 & 11.5 &   0.21 & 4.7 & 10.5 &   0.16 &  4.1 &10.6 &  0.12\\
108.87 & --0.92 & --50.34 &    1.30 & 5.06 &   0.157 &kout  & hcs01 & 5.5 & 11.5 &   0.19 & 4.8 & 10.5 &   0.14 &  4.2 &10.6 &  0.11\\
108.89 &  0.32 & --52.61 &   50.00 & 5.29 &   6.050 &kout  & hcs01 & 5.7 & 11.7 &   7.02 & 5.0 & 10.7 &   5.40 &  4.4 &10.7 &  4.19\\
108.96 & --0.34 & --48.07 &   10.00 & 4.83 &   1.210 &kout  & hcs01 & 5.2 & 11.3 &   1.40 & 4.5 & 10.4 &   1.05 &  4.0 &10.5 &  0.83\\
108.97 &  0.31 & --56.09 &    1.10 & 5.65 &   0.133 &kout  & hcs01 & 6.1 & 12.0 &   0.16 & 5.3 & 10.9 &   0.12 &  4.7 &11.0 &  0.09\\
109.02 & --0.25 & --47.12 &    0.90 & 4.73 &   0.109 &kout  & hcs01 & 5.1 & 11.3 &   0.13 & 4.4 & 10.3 &   0.09 &  3.9 &10.4 &  0.07\\
109.11 & --0.86 & --48.82 &    2.60 & 4.89 &   0.315 &kout  & hcs01 & 5.3 & 11.4 &   0.37 & 4.6 & 10.5 &   0.28 &  4.0 &10.5 &  0.21\\
109.32 &  0.25 & --55.48 &    3.80 & 5.57 &   0.460 &kout  & hcs01 & 6.0 & 11.9 &   0.53 & 5.2 & 10.9 &   0.40 &  4.6 &10.9 &  0.31\\
109.35 & --0.37 & --56.85 &    1.00 & 5.71 &   0.121 &kout  & hcs01 & 6.2 & 12.0 &   0.14 & 5.4 & 11.0 &   0.11 &  4.7 &11.0 &  0.08\\
109.37 &  0.39 & --53.93 &    0.87 & 5.40 &   0.105 &kout  & hcs01 & 5.8 & 11.8 &   0.12 & 5.1 & 10.8 &   0.09 &  4.5 &10.8 &  0.07\\
109.50 &  6.50 &  --9.20 &         & 1.03 &   0.229 &kout  & sod91 & 1.4 &  9.1 &   0.42 & 1.0 &  8.4 &   0.22 &  0.6 & 8.7 &  0.08\\
109.56 &  1.62 & --52.63 &    1.50 & 5.26 &   0.182 &kout  & hcs01 & 5.7 & 11.7 &   0.21 & 4.9 & 10.7 &   0.16 &  4.4 &10.8 &  0.13\\
109.62 & --1.34 & --48.13 &    1.10 & 4.80 &   0.133 &kout  & hcs01 & 5.2 & 11.4 &   0.16 & 4.5 & 10.4 &   0.12 &  3.9 &10.5 &  0.09\\
109.64 &  1.94 & --56.86 &    7.60 & 5.70 &   0.920 &kout  & hcs01 & 6.2 & 12.1 &   1.09 & 5.4 & 11.0 &   0.83 &  4.7 &11.0 &  0.63\\
109.79 &  1.99 & --60.65 &    2.00 & 6.11 &   0.242 &kout  & hcs01 & 6.6 & 12.4 &   0.28 & 5.7 & 11.3 &   0.21 &  5.1 &11.3 &  0.17\\
109.86 & --1.29 & --34.33 &    1.60 & 3.43 &   0.194 &kout  & hcs01 & 3.8 & 10.4 &   0.24 & 3.2 &  9.6 &   0.17 &  2.7 & 9.8 &  0.12\\
109.88 &  2.85 & --81.29 &    1.30 & 8.63 &   0.157 &kout  & hcs01 & 9.2 & 14.5 &   0.18 & 8.1 & 13.2 &   0.14 &  7.2 &12.9 &  0.11\\
109.89 &  1.87 & --59.70 &    1.60 & 6.00 &   0.194 &kout  & hcs01 & 6.5 & 12.3 &   0.23 & 5.6 & 11.2 &   0.17 &  5.0 &11.2 &  0.13\\
109.89 &  3.05 & --65.17 &    1.40 & 6.62 &   0.169 &kout  & hcs01 & 7.1 & 12.8 &   0.19 & 6.2 & 11.7 &   0.15 &  5.5 &11.6 &  0.12\\
109.96 &  0.46 & --53.91 &    4.80 & 5.37 &   0.581 &kout  & hcs01 & 5.8 & 11.8 &   0.68 & 5.0 & 10.8 &   0.50 &  4.4 &10.9 &  0.39\\
109.97 & --0.24 & --52.70 &    1.40 & 5.24 &   0.169 &kout  & hcs01 & 5.7 & 11.7 &   0.20 & 4.9 & 10.7 &   0.15 &  4.3 &10.8 &  0.11\\
109.99 &  0.87 & --52.39 &    3.10 & 5.21 &   0.375 &kout  & hcs01 & 5.6 & 11.7 &   0.43 & 4.9 & 10.7 &   0.33 &  4.3 &10.8 &  0.26\\
110.00 &  0.00 & --52.10 &         & 5.18 &  91.201 &kout  & sod91 & 5.6 & 11.7 & 106.59 & 4.9 & 10.7 &  81.61 &  4.3 &10.7 & 62.85\\
110.05 & --0.17 & --50.46 &   66.00 & 5.00 &   7.986 &kout  & hcs01 & 5.4 & 11.5 &   9.31 & 4.7 & 10.6 &   7.06 &  4.1 &10.6 &  5.37\\
110.05 &  1.78 &  --5.57 &    0.59 & 0.63 &   0.071 &kout  & hcs01 & 1.0 &  8.9 &   0.18 & 0.6 &  8.2 &   0.06 &  0.1 & 8.5 &      \\
110.22 &  2.51 & --63.59 &    2.20 & 6.42 &   0.266 &kout  & hcs01 & 6.9 & 12.7 &   0.31 & 6.0 & 11.6 &   0.23 &  5.3 &11.5 &  0.18\\
110.24 &  0.03 & --52.32 &   10.00 & 5.19 &   1.210 &kout  & hcs01 & 5.6 & 11.7 &   1.41 & 4.9 & 10.7 &   1.08 &  4.3 &10.8 &  0.83\\
\hline
110.26 &  1.06 & --53.87 &    1.10 & 5.35 &   0.133 &kout  & hcs01 & 5.8 & 11.8 &   0.16 & 5.0 & 10.8 &   0.12 &  4.4 &10.9 &  0.09\\
110.28 & --0.94 & --39.17 &    1.10 & 3.87 &   0.133 &kout  & hcs01 & 4.3 & 10.7 &   0.16 & 3.6 &  9.9 &   0.12 &  3.1 &10.0 &  0.09\\
110.32 &  1.26 & --56.90 &    5.60 & 5.67 &   0.678 &kout  & hcs01 & 6.1 & 12.1 &   0.78 & 5.3 & 11.0 &   0.59 &  4.7 &11.0 &  0.47\\
110.32 &  3.45 & --71.96 &    1.10 & 7.41 &   0.133 &kout  & hcs01 & 7.9 & 13.5 &   0.15 & 7.0 & 12.3 &   0.12 &  6.2 &12.1 &  0.09\\
110.34 & --0.63 & --36.62 &    0.72 & 3.62 &   0.087 &kout  & hcs01 & 4.0 & 10.6 &   0.11 & 3.4 &  9.7 &   0.08 &  2.9 & 9.9 &  0.06\\
110.36 &  3.63 & --74.11 &    0.89 & 7.68 &   0.108 &kout  & hcs01 & 8.2 & 13.7 &   0.12 & 7.2 & 12.5 &   0.09 &  6.4 &12.3 &  0.08\\
110.40 & --0.51 & --45.50 &   72.00 & 4.49 &   8.712 &kout  & hcs01 & 4.9 & 11.2 &  10.38 & 4.2 & 10.3 &   7.62 &  3.7 &10.4 &  5.92\\
110.44 &  1.63 & --59.45 &    1.10 & 5.94 &   0.133 &kout  & hcs01 & 6.4 & 12.3 &   0.15 & 5.6 & 11.2 &   0.12 &  4.9 &11.2 &  0.09\\
110.45 & --1.65 & --48.80 &    1.10 & 4.82 &   0.133 &kout  & hcs01 & 5.2 & 11.4 &   0.15 & 4.5 & 10.5 &   0.12 &  3.9 &10.5 &  0.09\\
110.51 &  1.03 & --51.80 &    0.87 & 5.12 &   0.105 &kout  & hcs01 & 5.5 & 11.7 &   0.12 & 4.8 & 10.7 &   0.09 &  4.2 &10.7 &  0.07\\
110.56 &  0.18 & --45.90 &    0.86 & 4.52 &   0.104 &kout  & hcs01 & 4.9 & 11.2 &   0.12 & 4.2 & 10.3 &   0.09 &  3.7 &10.4 &  0.07\\
110.58 & --0.08 & --53.52 &    5.40 & 5.29 &   0.653 &kout  & hcs01 & 5.7 & 11.8 &   0.76 & 5.0 & 10.8 &   0.58 &  4.4 &10.8 &  0.45\\
110.60 & --2.16 & --37.41 &    1.40 & 3.69 &   0.169 &kout  & hcs01 & 4.1 & 10.6 &   0.21 & 3.5 &  9.8 &   0.15 &  2.9 & 9.9 &  0.10\\
110.64 & --0.92 & --48.08 &    3.10 & 4.73 &   0.375 &kout  & hcs01 & 5.1 & 11.4 &   0.44 & 4.4 & 10.4 &   0.32 &  3.9 &10.5 &  0.25\\
110.65 & --0.69 & --51.27 &   11.00 & 5.06 &   1.331 &kout  & hcs01 & 5.5 & 11.6 &   1.57 & 4.8 & 10.6 &   1.20 &  4.2 &10.7 &  0.92\\
110.66 & --0.33 & --52.02 &    4.20 & 5.13 &   0.508 &kout  & hcs01 & 5.6 & 11.7 &   0.61 & 4.8 & 10.7 &   0.44 &  4.2 &10.7 &  0.34\\
110.71 &  2.57 & --68.50 &    3.10 & 6.97 &   0.375 &kout  & hcs01 & 7.5 & 13.2 &   0.43 & 6.5 & 12.0 &   0.33 &  5.8 &11.9 &  0.26\\
110.72 & --0.60 & --49.34 &    2.90 & 4.86 &   0.351 &kout  & hcs01 & 5.3 & 11.5 &   0.42 & 4.6 & 10.5 &   0.31 &  4.0 &10.6 &  0.24\\
110.74 &  0.23 & --57.45 &    0.82 & 5.70 &   0.099 &kout  & hcs01 & 6.2 & 12.1 &   0.12 & 5.4 & 11.1 &   0.09 &  4.7 &11.1 &  0.07\\
110.77 & --2.71 & --34.75 &    5.60 & 3.43 &   0.678 &kout  & hcs01 & 3.8 & 10.5 &   0.83 & 3.2 &  9.6 &   0.59 &  2.7 & 9.8 &  0.42\\
110.77 &  1.89 & --52.98 &    0.76 & 5.23 &   0.092 &kout  & hcs01 & 5.7 & 11.8 &   0.11 & 4.9 & 10.8 &   0.08 &  4.3 &10.8 &  0.06\\
110.80 &  0.40 & --52.19 &    1.60 & 5.14 &   0.194 &kout  & hcs01 & 5.6 & 11.7 &   0.23 & 4.8 & 10.7 &   0.17 &  4.2 &10.8 &  0.13\\
110.88 &  1.88 & --10.46 &   13.00 & 1.11 &   1.573 &kout  & hcs01 & 1.4 &  9.1 &   2.50 & 1.0 &  8.4 &   1.28 &  0.6 & 8.7 &  0.46\\
110.89 & --0.29 & --55.63 &    1.40 & 5.50 &   0.169 &kout  & hcs01 & 5.9 & 12.0 &   0.19 & 5.2 & 11.0 &   0.15 &  4.5 &11.0 &  0.11\\
110.93 &  1.59 & --50.65 &    0.77 & 4.98 &   0.093 &kout  & hcs01 & 5.4 & 11.6 &   0.11 & 4.7 & 10.6 &   0.08 &  4.1 &10.7 &  0.06\\
110.94 & --0.04 & --52.05 &    0.79 & 5.12 &   0.096 &kout  & hcs01 & 5.6 & 11.7 &   0.11 & 4.8 & 10.7 &   0.08 &  4.2 &10.8 &  0.06\\
110.95 &  0.61 & --50.59 &    2.20 & 4.97 &   0.266 &kout  & hcs01 & 5.4 & 11.6 &   0.31 & 4.7 & 10.6 &   0.24 &  4.1 &10.7 &  0.18\\
110.98 & --2.86 & --37.11 &    5.80 & 3.65 &   0.702 &kout  & hcs01 & 4.0 & 10.6 &   0.84 & 3.4 &  9.8 &   0.61 &  2.9 & 9.9 &  0.44\\
110.99 & --0.99 & --45.10 &    0.70 & 4.42 &   0.085 &kout  & hcs01 & 4.8 & 11.2 &   0.10 & 4.2 & 10.2 &   0.08 &  3.6 &10.3 &  0.06\\
111.02 &  2.42 & --47.11 &    1.00 & 4.62 &   0.121 &kout  & hcs01 & 5.0 & 11.3 &   0.14 & 4.3 & 10.4 &   0.10 &  3.8 &10.5 &  0.08\\
111.03 &  3.01 & --40.30 &    0.77 & 3.95 &   0.093 &kout  & hcs01 & 4.3 & 10.8 &   0.11 & 3.7 & 10.0 &   0.08 &  3.2 &10.1 &  0.06\\
111.09 & --2.34 & --47.82 &    2.70 & 4.69 &   0.327 &kout  & hcs01 & 5.1 & 11.4 &   0.39 & 4.4 & 10.4 &   0.29 &  3.8 &10.5 &  0.21\\
111.17 & --1.03 & --43.10 &    1.50 & 4.21 &   0.182 &kout  & hcs01 & 4.6 & 11.0 &   0.22 & 4.0 & 10.1 &   0.16 &  3.4 &10.2 &  0.12\\
111.19 &  0.60 & --49.22 &    0.87 & 4.82 &   0.105 &kout  & hcs01 & 5.2 & 11.5 &   0.12 & 4.5 & 10.5 &   0.09 &  4.0 &10.6 &  0.07\\
111.20 & --2.25 & --52.54 &    0.91 & 5.17 &   0.110 &kout  & hcs01 & 5.6 & 11.7 &   0.13 & 4.9 & 10.8 &   0.10 &  4.2 &10.8 &  0.07\\
111.20 & --0.84 & --52.98 &    1.10 & 5.21 &   0.133 &kout  & hcs01 & 5.6 & 11.8 &   0.15 & 4.9 & 10.8 &   0.12 &  4.3 &10.8 &  0.09\\
111.20 &  3.21 & --71.61 &    1.20 & 7.33 &   0.145 &kout  & hcs01 & 7.9 & 13.5 &   0.17 & 6.9 & 12.3 &   0.13 &  6.1 &12.1 &  0.10\\
111.22 & --2.26 & --49.77 &    2.20 & 4.88 &   0.266 &kout  & hcs01 & 5.3 & 11.5 &   0.31 & 4.6 & 10.6 &   0.24 &  4.0 &10.6 &  0.18\\
111.23 &  1.00 & --57.96 &    0.81 & 5.73 &   0.098 &kout  & hcs01 & 6.2 & 12.2 &   0.11 & 5.4 & 11.1 &   0.09 &  4.8 &11.1 &  0.07\\
111.27 &  2.30 & --49.26 &    4.30 & 4.83 &   0.520 &kout  & hcs01 & 5.2 & 11.5 &   0.60 & 4.5 & 10.5 &   0.45 &  4.0 &10.6 &  0.36\\
111.28 & --1.44 & --39.84 &   15.00 & 3.89 &   1.815 &kout  & hcs01 & 4.3 & 10.8 &   2.22 & 3.7 &  9.9 &   1.64 &  3.1 &10.1 &  1.15\\
111.30 & --0.59 & --48.54 &    2.80 & 4.74 &   0.339 &kout  & hcs01 & 5.2 & 11.4 &   0.41 & 4.5 & 10.5 &   0.31 &  3.9 &10.5 &  0.23\\
111.32 & --0.68 & --45.00 &    9.70 & 4.39 &   1.174 &kout  & hcs01 & 4.8 & 11.2 &   1.40 & 4.1 & 10.2 &   1.02 &  3.6 &10.3 &  0.79\\
111.32 &  3.36 & --65.79 &    1.20 & 6.63 &   0.145 &kout  & hcs01 & 7.1 & 12.9 &   0.17 & 6.2 & 11.8 &   0.13 &  5.5 &11.7 &  0.10\\
111.37 &  0.46 & --50.58 &    3.60 & 4.95 &   0.436 &kout  & hcs01 & 5.4 & 11.6 &   0.52 & 4.7 & 10.6 &   0.39 &  4.1 &10.7 &  0.30\\
111.38 & --1.55 & --36.27 &    0.69 & 3.54 &   0.083 &kout  & hcs01 & 3.9 & 10.6 &   0.10 & 3.3 &  9.7 &   0.07 &  2.8 & 9.9 &  0.05\\
111.40 & --2.60 & --36.77 &    1.50 & 3.59 &   0.182 &kout  & hcs01 & 4.0 & 10.6 &   0.23 & 3.4 &  9.8 &   0.16 &  2.8 & 9.9 &  0.11\\
111.43 & --2.64 & --46.59 &    1.40 & 4.55 &   0.169 &kout  & hcs01 & 5.0 & 11.3 &   0.20 & 4.3 & 10.4 &   0.15 &  3.7 &10.4 &  0.11\\
111.44 &  1.13 & --49.67 &    1.00 & 4.85 &   0.121 &kout  & hcs01 & 5.3 & 11.5 &   0.14 & 4.6 & 10.6 &   0.11 &  4.0 &10.6 &  0.08\\
111.45 &  1.23 & --47.21 &    1.20 & 4.60 &   0.145 &kout  & hcs01 & 5.0 & 11.3 &   0.17 & 4.3 & 10.4 &   0.13 &  3.8 &10.5 &  0.10\\
111.49 &  0.73 & --53.81 &  200.00 & 5.28 &  24.200 &kout  & hcs01 & 5.7 & 11.9 &  28.20 & 5.0 & 10.8 &  21.70 &  4.4 &10.9 & 16.81\\
111.49 &  1.28 & --46.17 &    3.60 & 4.50 &   0.436 &kout  & hcs01 & 4.9 & 11.3 &   0.52 & 4.2 & 10.3 &   0.38 &  3.7 &10.4 &  0.29\\
111.50 & --3.00 & --41.70 &         & 4.07 &  27.542 &kout  & sod91 & 4.4 & 10.9 &  32.19 & 3.8 & 10.0 &  24.01 &  3.2 &10.2 & 17.03\\
111.50 &  2.00 & --10.50 &         & 1.10 &   5.370 &kout  & sod91 & 1.4 &  9.1 &   8.70 & 1.0 &  8.4 &   4.44 &  0.6 & 8.7 &  1.60\\
111.57 & --2.73 & --39.52 &   33.00 & 3.85 &   3.993 &kout  & hcs01 & 4.2 & 10.8 &   4.75 & 3.6 &  9.9 &   3.49 &  3.1 &10.0 &  2.59\\
111.61 & --2.69 & --37.91 &    1.90 & 3.69 &   0.230 &kout  & hcs01 & 4.1 & 10.7 &   0.28 & 3.5 &  9.8 &   0.21 &  2.9 &10.0 &  0.14\\
111.64 &  1.32 & --43.43 &    0.89 & 4.22 &   0.108 &kout  & hcs01 & 4.6 & 11.1 &   0.13 & 4.0 & 10.2 &   0.10 &  3.4 &10.3 &  0.07\\
111.67 & --2.29 & --47.11 &    5.00 & 4.59 &   0.605 &kout  & hcs01 & 5.0 & 11.3 &   0.72 & 4.3 & 10.4 &   0.53 &  3.7 &10.5 &  0.39\\
111.68 &  3.17 & --68.98 &    1.60 & 6.99 &   0.194 &kout  & hcs01 & 7.5 & 13.3 &   0.22 & 6.6 & 12.1 &   0.17 &  5.8 &11.9 &  0.13\\
111.70 &  0.01 & --30.14 &    1.10 & 2.94 &   0.133 &kout  & hcs01 & 3.3 & 10.2 &   0.17 & 2.8 &  9.4 &   0.12 &  2.3 & 9.6 &  0.08\\
111.70 &  2.61 & --56.17 &    3.00 & 5.53 &   0.363 &kout  & hcs01 & 6.0 & 12.1 &   0.43 & 5.2 & 11.0 &   0.32 &  4.6 &11.0 &  0.25\\
111.70 &  4.20 & --29.53 &    1.60 & 2.90 &   0.194 &kout  & hcs01 & 3.2 & 10.2 &   0.24 & 2.7 &  9.4 &   0.17 &  2.3 & 9.6 &  0.12\\
111.71 &  2.48 & --54.70 &    4.60 & 5.37 &   0.557 &kout  & hcs01 & 5.8 & 11.9 &   0.65 & 5.0 & 10.9 &   0.48 &  4.4 &11.0 &  0.37\\
111.84 &  1.87 & --51.99 &    0.83 & 5.08 &   0.100 &kout  & hcs01 & 5.5 & 11.7 &   0.12 & 4.8 & 10.7 &   0.09 &  4.2 &10.8 &  0.07\\
\hline
111.89 & --2.73 & --36.90 &    1.00 & 3.58 &   0.121 &kout  & hcs01 & 3.9 & 10.6 &   0.14 & 3.4 &  9.8 &   0.11 &  2.8 & 9.9 &  0.07\\
111.90 &  2.77 & --78.40 &    3.10 & 8.18 &   0.375 &kout  & hcs01 & 8.8 & 14.3 &   0.43 & 7.7 & 13.0 &   0.33 &  6.8 &12.7 &  0.26\\
111.92 &  0.93 & --50.11 &   17.00 & 4.87 &   2.057 &kout  & hcs01 & 5.3 & 11.6 &   2.44 & 4.6 & 10.6 &   1.84 &  4.0 &10.7 &  1.39\\
112.05 &  2.35 & --46.01 &    0.92 & 4.46 &   0.111 &kout  & hcs01 & 4.9 & 11.3 &   0.13 & 4.2 & 10.3 &   0.10 &  3.7 &10.4 &  0.08\\
112.07 &  2.14 & --52.67 &    0.73 & 5.14 &   0.088 &kout  & hcs01 & 5.6 & 11.8 &   0.10 & 4.8 & 10.8 &   0.08 &  4.2 &10.8 &  0.06\\
112.09 &  2.75 & --54.64 &    1.20 & 5.35 &   0.145 &kout  & hcs01 & 5.8 & 11.9 &   0.17 & 5.0 & 10.9 &   0.13 &  4.4 &11.0 &  0.10\\
112.13 &  2.39 & --78.74 &    1.10 & 8.21 &   0.133 &kout  & hcs01 & 8.8 & 14.4 &   0.15 & 7.7 & 13.0 &   0.12 &  6.8 &12.7 &  0.09\\
112.17 & --2.30 & --37.65 &   20.00 & 3.64 &   2.420 &kout  & hcs01 & 4.0 & 10.7 &   2.92 & 3.4 &  9.8 &   2.11 &  2.9 &10.0 &  1.54\\
112.18 & --1.86 & --45.14 &    6.50 & 4.36 &   0.787 &kout  & hcs01 & 4.8 & 11.2 &   0.95 & 4.1 & 10.3 &   0.70 &  3.5 &10.4 &  0.51\\
112.20 &  5.21 & --52.70 &    0.89 & 5.17 &   0.108 &kout  & hcs01 & 5.6 & 11.8 &   0.13 & 4.8 & 10.8 &   0.09 &  4.3 &10.9 &  0.07\\
112.22 & --2.54 & --44.53 &    0.77 & 4.31 &   0.093 &kout  & hcs01 & 4.7 & 11.2 &   0.11 & 4.0 & 10.2 &   0.08 &  3.5 &10.3 &  0.06\\
112.27 & --1.08 & --42.56 &    0.94 & 4.10 &   0.114 &kout  & hcs01 & 4.5 & 11.0 &   0.14 & 3.9 & 10.1 &   0.10 &  3.3 &10.2 &  0.07\\
112.31 &  2.50 & --78.14 &    0.77 & 8.13 &   0.093 &kout  & hcs01 & 8.7 & 14.3 &   0.11 & 7.6 & 13.0 &   0.08 &  6.7 &12.7 &  0.06\\
112.62 & --0.78 & --33.14 &    0.71 & 3.18 &   0.086 &kout  & hcs01 & 3.5 & 10.4 &   0.10 & 3.0 &  9.6 &   0.08 &  2.5 & 9.7 &  0.05\\
112.73 &  2.70 & --16.19 &    6.80 & 1.60 &   0.823 &kout  & hcs01 & 1.9 &  9.4 &   1.16 & 1.5 &  8.7 &   0.72 &  1.1 & 9.0 &  0.39\\
112.86 & --0.69 & --35.33 &    2.10 & 3.38 &   0.254 &kout  & hcs01 & 3.7 & 10.5 &   0.30 & 3.2 &  9.7 &   0.23 &  2.7 & 9.8 &  0.16\\
112.90 &  0.34 & --52.42 &    0.93 & 5.07 &   0.113 &kout  & hcs01 & 5.5 & 11.8 &   0.13 & 4.8 & 10.8 &   0.10 &  4.2 &10.8 &  0.08\\
112.99 & --2.18 & --34.98 &    1.20 & 3.34 &   0.145 &kout  & hcs01 & 3.7 & 10.5 &   0.18 & 3.1 &  9.7 &   0.12 &  2.6 & 9.8 &  0.09\\
113.03 & --1.52 & --30.52 &    0.99 & 2.92 &   0.120 &kout  & hcs01 & 3.3 & 10.2 &   0.15 & 2.7 &  9.4 &   0.10 &  2.2 & 9.6 &  0.07\\
113.46 &  0.94 & --59.37 &    0.87 & 5.80 &   0.105 &kout  & hcs01 & 6.3 & 12.4 &   0.12 & 5.4 & 11.3 &   0.09 &  4.8 &11.3 &  0.07\\
113.49 &  1.61 & --12.09 &    0.96 & 1.19 &   0.116 &kout  & hcs01 & 1.5 &  9.2 &   0.18 & 1.1 &  8.5 &   0.10 &  0.7 & 8.8 &  0.04\\
113.60 & --0.90 & --49.25 &    2.10 & 4.71 &   0.254 &kout  & hcs01 & 5.1 & 11.6 &   0.30 & 4.4 & 10.6 &   0.22 &  3.8 &10.6 &  0.17\\
113.70 & --0.62 & --43.19 &    1.80 & 4.10 &   0.218 &kout  & hcs01 & 4.5 & 11.1 &   0.26 & 3.9 & 10.2 &   0.20 &  3.3 &10.3 &  0.14\\
113.87 & --0.19 & --38.59 &    1.10 & 3.64 &   0.133 &kout  & hcs01 & 4.0 & 10.8 &   0.16 & 3.4 &  9.9 &   0.12 &  2.9 &10.0 &  0.08\\
114.09 & --0.58 & --36.19 &    1.00 & 3.41 &   0.121 &kout  & hcs01 & 3.8 & 10.6 &   0.15 & 3.2 &  9.8 &   0.11 &  2.7 & 9.9 &  0.08\\
114.12 & --1.15 & --32.95 &    0.69 & 3.10 &   0.083 &kout  & hcs01 & 3.5 & 10.4 &   0.11 & 2.9 &  9.6 &   0.07 &  2.4 & 9.7 &  0.05\\
114.16 &  2.04 & --65.99 &    1.30 & 6.54 &   0.157 &kout  & hcs01 & 7.0 & 13.1 &   0.18 & 6.1 & 11.9 &   0.14 &  5.4 &11.8 &  0.11\\
114.20 &  2.66 & --51.63 &    1.40 & 4.95 &   0.169 &kout  & hcs01 & 5.4 & 11.8 &   0.20 & 4.6 & 10.8 &   0.15 &  4.1 &10.8 &  0.12\\
114.21 & --0.98 & --31.87 &    0.91 & 3.00 &   0.110 &kout  & hcs01 & 3.3 & 10.3 &   0.13 & 2.8 &  9.5 &   0.10 &  2.3 & 9.7 &  0.06\\
114.24 &  0.33 & --53.87 &    1.50 & 5.17 &   0.182 &kout  & hcs01 & 5.6 & 12.0 &   0.21 & 4.9 & 10.9 &   0.16 &  4.2 &10.9 &  0.12\\
114.33 &  0.79 &--101.15 &         &12.35 &   0.140 &kout  &  bw94 &12.7 & 17.9 &   0.15 &11.1 & 16.1 &   0.11 &  9.6 &15.2 &  0.08\\
114.41 &  3.21 & --69.82 &    0.77 & 7.01 &   0.093 &kout  & hcs01 & 7.5 & 13.5 &   0.11 & 6.6 & 12.3 &   0.08 &  5.8 &12.1 &  0.06\\
114.47 & --0.89 & --46.94 &    0.77 & 4.44 &   0.093 &kout  & hcs01 & 4.8 & 11.4 &   0.11 & 4.2 & 10.4 &   0.08 &  3.6 &10.5 &  0.06\\
114.48 &  2.22 & --64.34 &    0.78 & 6.34 &   0.094 &kout  & hcs01 & 6.8 & 12.9 &   0.11 & 6.0 & 11.8 &   0.08 &  5.2 &11.7 &  0.06\\
114.54 &  2.88 & --57.11 &    0.87 & 5.52 &   0.105 &kout  & hcs01 & 6.0 & 12.2 &   0.12 & 5.2 & 11.2 &   0.09 &  4.6 &11.2 &  0.07\\
114.56 &  2.91 & --72.09 &    0.97 & 7.30 &   0.117 &kout  & hcs01 & 7.8 & 13.7 &   0.13 & 6.8 & 12.5 &   0.10 &  6.0 &12.3 &  0.08\\
114.57 & --0.19 & --37.56 &    1.30 & 3.52 &   0.157 &kout  & hcs01 & 3.9 & 10.7 &   0.19 & 3.3 &  9.8 &   0.14 &  2.8 &10.0 &  0.10\\
114.58 & --0.42 & --49.68 &   10.00 & 4.72 &   1.210 &kout  & hcs01 & 5.1 & 11.6 &   1.41 & 4.4 & 10.6 &   1.05 &  3.8 &10.7 &  0.78\\
114.89 &  3.81 & --22.32 &    0.92 & 2.10 &   0.111 &kout  & hcs01 & 2.4 &  9.8 &   0.14 & 2.0 &  9.0 &   0.10 &  1.6 & 9.3 &  0.06\\
114.93 &  3.18 & --64.42 &    1.10 & 6.35 &   0.133 &kout  & hcs01 & 6.8 & 13.0 &   0.15 & 6.0 & 11.8 &   0.12 &  5.2 &11.7 &  0.09\\
114.98 &  4.88 & --18.27 &    0.60 & 1.73 &   0.073 &kout  & hcs01 & 2.0 &  9.5 &   0.10 & 1.6 &  8.8 &   0.06 &  1.2 & 9.1 &  0.04\\
115.02 &  1.05 & --13.06 &    1.40 & 1.24 &   0.169 &kout  & hcs01 & 1.5 &  9.3 &   0.25 & 1.2 &  8.6 &   0.16 &  0.8 & 8.8 &  0.07\\
115.14 & --1.51 & --40.54 &    3.40 & 3.79 &   0.411 &kout  & hcs01 & 4.2 & 10.9 &   0.50 & 3.6 & 10.0 &   0.37 &  3.0 &10.2 &  0.26\\
115.26 &  3.62 & --60.99 &    1.00 & 5.95 &   0.121 &kout  & hcs01 & 6.4 & 12.6 &   0.14 & 5.6 & 11.5 &   0.11 &  4.9 &11.5 &  0.08\\
115.38 &  2.06 & --10.95 &    1.30 & 1.04 &   0.157 &kout  & hcs01 & 1.3 &  9.2 &   0.25 & 1.0 &  8.5 &   0.15 &  0.6 & 8.8 &  0.05\\
115.58 &  3.97 & --24.83 &    9.40 & 2.31 &   1.137 &kout  & hcs01 & 2.6 &  9.9 &   1.44 & 2.2 &  9.1 &   1.03 &  1.8 & 9.4 &  0.69\\
115.71 & --1.66 & --40.16 &    6.00 & 3.73 &   0.726 &kout  & hcs01 & 4.1 & 10.9 &   0.88 & 3.5 & 10.0 &   0.64 &  3.0 &10.1 &  0.47\\
115.82 &  3.81 & --67.34 &    1.10 & 6.69 &   0.133 &kout  & hcs01 & 7.2 & 13.3 &   0.15 & 6.3 & 12.1 &   0.12 &  5.5 &12.0 &  0.09\\
116.02 &  1.65 &  --8.08 &    1.10 & 0.77 &   0.133 &kout  & hcs01 & 1.1 &  9.0 &   0.27 & 0.7 &  8.3 &   0.11 &  0.3 & 8.6 &  0.02\\
116.07 &  4.19 & --68.23 &    0.77 & 6.81 &   0.093 &kout  & hcs01 & 7.3 & 13.4 &   0.11 & 6.4 & 12.2 &   0.08 &  5.6 &12.1 &  0.06\\
116.14 & --1.63 & --42.87 &    1.10 & 3.98 &   0.133 &kout  & hcs01 & 4.4 & 11.1 &   0.16 & 3.7 & 10.2 &   0.11 &  3.2 &10.3 &  0.09\\
116.16 & --0.67 & --45.04 &    1.10 & 4.19 &   0.133 &kout  & hcs01 & 4.6 & 11.3 &   0.16 & 3.9 & 10.4 &   0.12 &  3.4 &10.4 &  0.09\\
116.34 &  0.65 & --45.85 &    1.10 & 4.27 &   0.133 &kout  & hcs01 & 4.7 & 11.4 &   0.16 & 4.0 & 10.4 &   0.12 &  3.5 &10.5 &  0.09\\
116.58 & --2.97 & --36.61 &    1.10 & 3.37 &   0.133 &kout  & hcs01 & 3.7 & 10.7 &   0.16 & 3.2 &  9.8 &   0.12 &  2.6 &10.0 &  0.08\\
116.94 & --2.24 & --44.51 &    4.10 & 4.12 &   0.496 &kout  & hcs01 & 4.5 & 11.3 &   0.59 & 3.9 & 10.3 &   0.44 &  3.3 &10.4 &  0.32\\
117.10 & --2.42 & --37.92 &    0.80 & 3.47 &   0.097 &kout  & hcs01 & 3.8 & 10.8 &   0.12 & 3.3 &  9.9 &   0.09 &  2.7 &10.0 &  0.06\\
117.15 &  4.78 &  --7.47 &    4.70 & 0.70 &   0.569 &kout  & hcs01 & 1.0 &  9.0 &   1.16 & 0.7 &  8.3 &   0.57 &  0.3 & 8.6 &  0.10\\
117.19 &  5.06 & --73.29 &    1.20 & 7.48 &   0.145 &kout  & hcs01 & 8.0 & 14.1 &   0.17 & 7.0 & 12.8 &   0.13 &  6.1 &12.6 &  0.10\\
117.26 & --0.26 & --48.23 &    1.40 & 4.49 &   0.169 &kout  & hcs01 & 4.9 & 11.6 &   0.20 & 4.2 & 10.6 &   0.15 &  3.6 &10.7 &  0.11\\
117.43 &  1.12 & --61.56 &    1.30 & 5.94 &   0.157 &kout  & hcs01 & 6.4 & 12.8 &   0.18 & 5.6 & 11.7 &   0.14 &  4.9 &11.6 &  0.11\\
117.60 &  2.32 & --45.17 &    0.99 & 4.17 &   0.120 &kout  & hcs01 & 4.6 & 11.4 &   0.15 & 3.9 & 10.4 &   0.10 &  3.4 &10.5 &  0.08\\
117.61 &  0.00 & --46.51 &    1.80 & 4.30 &   0.218 &kout  & hcs01 & 4.7 & 11.5 &   0.26 & 4.0 & 10.5 &   0.19 &  3.5 &10.6 &  0.14\\
117.72 &  3.15 & --17.43 &    7.70 & 1.58 &   0.932 &kout  & hcs01 & 1.9 &  9.5 &   1.35 & 1.5 &  8.8 &   0.84 &  1.1 & 9.1 &  0.45\\
118.00 &  5.00 & --14.02 &    1.30 & 1.27 &   0.157 &kout  & hcs01 & 1.6 &  9.3 &   0.25 & 1.2 &  8.6 &   0.14 &  0.8 & 8.9 &  0.06\\
\hline
118.12 &  1.58 & --92.50 &    0.83 &10.41 &   0.100 &kout  & hcs01 &11.2 & 16.9 &   0.12 & 9.8 & 15.3 &   0.09 &  8.4 &14.5 &  0.07\\
118.46 &  1.20 & --61.22 &    1.00 & 5.89 &   0.121 &kout  & hcs01 & 6.4 & 12.8 &   0.14 & 5.5 & 11.7 &   0.11 &  4.8 &11.6 &  0.08\\
118.57 &  3.62 &  --5.59 &    0.77 & 0.51 &   0.093 &kout  & hcs01 & 0.8 &  8.9 &   0.23 & 0.5 &  8.2 &   0.09 &  0.1 & 8.6 &      \\
118.62 &  2.78 & --63.24 &    1.40 & 6.15 &   0.169 &kout  & hcs01 & 6.6 & 13.0 &   0.19 & 5.8 & 11.9 &   0.15 &  5.1 &11.8 &  0.12\\
118.66 & --0.38 & --50.16 &    0.84 & 4.66 &   0.102 &kout  & hcs01 & 5.1 & 11.8 &   0.12 & 4.4 & 10.8 &   0.09 &  3.8 &10.8 &  0.07\\
118.68 & --1.33 & --39.03 &    1.20 & 3.53 &   0.145 &kout  & hcs01 & 3.9 & 10.9 &   0.18 & 3.3 & 10.0 &   0.13 &  2.8 &10.1 &  0.09\\
118.73 &  2.90 & --67.77 &    0.92 & 6.71 &   0.111 &kout  & hcs01 & 7.2 & 13.6 &   0.13 & 6.3 & 12.3 &   0.10 &  5.5 &12.2 &  0.07\\
118.77 &  3.13 & --69.33 &    1.60 & 6.92 &   0.194 &kout  & hcs01 & 7.4 & 13.7 &   0.22 & 6.5 & 12.5 &   0.17 &  5.7 &12.3 &  0.13\\
118.90 &  3.19 & --20.17 &    1.90 & 1.79 &   0.230 &kout  & hcs01 & 2.1 &  9.7 &   0.32 & 1.7 &  8.9 &   0.21 &  1.3 & 9.2 &  0.12\\
119.18 & --0.89 & --32.94 &    0.78 & 2.94 &   0.094 &kout  & hcs01 & 3.3 & 10.5 &   0.12 & 2.8 &  9.7 &   0.09 &  2.3 & 9.8 &  0.06\\
119.22 &  1.71 & --62.46 &    1.40 & 6.04 &   0.169 &kout  & hcs01 & 6.5 & 13.0 &   0.20 & 5.7 & 11.9 &   0.15 &  5.0 &11.7 &  0.12\\
119.41 &  1.53 & --61.60 &    1.60 & 5.93 &   0.194 &kout  & hcs01 & 6.4 & 12.9 &   0.23 & 5.6 & 11.8 &   0.17 &  4.9 &11.7 &  0.13\\
119.42 &  0.74 & --53.88 &    1.70 & 5.05 &   0.206 &kout  & hcs01 & 5.5 & 12.2 &   0.24 & 4.7 & 11.1 &   0.18 &  4.1 &11.1 &  0.14\\
119.51 &  0.39 & --50.22 &    0.86 & 4.65 &   0.104 &kout  & hcs01 & 5.1 & 11.8 &   0.13 & 4.4 & 10.8 &   0.09 &  3.8 &10.9 &  0.07\\
119.58 &  3.22 & --46.78 &    2.10 & 4.30 &   0.254 &kout  & hcs01 & 4.7 & 11.6 &   0.30 & 4.0 & 10.6 &   0.22 &  3.5 &10.7 &  0.17\\
119.60 &  1.25 & --61.49 &    0.73 & 5.92 &   0.088 &kout  & hcs01 & 6.4 & 12.9 &   0.10 & 5.6 & 11.8 &   0.08 &  4.9 &11.7 &  0.06\\
119.64 & --0.50 & --50.36 &    1.60 & 4.66 &   0.194 &kout  & hcs01 & 5.1 & 11.9 &   0.23 & 4.4 & 10.9 &   0.17 &  3.8 &10.9 &  0.13\\
119.92 &  1.95 & --68.14 &    1.00 & 6.75 &   0.121 &kout  & hcs01 & 7.3 & 13.7 &   0.14 & 6.3 & 12.4 &   0.11 &  5.5 &12.2 &  0.08\\
120.12 & --0.18 & --49.32 &    1.20 & 4.54 &   0.145 &kout  & hcs01 & 5.0 & 11.8 &   0.18 & 4.3 & 10.8 &   0.13 &  3.7 &10.8 &  0.10\\
120.13 &  1.45 & --65.26 &    1.20 & 6.38 &   0.145 &kout  & hcs01 & 6.9 & 13.4 &   0.17 & 6.0 & 12.2 &   0.13 &  5.2 &12.0 &  0.10\\
120.13 &  3.46 & --68.99 &    8.00 & 6.89 &   0.968 &kout  & hcs01 & 7.4 & 13.8 &   1.12 & 6.5 & 12.6 &   0.86 &  5.7 &12.3 &  0.66\\
120.26 &  2.89 & --18.98 &    8.40 & 1.66 &   1.016 &kout  & hcs01 & 1.9 &  9.6 &   1.33 & 1.6 &  8.9 &   0.94 &  1.2 & 9.2 &  0.53\\
120.28 &  2.12 & --50.03 &    3.30 & 4.62 &   0.399 &kout  & hcs01 & 5.0 & 11.9 &   0.47 & 4.3 & 10.9 &   0.35 &  3.8 &10.9 &  0.27\\
120.30 &  2.85 & --67.89 &    1.00 & 6.74 &   0.121 &kout  & hcs01 & 7.3 & 13.7 &   0.14 & 6.3 & 12.5 &   0.11 &  5.5 &12.3 &  0.08\\
120.39 &  0.13 & --48.06 &    1.80 & 4.41 &   0.218 &kout  & hcs01 & 4.8 & 11.7 &   0.26 & 4.1 & 10.7 &   0.19 &  3.6 &10.8 &  0.15\\
120.42 &  3.63 & --71.72 &    0.89 & 7.26 &   0.108 &kout  & hcs01 & 7.8 & 14.2 &   0.12 & 6.8 & 12.9 &   0.09 &  6.0 &12.6 &  0.07\\
120.48 &  1.11 & --60.01 &    3.70 & 5.74 &   0.448 &kout  & hcs01 & 6.2 & 12.8 &   0.52 & 5.4 & 11.7 &   0.40 &  4.7 &11.6 &  0.30\\
120.49 &  1.78 & --62.67 &    4.90 & 6.07 &   0.593 &kout  & hcs01 & 6.6 & 13.1 &   0.70 & 5.7 & 11.9 &   0.52 &  5.0 &11.8 &  0.40\\
120.52 &  3.65 & --38.62 &    1.50 & 3.47 &   0.182 &kout  & hcs01 & 3.8 & 10.9 &   0.22 & 3.3 & 10.0 &   0.16 &  2.8 &10.2 &  0.12\\
120.70 &  4.20 & --11.55 &    1.10 & 1.01 &   0.133 &kout  & hcs01 & 1.3 &  9.2 &   0.22 & 0.9 &  8.5 &   0.11 &  0.6 & 8.8 &  0.05\\
120.77 &  4.98 &  --9.58 &    0.63 & 0.84 &   0.076 &kout  & hcs01 & 1.1 &  9.1 &   0.13 & 0.8 &  8.4 &   0.07 &  0.5 & 8.7 &  0.03\\
120.79 & --0.38 & --46.18 &    3.30 & 4.21 &   0.399 &kout  & hcs01 & 4.6 & 11.6 &   0.48 & 4.0 & 10.6 &   0.36 &  3.4 &10.6 &  0.26\\
120.80 &  1.10 & --63.70 &         & 6.20 &   1.100 &kout  &  mk88 & 6.7 & 13.2 &   1.28 & 5.8 & 12.1 &   0.96 &  5.1 &11.9 &  0.74\\
120.87 &  1.08 & --64.45 &    3.40 & 6.29 &   0.411 &kout  & hcs01 & 6.8 & 13.3 &   0.48 & 5.9 & 12.1 &   0.36 &  5.1 &12.0 &  0.27\\
120.93 &  1.15 & --62.64 &   11.00 & 6.06 &   1.331 &kout  & hcs01 & 6.6 & 13.1 &   1.58 & 5.7 & 12.0 &   1.18 &  5.0 &11.8 &  0.91\\
121.02 & --1.39 & --19.65 &    1.50 & 1.70 &   0.182 &kout  & hcs01 & 2.0 &  9.7 &   0.25 & 1.6 &  8.9 &   0.16 &  1.2 & 9.2 &  0.09\\
121.05 & --0.22 & --47.23 &    0.96 & 4.31 &   0.116 &kout  & hcs01 & 4.7 & 11.7 &   0.14 & 4.1 & 10.7 &   0.10 &  3.5 &10.7 &  0.08\\
121.46 &  0.78 & --59.76 &    1.20 & 5.71 &   0.145 &kout  & hcs01 & 6.2 & 12.9 &   0.17 & 5.4 & 11.7 &   0.13 &  4.7 &11.6 &  0.10\\
121.52 &  3.00 & --67.98 &    0.85 & 6.77 &   0.103 &kout  & hcs01 & 7.3 & 13.8 &   0.12 & 6.4 & 12.6 &   0.09 &  5.6 &12.3 &  0.07\\
121.54 &  3.85 & --40.17 &    0.78 & 3.60 &   0.094 &kout  & hcs01 & 4.0 & 11.1 &   0.12 & 3.4 & 10.2 &   0.08 &  2.9 &10.3 &  0.06\\
121.69 & --0.94 & --44.70 &    0.85 & 4.04 &   0.103 &kout  & hcs01 & 4.4 & 11.5 &   0.12 & 3.8 & 10.5 &   0.09 &  3.2 &10.6 &  0.06\\
121.96 &  4.10 & --67.94 &    1.70 & 6.79 &   0.206 &kout  & hcs01 & 7.3 & 13.8 &   0.24 & 6.4 & 12.6 &   0.18 &  5.6 &12.4 &  0.14\\
122.00 &  0.00 & --17.60 &         & 1.51 &   1.175 &kout  & sod91 & 1.8 &  9.6 &   1.67 & 1.4 &  8.8 &   1.01 &  1.0 & 9.1 &  0.52\\
122.00 &  3.68 &  --5.06 &    1.30 & 0.43 &   0.157 &kout  & hcs01 & 0.7 &  8.9 &   0.42 & 0.4 &  8.2 &   0.14 &  0.1 & 8.5 &  0.01\\
122.13 &  3.18 & --61.28 &    1.00 & 5.92 &   0.121 &kout  & hcs01 & 6.4 & 13.1 &   0.14 & 5.6 & 11.9 &   0.11 &  4.8 &11.8 &  0.08\\
122.14 & --1.37 & --13.85 &    1.40 & 1.18 &   0.169 &kout  & hcs01 & 1.5 &  9.4 &   0.27 & 1.1 &  8.6 &   0.15 &  0.7 & 8.9 &  0.06\\
122.18 &  3.00 & --60.12 &    0.71 & 5.78 &   0.086 &kout  & hcs01 & 6.3 & 13.0 &   0.10 & 5.4 & 11.8 &   0.08 &  4.7 &11.7 &  0.06\\
122.20 & --0.87 & --42.30 &    1.20 & 3.79 &   0.145 &kout  & hcs01 & 4.2 & 11.3 &   0.18 & 3.6 & 10.3 &   0.13 &  3.0 &10.4 &  0.09\\
122.22 &  2.70 & --60.53 &    1.10 & 5.83 &   0.133 &kout  & hcs01 & 6.3 & 13.0 &   0.16 & 5.5 & 11.9 &   0.12 &  4.8 &11.8 &  0.09\\
122.26 &  4.08 & --68.55 &    1.70 & 6.88 &   0.206 &kout  & hcs01 & 7.4 & 13.9 &   0.24 & 6.5 & 12.7 &   0.18 &  5.6 &12.5 &  0.14\\
122.45 &  2.62 & --55.17 &    1.20 & 5.19 &   0.145 &kout  & hcs01 & 5.6 & 12.5 &   0.17 & 4.9 & 11.4 &   0.13 &  4.2 &11.4 &  0.09\\
122.52 &  1.66 & --58.86 &    0.72 & 5.62 &   0.087 &kout  & hcs01 & 6.1 & 12.8 &   0.10 & 5.3 & 11.7 &   0.08 &  4.6 &11.6 &  0.06\\
122.53 &  1.55 & --57.22 &    2.10 & 5.42 &   0.254 &kout  & hcs01 & 5.9 & 12.7 &   0.30 & 5.1 & 11.6 &   0.22 &  4.4 &11.5 &  0.17\\
122.56 &  1.74 & --96.25 &    0.89 &11.60 &   0.108 &kout  & hcs01 &12.5 & 18.5 &   0.13 &10.9 & 16.6 &   0.10 &  9.2 &15.6 &  0.07\\
122.59 & --0.62 & --19.09 &   11.00 & 1.63 &   1.331 &kout  & hcs01 & 1.9 &  9.7 &   1.81 & 1.5 &  8.9 &   1.13 &  1.1 & 9.2 &  0.61\\
122.63 &  2.51 & --66.57 &    5.00 & 6.60 &   0.605 &kout  & hcs01 & 7.1 & 13.7 &   0.70 & 6.2 & 12.5 &   0.53 &  5.4 &12.3 &  0.41\\
122.80 &  2.89 & --63.95 &    2.40 & 6.27 &   0.290 &kout  & hcs01 & 6.8 & 13.4 &   0.34 & 5.9 & 12.2 &   0.26 &  5.1 &12.1 &  0.19\\
122.90 &  1.36 & --55.75 &    0.72 & 5.25 &   0.087 &kout  & hcs01 & 5.7 & 12.5 &   0.10 & 4.9 & 11.5 &   0.08 &  4.3 &11.4 &  0.06\\
122.96 &  3.14 & --10.27 &    1.40 & 0.87 &   0.169 &kout  & hcs01 & 1.1 &  9.2 &   0.27 & 0.8 &  8.5 &   0.14 &  0.5 & 8.8 &  0.06\\
123.06 & --6.31 & --30.40 &    1.10 & 2.20 &   0.133 &stel  & css90 &     &      &   0.13 &     &      &   0.13 &      &     &  0.13\\
123.06 & --0.78 & --44.77 &   21.00 & 4.04 &   2.541 &kout  & hcs01 & 4.4 & 11.5 &   3.01 & 3.8 & 10.6 &   2.25 &  3.2 &10.6 &  1.59\\
123.24 &  3.07 & --63.79 &    1.40 & 6.26 &   0.169 &kout  & hcs01 & 6.8 & 13.5 &   0.20 & 5.9 & 12.2 &   0.15 &  5.1 &12.1 &  0.11\\
123.45 &  3.21 & --61.50 &    9.20 & 5.97 &   1.113 &kout  & hcs01 & 6.5 & 13.2 &   1.32 & 5.6 & 12.0 &   0.98 &  4.9 &11.9 &  0.75\\
123.50 &  2.00 & --66.40 &         & 6.40 &   0.110 &kout  &  mk88 & 7.1 & 13.8 &   0.14 & 6.2 & 12.5 &   0.10 &  5.4 &12.3 &  0.08\\
123.62 & --1.36 & --21.23 &    0.96 & 1.80 &   0.116 &kout  & hcs01 & 2.1 &  9.8 &   0.16 & 1.7 &  9.0 &   0.10 &  1.3 & 9.3 &  0.06\\
\hline
123.68 &  2.86 & --63.72 &    1.80 & 6.26 &   0.218 &kout  & hcs01 & 6.8 & 13.5 &   0.26 & 5.9 & 12.3 &   0.19 &  5.1 &12.1 &  0.14\\
123.94 & --0.12 & --44.03 &    2.30 & 3.96 &   0.278 &kout  & hcs01 & 4.3 & 11.5 &   0.33 & 3.7 & 10.5 &   0.24 &  3.2 &10.6 &  0.18\\
124.01 &  1.93 & --51.53 &    0.78 & 4.78 &   0.094 &kout  & hcs01 & 5.2 & 12.2 &   0.11 & 4.5 & 11.1 &   0.08 &  3.9 &11.1 &  0.06\\
124.12 & --0.18 & --44.61 &    1.20 & 4.02 &   0.145 &kout  & hcs01 & 4.4 & 11.6 &   0.17 & 3.8 & 10.6 &   0.13 &  3.2 &10.7 &  0.09\\
124.26 & --0.17 & --46.45 &    0.87 & 4.21 &   0.105 &kout  & hcs01 & 4.6 & 11.7 &   0.13 & 4.0 & 10.7 &   0.09 &  3.4 &10.8 &  0.07\\
124.28 &  0.22 & --43.59 &    1.70 & 3.91 &   0.206 &kout  & hcs01 & 4.3 & 11.5 &   0.25 & 3.7 & 10.5 &   0.18 &  3.1 &10.6 &  0.13\\
124.33 &  0.50 & --46.07 &    1.70 & 4.17 &   0.206 &kout  & hcs01 & 4.6 & 11.7 &   0.25 & 3.9 & 10.7 &   0.18 &  3.4 &10.8 &  0.14\\
124.39 &  1.98 & --71.98 &    2.60 & 7.42 &   0.315 &kout  & hcs01 & 8.0 & 14.6 &   0.37 & 7.0 & 13.3 &   0.28 &  6.0 &12.9 &  0.21\\
124.60 &  2.54 & --86.91 &         &10.43 &   0.520 &kout  &  bw94 &10.7 & 17.1 &   0.55 & 9.4 & 15.4 &   0.42 &  8.0 &14.6 &  0.31\\
124.60 &  3.15 & --11.01 &    0.97 & 0.92 &   0.117 &kout  & hcs01 & 1.2 &  9.2 &   0.20 & 0.9 &  8.5 &   0.11 &  0.5 & 8.8 &  0.03\\
124.61 &  2.50 & --87.43 &    4.70 &10.07 &   0.569 &kout  & hcs01 &10.8 & 17.2 &   0.65 & 9.5 & 15.5 &   0.51 &  8.1 &14.7 &  0.37\\
124.78 &  2.93 & --61.85 &    1.10 & 6.05 &   0.133 &kout  & hcs01 & 6.5 & 13.4 &   0.15 & 5.7 & 12.2 &   0.12 &  4.9 &12.0 &  0.09\\
124.94 & --0.16 & --42.01 &    0.75 & 3.75 &   0.091 &kout  & hcs01 & 4.1 & 11.4 &   0.11 & 3.5 & 10.4 &   0.08 &  3.0 &10.5 &  0.06\\
125.08 &  0.78 & --67.75 &    2.50 & 6.84 &   0.303 &kout  & hcs01 & 7.4 & 14.1 &   0.35 & 6.4 & 12.8 &   0.27 &  5.6 &12.5 &  0.20\\
125.29 &  0.58 & --65.85 &    1.20 & 6.58 &   0.145 &kout  & hcs01 & 7.1 & 13.9 &   0.17 & 6.2 & 12.6 &   0.13 &  5.4 &12.4 &  0.10\\
125.58 &  2.07 & --53.78 &    4.90 & 5.06 &   0.593 &kout  & hcs01 & 5.5 & 12.5 &   0.70 & 4.7 & 11.4 &   0.51 &  4.1 &11.4 &  0.39\\
125.78 &  1.66 & --65.47 &    3.50 & 6.55 &   0.424 &kout  & hcs01 & 7.1 & 13.9 &   0.50 & 6.2 & 12.6 &   0.38 &  5.3 &12.4 &  0.28\\
126.03 &  4.37 &  --7.92 &    0.79 & 0.65 &   0.096 &kout  & hcs01 & 0.9 &  9.1 &   0.18 & 0.6 &  8.4 &   0.08 &  0.3 & 8.7 &  0.02\\
126.40 & --0.79 & --11.86 &    9.70 & 0.97 &   1.174 &kout  & hcs01 & 1.2 &  9.3 &   1.80 & 0.9 &  8.6 &   1.01 &  0.6 & 8.9 &  0.45\\
126.43 & --0.24 & --57.89 &    1.50 & 5.57 &   0.182 &kout  & hcs01 & 6.0 & 13.0 &   0.21 & 5.2 & 11.9 &   0.16 &  4.5 &11.8 &  0.12\\
127.00 & --1.00 & --11.80 &         & 0.96 &   4.365 &kout  & sod91 & 1.2 &  9.3 &   6.82 & 0.9 &  8.6 &   3.84 &  0.6 & 8.8 &  1.71\\
127.52 &  1.86 & --80.95 &    1.40 & 9.19 &   0.169 &kout  & hcs01 & 9.9 & 16.5 &   0.20 & 8.6 & 14.9 &   0.15 &  7.4 &14.3 &  0.11\\
127.70 & --0.11 & --49.78 &    0.82 & 4.62 &   0.099 &kout  & hcs01 & 5.0 & 12.3 &   0.12 & 4.3 & 11.2 &   0.09 &  3.7 &11.2 &  0.06\\
127.87 &  1.17 & --59.33 &    0.79 & 5.81 &   0.096 &kout  & hcs01 & 6.3 & 13.3 &   0.11 & 5.5 & 12.1 &   0.09 &  4.7 &12.0 &  0.06\\
128.30 &  1.75 & --48.24 &    0.94 & 4.47 &   0.114 &kout  & hcs01 & 4.9 & 12.1 &   0.14 & 4.2 & 11.1 &   0.10 &  3.6 &11.1 &  0.07\\
128.55 & --0.27 & --14.33 &    4.00 & 1.17 &   0.484 &kout  & hcs01 & 1.4 &  9.4 &   0.69 & 1.1 &  8.7 &   0.43 &  0.7 & 9.0 &  0.17\\
128.62 &  1.22 & --51.83 &    0.92 & 4.90 &   0.111 &kout  & hcs01 & 5.3 & 12.5 &   0.13 & 4.6 & 11.4 &   0.10 &  4.0 &11.4 &  0.07\\
128.66 &  1.45 & --54.30 &    0.88 & 5.20 &   0.106 &kout  & hcs01 & 5.7 & 12.8 &   0.13 & 4.9 & 11.7 &   0.09 &  4.2 &11.6 &  0.07\\
128.76 &  1.94 & --82.04 &    3.30 & 9.57 &   0.399 &kout  & hcs01 &10.3 & 17.0 &   0.46 & 9.0 & 15.3 &   0.35 &  7.7 &14.6 &  0.26\\
128.81 &  1.33 & --83.16 &    1.60 & 9.80 &   0.194 &kout  & hcs01 &10.6 & 17.2 &   0.23 & 9.2 & 15.5 &   0.17 &  7.8 &14.7 &  0.12\\
128.82 &  0.06 & --32.89 &    1.60 & 2.84 &   0.194 &kout  & hcs01 & 3.2 & 10.8 &   0.25 & 2.7 &  9.9 &   0.18 &  2.2 &10.0 &  0.12\\
128.99 & --0.16 & --36.78 &    0.90 & 3.23 &   0.109 &kout  & hcs01 & 3.6 & 11.1 &   0.14 & 3.0 & 10.2 &   0.09 &  2.5 &10.3 &  0.07\\
129.16 &  0.62 & --38.81 &    0.73 & 3.44 &   0.088 &kout  & hcs01 & 3.8 & 11.3 &   0.11 & 3.2 & 10.4 &   0.08 &  2.7 &10.4 &  0.05\\
129.20 & --0.37 & --37.55 &    1.00 & 3.31 &   0.121 &kout  & hcs01 & 3.7 & 11.2 &   0.15 & 3.1 & 10.3 &   0.11 &  2.6 &10.4 &  0.07\\
129.39 & --0.26 & --34.45 &    0.81 & 3.00 &   0.098 &kout  & hcs01 & 3.3 & 10.9 &   0.12 & 2.8 & 10.0 &   0.09 &  2.3 &10.2 &  0.06\\
130.07 &  5.08 & --14.09 &    0.64 & 1.15 &   0.077 &kout  & hcs01 & 1.4 &  9.5 &   0.11 & 1.1 &  8.7 &   0.07 &  0.8 & 9.0 &  0.04\\
130.59 &  1.83 & --44.61 &    3.10 & 4.11 &   0.375 &kout  & hcs01 & 4.5 & 11.9 &   0.45 & 3.9 & 10.9 &   0.34 &  3.3 &10.9 &  0.24\\
130.66 &  0.22 & --60.08 &    0.75 & 6.07 &   0.091 &kout  & hcs01 & 6.6 & 13.7 &   0.11 & 5.7 & 12.5 &   0.08 &  4.9 &12.3 &  0.06\\
131.79 & --0.70 & --53.83 &    2.30 & 5.29 &   0.278 &kout  & hcs01 & 5.8 & 13.1 &   0.33 & 5.0 & 11.9 &   0.25 &  4.2 &11.8 &  0.18\\
131.86 &  1.33 & --78.76 &    0.76 & 9.43 &   0.092 &kout  & hcs01 &10.2 & 17.1 &   0.11 & 8.9 & 15.4 &   0.08 &  7.5 &14.6 &  0.06\\
132.19 & --0.88 & --55.40 &    3.80 & 5.52 &   0.460 &kout  & hcs01 & 6.0 & 13.3 &   0.54 & 5.2 & 12.1 &   0.41 &  4.4 &11.9 &  0.29\\
132.54 & --1.62 & --37.55 &    0.84 & 3.38 &   0.102 &kout  & hcs01 & 3.7 & 11.4 &   0.12 & 3.2 & 10.4 &   0.09 &  2.7 &10.5 &  0.07\\
133.00 &  0.90 & --43.00 &    6.22 & 2.50 &   0.260 &stel  &dlp+96 &     &      &   0.26 &     &      &   0.26 &      &     &  0.26\\
133.29 &  0.81 & --48.12 &    0.89 & 4.63 &   0.108 &kout  & hcs01 & 5.1 & 12.5 &   0.13 & 4.4 & 11.4 &   0.10 &  3.7 &11.4 &  0.07\\
133.37 & --0.08 & --15.57 &    1.50 & 1.27 &   0.182 &kout  & hcs01 & 1.5 &  9.6 &   0.25 & 1.2 &  8.9 &   0.16 &  0.9 & 9.1 &  0.09\\
133.37 &  1.25 & --40.68 &    0.90 & 3.74 &   0.109 &kout  & hcs01 & 4.1 & 11.7 &   0.13 & 3.5 & 10.7 &   0.10 &  3.0 &10.8 &  0.07\\
133.39 &  1.01 & --42.43 &   44.00 & 3.95 &   5.324 &kout  & hcs01 & 4.3 & 11.9 &   6.31 & 3.7 & 10.9 &   4.67 &  3.2 &10.9 &  3.49\\
133.41 &  0.25 & --50.53 &   19.00 & 4.95 &   2.299 &kout  & hcs01 & 5.4 & 12.8 &   2.74 & 4.6 & 11.7 &   1.99 &  4.0 &11.6 &  1.50\\
133.47 &  0.02 & --48.27 &    1.80 & 4.66 &   0.218 &kout  & hcs01 & 5.1 & 12.6 &   0.26 & 4.4 & 11.5 &   0.19 &  3.7 &11.4 &  0.14\\
133.50 &  9.00 & --15.70 &         & 1.31 &   2.884 &kout  & sod91 & 1.5 &  9.6 &   3.78 & 1.2 &  8.9 &   2.42 &  1.0 & 9.2 &  1.68\\
133.67 &  0.62 & --44.55 &    2.80 & 4.21 &   0.339 &kout  & hcs01 & 4.6 & 12.1 &   0.40 & 4.0 & 11.1 &   0.31 &  3.4 &11.1 &  0.22\\
133.70 &  1.20 & --40.90 &    7.97 & 2.50 &   0.330 &stel  &dlp+96 &     &      &   0.33 &     &      &   0.33 &      &     &  0.33\\
133.79 &  1.26 & --39.31 &    1.50 & 3.60 &   0.182 &kout  & hcs01 & 4.0 & 11.6 &   0.22 & 3.4 & 10.6 &   0.16 &  2.9 &10.7 &  0.12\\
133.90 &  0.40 & --47.60 &    7.84 & 2.50 &   0.330 &stel  &dlp+96 &     &      &   0.33 &     &      &   0.33 &      &     &  0.33\\
133.90 &  1.31 & --45.17 &    0.86 & 4.30 &   0.104 &kout  & hcs01 & 4.7 & 12.2 &   0.12 & 4.0 & 11.2 &   0.09 &  3.5 &11.2 &  0.07\\
134.01 &  0.65 & --48.42 &   45.00 & 4.71 &   5.445 &kout  & hcs01 & 5.1 & 12.6 &   6.38 & 4.4 & 11.5 &   4.75 &  3.8 &11.5 &  3.54\\
134.01 &  0.60 & --46.59 &    1.20 & 4.48 &   0.145 &kout  & hcs01 & 4.9 & 12.4 &   0.17 & 4.2 & 11.3 &   0.13 &  3.6 &11.3 &  0.09\\
134.10 &  0.80 & --48.40 &    7.04 & 2.50 &   0.290 &stel  &dlp+96 &     &      &   0.29 &     &      &   0.29 &      &     &  0.29\\
134.29 & --1.85 & --49.09 &    1.60 & 4.82 &   0.194 &kout  & hcs01 & 5.3 & 12.7 &   0.23 & 4.5 & 11.6 &   0.17 &  3.8 &11.5 &  0.12\\
134.42 & --1.62 & --47.62 &    1.00 & 4.63 &   0.121 &kout  & hcs01 & 5.1 & 12.6 &   0.15 & 4.4 & 11.5 &   0.11 &  3.7 &11.4 &  0.08\\
134.45 & --0.14 & --13.81 &    1.80 & 1.12 &   0.218 &kout  & hcs01 & 1.4 &  9.5 &   0.34 & 1.1 &  8.8 &   0.21 &  0.7 & 9.0 &  0.09\\
134.48 &  2.20 & --30.92 &    0.68 & 2.72 &   0.082 &kout  & hcs01 & 3.0 & 10.8 &   0.10 & 2.6 & 10.0 &   0.07 &  2.1 &10.1 &  0.05\\
134.72 & --0.68 & --42.90 &    3.10 & 4.06 &   0.375 &kout  & hcs01 & 4.4 & 12.1 &   0.44 & 3.8 & 11.0 &   0.33 &  3.2 &11.0 &  0.23\\
134.84 &  1.41 & --40.94 &    0.76 & 3.83 &   0.092 &kout  & hcs01 & 4.2 & 11.8 &   0.11 & 3.6 & 10.8 &   0.08 &  3.1 &10.9 &  0.06\\
134.91 &  1.59 & --39.22 &    0.69 & 3.63 &   0.083 &kout  & hcs01 & 4.0 & 11.7 &   0.10 & 3.4 & 10.7 &   0.07 &  2.9 &10.7 &  0.05\\
\hline
135.16 & --0.06 & --45.02 &    2.10 & 4.34 &   0.254 &kout  & hcs01 & 4.7 & 12.3 &   0.30 & 4.1 & 11.3 &   0.23 &  3.5 &11.2 &  0.17\\
135.24 & --0.38 & --41.78 &    1.30 & 3.95 &   0.157 &kout  & hcs01 & 4.3 & 12.0 &   0.19 & 3.7 & 11.0 &   0.14 &  3.1 &11.0 &  0.10\\
135.27 &  1.06 & --45.42 &    1.90 & 4.40 &   0.230 &kout  & hcs01 & 4.8 & 12.4 &   0.27 & 4.1 & 11.3 &   0.20 &  3.5 &11.3 &  0.15\\
136.30 &  0.19 & --57.26 &    1.70 & 6.16 &   0.206 &kout  & hcs01 & 6.7 & 14.1 &   0.24 & 5.8 & 12.8 &   0.18 &  4.9 &12.5 &  0.13\\
136.34 & --1.72 & --10.78 &    1.20 & 0.87 &   0.145 &kout  & hcs01 & 1.1 &  9.3 &   0.23 & 0.8 &  8.6 &   0.12 &  0.5 & 8.9 &  0.05\\
136.34 &  0.85 & --74.35 &    1.30 & 9.44 &   0.157 &kout  & hcs01 &10.2 & 17.4 &   0.18 & 8.9 & 15.7 &   0.14 &  7.5 &14.8 &  0.10\\
136.50 &  1.20 & --36.40 &    4.40 & 2.50 &   0.180 &stel  &dlp+96 &     &      &   0.18 &     &      &   0.18 &      &     &  0.18\\
136.62 &  0.34 & --56.80 &    0.94 & 6.13 &   0.114 &kout  & hcs01 & 6.7 & 14.1 &   0.14 & 5.8 & 12.8 &   0.10 &  4.9 &12.5 &  0.07\\
136.70 &  1.23 & --36.88 &   15.00 & 3.44 &   1.815 &kout  & hcs01 & 3.8 & 11.6 &   2.21 & 3.2 & 10.6 &   1.57 &  2.7 &10.7 &  1.12\\
136.79 &  0.44 & --59.57 &    1.70 & 6.61 &   0.206 &kout  & hcs01 & 7.2 & 14.6 &   0.24 & 6.2 & 13.2 &   0.18 &  5.3 &12.9 &  0.13\\
136.88 &  0.26 & --56.82 &    1.30 & 6.16 &   0.157 &kout  & hcs01 & 6.7 & 14.1 &   0.19 & 5.8 & 12.9 &   0.14 &  4.9 &12.6 &  0.10\\
136.92 &  1.16 & --40.47 &    8.70 & 3.87 &   1.053 &kout  & hcs01 & 4.3 & 12.0 &   1.30 & 3.6 & 10.9 &   0.91 &  3.1 &11.0 &  0.68\\
137.00 &  1.30 & --38.50 &    9.19 & 2.50 &   0.380 &stel  &dlp+96 &     &      &   0.38 &     &      &   0.38 &      &     &  0.38\\
137.15 &  3.13 & --50.96 &    0.92 & 5.31 &   0.111 &kout  & hcs01 & 5.8 & 13.3 &   0.13 & 5.0 & 12.1 &   0.10 &  4.3 &12.0 &  0.07\\
137.22 &  1.36 & --40.13 &    0.92 & 3.85 &   0.111 &kout  & hcs01 & 4.2 & 12.0 &   0.13 & 3.6 & 10.9 &   0.10 &  3.1 &11.0 &  0.07\\
137.35 &  1.80 & --38.36 &    1.20 & 3.64 &   0.145 &kout  & hcs01 & 4.0 & 11.8 &   0.18 & 3.4 & 10.8 &   0.13 &  2.9 &10.8 &  0.09\\
137.70 &  1.50 & --38.90 &    3.86 & 2.50 &   0.160 &stel  &dlp+96 &     &      &   0.16 &     &      &   0.16 &      &     &  0.16\\
137.72 & --0.71 & --45.39 &    0.81 & 4.56 &   0.098 &kout  & hcs01 & 5.0 & 12.6 &   0.12 & 4.3 & 11.5 &   0.09 &  3.6 &11.4 &  0.06\\
137.72 &  4.18 &  --9.85 &    0.93 & 0.80 &   0.113 &kout  & hcs01 & 1.0 &  9.3 &   0.18 & 0.8 &  8.6 &   0.11 &  0.5 & 8.9 &  0.04\\
137.76 & --0.97 &--103.73 &    1.60 &21.60 &   0.194 &kout  & hcs01 &23.5 & 30.4 &   0.23 &20.3 & 26.8 &   0.17 &  15. & 22. &  0.09\\
137.95 &  1.72 & --42.72 &    0.83 & 4.22 &   0.100 &kout  & hcs01 & 4.6 & 12.3 &   0.12 & 4.0 & 11.3 &   0.09 &  3.4 &11.2 &  0.06\\
138.00 &  4.00 &  --9.80 &         & 0.80 &   0.170 &kout  & sod91 & 1.0 &  9.3 &   0.27 & 0.7 &  8.6 &   0.13 &  0.5 & 8.9 &  0.07\\
138.00 &  2.50 & --11.80 &         & 0.97 &   0.224 &kout  & sod91 & 1.2 &  9.4 &   0.34 & 0.9 &  8.7 &   0.19 &  0.6 & 9.0 &  0.09\\
138.29 &  2.42 & --12.60 &    1.20 & 1.04 &   0.145 &kout  & hcs01 & 1.3 &  9.5 &   0.23 & 1.0 &  8.8 &   0.13 &  0.7 & 9.0 &  0.07\\
138.38 & --2.14 & --46.63 &    1.20 & 4.79 &   0.145 &kout  & hcs01 & 5.2 & 12.9 &   0.17 & 4.5 & 11.7 &   0.13 &  3.8 &11.6 &  0.09\\
138.50 &  1.60 & --38.70 &    4.84 & 2.50 &   0.200 &stel  &dlp+96 &     &      &   0.20 &     &      &   0.20 &      &     &  0.20\\
138.52 & --0.81 & --45.75 &    2.50 & 4.67 &   0.303 &kout  & hcs01 & 5.1 & 12.8 &   0.36 & 4.4 & 11.7 &   0.27 &  3.7 &11.5 &  0.19\\
138.58 &  1.74 & --37.92 &    0.98 & 3.66 &   0.119 &kout  & hcs01 & 4.0 & 11.8 &   0.14 & 3.4 & 10.8 &   0.10 &  2.9 &10.9 &  0.07\\
138.66 &  5.16 & --15.38 &    1.20 & 1.30 &   0.145 &kout  & hcs01 & 1.5 &  9.7 &   0.19 & 1.2 &  9.0 &   0.12 &  0.9 & 9.2 &  0.07\\
139.20 &  1.00 & --39.60 &    1.71 & 2.50 &   0.071 &stel  &dlp+96 &     &      &   0.07 &     &      &   0.07 &      &     &  0.07\\
139.22 &  2.71 & --51.19 &    0.73 & 5.56 &   0.088 &kout  & hcs01 & 6.0 & 13.7 &   0.10 & 5.2 & 12.4 &   0.08 &  4.5 &12.2 &  0.06\\
139.24 & --2.57 & --13.87 &    0.79 & 1.16 &   0.096 &kout  & hcs01 & 1.4 &  9.6 &   0.14 & 1.1 &  8.9 &   0.09 &  0.8 & 9.1 &  0.05\\
139.24 &  1.01 & --39.86 &    1.90 & 3.94 &   0.230 &kout  & hcs01 & 4.3 & 12.1 &   0.27 & 3.7 & 11.1 &   0.20 &  3.1 &11.1 &  0.14\\
139.31 & --2.05 & --48.73 &    4.00 & 5.18 &   0.484 &kout  & hcs01 & 5.6 & 13.3 &   0.57 & 4.9 & 12.1 &   0.43 &  4.1 &11.9 &  0.30\\
139.36 &  2.87 & --53.02 &    1.00 & 5.87 &   0.121 &kout  & hcs01 & 6.4 & 14.0 &   0.14 & 5.5 & 12.7 &   0.11 &  4.7 &12.5 &  0.08\\
139.41 & --0.40 & --37.07 &    0.92 & 3.59 &   0.111 &kout  & hcs01 & 4.0 & 11.8 &   0.14 & 3.4 & 10.8 &   0.10 &  2.8 &10.8 &  0.07\\
139.42 & --1.50 & --49.59 &    0.97 & 5.32 &   0.117 &kout  & hcs01 & 5.8 & 13.4 &   0.14 & 5.0 & 12.2 &   0.10 &  4.2 &12.0 &  0.07\\
139.53 & --2.81 & --15.26 &    0.73 & 1.29 &   0.088 &kout  & hcs01 & 1.5 &  9.7 &   0.12 & 1.2 &  9.0 &   0.08 &  0.9 & 9.2 &  0.04\\
139.72 & --0.56 & --39.45 &    0.78 & 3.92 &   0.094 &kout  & hcs01 & 4.3 & 12.1 &   0.11 & 3.7 & 11.1 &   0.08 &  3.1 &11.0 &  0.06\\
139.90 &  0.22 & --40.16 &    2.20 & 4.02 &   0.266 &kout  & hcs01 & 4.4 & 12.2 &   0.32 & 3.8 & 11.2 &   0.24 &  3.2 &11.1 &  0.17\\
139.91 &  2.61 & --50.80 &    0.77 & 5.58 &   0.093 &kout  & hcs01 & 6.1 & 13.7 &   0.11 & 5.2 & 12.5 &   0.08 &  4.5 &12.3 &  0.06\\
140.27 &  2.11 & --12.63 &    3.30 & 1.06 &   0.399 &kout  & hcs01 & 1.3 &  9.5 &   0.60 & 1.0 &  8.8 &   0.36 &  0.7 & 9.1 &  0.17\\
140.27 &  1.52 & --15.42 &    0.66 & 1.31 &   0.080 &kout  & hcs01 & 1.5 &  9.7 &   0.10 & 1.2 &  9.0 &   0.07 &  0.9 & 9.2 &  0.04\\
140.27 &  3.48 &  --9.92 &    0.73 & 0.82 &   0.088 &kout  & hcs01 & 1.0 &  9.3 &   0.13 & 0.8 &  8.6 &   0.08 &  0.5 & 8.9 &  0.03\\
140.41 &  4.38 & --13.06 &    0.75 & 1.11 &   0.091 &kout  & hcs01 & 1.3 &  9.6 &   0.12 & 1.0 &  8.8 &   0.07 &  0.8 & 9.1 &  0.05\\
140.46 &  0.80 & --36.46 &    0.79 & 3.58 &   0.096 &kout  & hcs01 & 3.9 & 11.8 &   0.11 & 3.4 & 10.8 &   0.09 &  2.8 &10.8 &  0.06\\
140.60 & --0.10 & --39.20 &    3.22 & 2.50 &   0.140 &stel  &dlp+96 &     &      &   0.14 &     &      &   0.14 &      &     &  0.14\\
140.60 &  0.70 & --37.90 &    4.06 & 2.50 &   0.170 &stel  &dlp+96 &     &      &   0.17 &     &      &   0.17 &      &     &  0.17\\
140.68 &  4.94 & --14.22 &    0.80 & 1.22 &   0.097 &kout  & hcs01 & 1.4 &  9.7 &   0.13 & 1.1 &  8.9 &   0.08 &  0.9 & 9.2 &  0.05\\
140.73 & --0.98 & --38.56 &    0.96 & 3.87 &   0.116 &kout  & hcs01 & 4.3 & 12.1 &   0.14 & 3.6 & 11.1 &   0.10 &  3.1 &11.0 &  0.07\\
140.73 &  0.82 & --38.44 &    2.70 & 3.86 &   0.327 &kout  & hcs01 & 4.2 & 12.1 &   0.39 & 3.6 & 11.0 &   0.28 &  3.1 &11.0 &  0.21\\
140.77 & --0.16 & --41.09 &    0.71 & 4.21 &   0.086 &kout  & hcs01 & 4.6 & 12.4 &   0.10 & 4.0 & 11.3 &   0.08 &  3.3 &11.3 &  0.05\\
140.85 & --0.82 & --40.95 &    1.90 & 4.20 &   0.230 &kout  & hcs01 & 4.6 & 12.4 &   0.28 & 3.9 & 11.3 &   0.20 &  3.3 &11.3 &  0.14\\
140.90 & --1.10 & --40.40 &    5.61 & 2.50 &   0.240 &stel  &dlp+96 &     &      &   0.24 &     &      &   0.24 &      &     &  0.24\\
141.00 &  5.50 & --11.10 &         & 0.94 &   0.794 &kout  & sod91 & 1.1 &  9.4 &   1.09 & 0.9 &  8.7 &   0.73 &  0.6 & 9.0 &  0.32\\
141.46 & --1.18 & --40.79 &    4.60 & 4.23 &   0.557 &kout  & hcs01 & 4.6 & 12.5 &   0.66 & 4.0 & 11.4 &   0.50 &  3.3 &11.3 &  0.34\\
141.50 & --1.20 & --40.80 &    3.31 & 2.50 &   0.140 &stel  &dlp+96 &     &      &   0.14 &     &      &   0.14 &      &     &  0.14\\
142.00 &  1.00 & --15.70 &         & 1.37 &   4.467 &kout  & sod91 & 1.6 &  9.8 &   6.09 & 1.3 &  9.0 &   4.02 &  1.0 & 9.3 &  2.38\\
143.80 & --1.50 & --32.20 &    1.98 & 2.50 &   0.083 &stel  &dlp+96 &     &      &   0.08 &     &      &   0.08 &      &     &  0.08\\
145.20 &  2.99 & --58.39 &         & 8.59 &   0.150 &kout  &  bw94 & 8.9 & 16.6 &   0.16 & 7.7 & 15.0 &   0.12 &  6.5 &14.3 &  0.09\\
145.30 & --0.25 & --67.60 &    2.40 &11.00 &   0.120 &kout  & dgt94 &11.8 & 19.4 &   0.14 &10.2 & 17.4 &   0.10 &  8.3 &16.1 &  0.07\\
145.40 & --0.10 & --70.60 &    1.70 &12.00 &   0.087 &kout  & dgt94 &13.0 & 20.5 &   0.10 &11.2 & 18.4 &   0.08 &  9.1 &16.8 &  0.05\\
146.95 &  0.80 & --67.30 &    2.70 &12.00 &   0.140 &kout  & dgt94 &12.7 & 20.3 &   0.16 &11.0 & 18.2 &   0.12 &  8.9 &16.7 &  0.08\\
148.90 &  0.90 & --59.60 &    3.80 &10.00 &   0.200 &kout  & dgt94 &10.9 & 18.7 &   0.24 & 9.4 & 16.8 &   0.18 &  7.8 &15.7 &  0.12\\
\hline
150.15 &  1.10 & --55.90 &    2.10 &10.00 &   0.110 &kout  & dgt94 &10.3 & 18.2 &   0.12 & 8.9 & 16.3 &   0.09 &  7.4 &15.3 &  0.06\\
151.23 &  1.02 & --53.86 &         & 9.91 &   0.140 &kout  &  bw94 &10.2 & 18.1 &   0.15 & 8.8 & 16.3 &   0.11 &  7.3 &15.3 &  0.08\\
151.50 & --1.00 & --30.60 &         & 3.82 &  10.471 &kout  & sod91 & 4.2 & 12.4 &  12.66 & 3.6 & 11.3 &   9.30 &  3.0 &11.2 &  6.46\\
154.00 & --2.00 & --14.40 &         & 1.63 &   2.188 &kout  & sod91 & 1.9 & 10.2 &   2.97 & 1.5 &  9.4 &   1.85 &  1.2 & 9.6 &  1.19\\
154.35 &  2.61 & --35.80 &    7.40 & 6.00 &   0.895 &stel  & css90 &     &      &   0.89 &     &      &   0.89 &      &     &  0.89\\
154.64 &  2.45 & --38.70 &    3.00 & 6.00 &   0.363 &stel  & css90 &     &      &   0.36 &     &      &   0.36 &      &     &  0.36\\
156.89 &  0.21 & --47.44 &         &11.33 &   0.140 &kout  &  bw94 &11.7 & 19.8 &   0.15 &10.1 & 17.7 &   0.11 &  8.2 &16.3 &  0.07\\
173.71 &  2.70 & --17.40 &    7.60 & 1.80 &   0.920 &stel  & css90 & 1.8 & 10.3 &   0.92 & 1.8 &  9.8 &   0.92 &      &     &      \\
189.03 &  0.78 &   2.90 &    2.20 & 1.50 &   0.266 &stel  & css90 & 1.5 & 10.0 &   0.27 & 1.5 &  9.5 &   0.27 &      &     &      \\
190.05 &  0.53 &   8.40 &    0.88 & 1.50 &   0.106 &stel  & css90 & 1.5 & 10.0 &   0.11 & 1.5 &  9.5 &   0.11 &      &     &      \\
194.00 & --0.50 &  20.90 &         & 5.90 &   2.800 &kout  & mab97 & 6.1 & 14.5 &   2.99 & 5.3 & 13.2 &   2.26 &  4.5 &13.0 &  1.63\\
194.25 & --0.50 &  20.00 &         & 5.40 &   1.520 &kout  & mab97 & 5.5 & 13.9 &   1.58 & 4.8 & 12.7 &   1.20 &  4.1 &12.5 &  0.88\\
194.50 & --1.26 &  14.40 &         & 3.20 &   0.210 &kout  & mab97 & 3.4 & 11.8 &   0.24 & 2.9 & 10.8 &   0.17 &  2.5 &11.0 &  0.13\\
194.50 & --1.00 &  22.60 &         & 6.40 &   0.560 &kout  & mab97 & 6.6 & 15.0 &   0.60 & 5.7 & 13.6 &   0.44 &  4.9 &13.3 &  0.33\\
195.50 &  0.00 &  20.40 &         & 4.90 &   0.190 &kout  & mab97 & 5.1 & 13.4 &   0.21 & 4.3 & 12.2 &   0.15 &  3.7 &12.1 &  0.11\\
195.75 & --0.13 &  32.60 &         &11.00 &   1.920 &kout  & mab97 &11.3 & 19.6 &   2.03 & 9.8 & 17.6 &   1.52 &  8.1 &16.4 &  1.04\\
196.00 &  0.50 &  20.20 &         & 4.60 &   0.480 &kout  & mab97 & 4.8 & 13.2 &   0.52 & 4.1 & 12.0 &   0.38 &  3.5 &11.9 &  0.28\\
196.45 & --1.68 &  17.70 &    3.00 & 3.80 &   0.363 &stel  & css90 &     &      &   0.36 &     &      &   0.36 &      &     &  0.36\\
196.83 & --3.10 &  25.30 &    1.10 & 4.80 &   0.133 &stel  & css90 &     &      &   0.13 &     &      &   0.13 &      &     &  0.13\\
197.00 &  0.75 &  19.60 &         & 4.00 &   0.890 &kout  & mab97 & 4.2 & 12.6 &   0.98 & 3.6 & 11.5 &   0.72 &  3.1 &11.5 &  0.53\\
203.00 &  1.00 &  32.40 &         & 5.70 &   0.200 &kout  & mab97 & 5.9 & 14.1 &   0.21 & 5.1 & 12.8 &   0.16 &  4.3 &12.6 &  0.11\\
204.50 & --0.50 &  26.80 &         & 4.00 &   0.190 &kout  & mab97 & 4.1 & 12.4 &   0.20 & 3.5 & 11.3 &   0.15 &  3.0 &11.3 &  0.11\\
204.63 &  0.13 &  45.70 &         & 9.30 &   0.300 &kout  & mab97 & 9.6 & 17.7 &   0.32 & 8.3 & 15.9 &   0.24 &  7.0 &15.1 &  0.17\\
205.00 &  0.75 &  33.30 &         & 5.30 &   0.620 &kout  & mab97 & 5.5 & 13.7 &   0.67 & 4.7 & 12.5 &   0.49 &  4.0 &12.3 &  0.35\\
205.25 &  0.00 &  20.70 &         & 2.70 &   0.110 &kout  & mab97 & 2.9 & 11.2 &   0.13 & 2.4 & 10.3 &   0.09 &  2.1 &10.4 &  0.07\\
206.00 & --0.50 &  20.70 &         & 2.60 &   0.090 &kout  & mab97 & 2.8 & 11.1 &   0.10 & 2.4 & 10.2 &   0.08 &  2.0 &10.3 &  0.05\\
207.50 &  0.25 &  34.40 &         & 4.90 &   0.230 &kout  & mab97 & 5.1 & 13.3 &   0.25 & 4.4 & 12.1 &   0.19 &  3.8 &12.0 &  0.14\\
207.75 &  1.00 &  39.50 &         & 6.00 &   0.150 &kout  & mab97 & 6.3 & 14.3 &   0.17 & 5.4 & 13.0 &   0.12 &  4.6 &12.7 &  0.09\\
208.50 &  0.75 &  39.20 &         & 5.70 &   0.350 &kout  & mab97 & 6.0 & 14.0 &   0.39 & 5.1 & 12.8 &   0.28 &  4.4 &12.5 &  0.21\\
208.75 & --1.75 &  51.10 &         & 8.90 &   0.590 &kout  & mab97 & 9.2 & 17.2 &   0.63 & 8.0 & 15.5 &   0.48 &  6.8 &14.8 &  0.34\\
209.50 & --1.25 &  24.90 &         & 2.90 &   0.140 &kout  & mab97 & 3.1 & 11.3 &   0.16 & 2.6 & 10.4 &   0.11 &  2.3 &10.5 &  0.09\\
209.50 & --1.00 &  44.90 &         & 6.90 &   0.260 &kout  & mab97 & 7.1 & 15.1 &   0.28 & 6.1 & 13.7 &   0.20 &  5.2 &13.3 &  0.15\\
209.50 &  0.50 &  29.80 &         & 3.70 &   0.370 &kout  & mab97 & 3.9 & 12.1 &   0.41 & 3.3 & 11.0 &   0.29 &  2.8 &11.1 &  0.21\\
210.00 & --2.25 &  22.20 &         & 2.50 &   0.130 &kout  & mab97 & 2.7 & 10.9 &   0.15 & 2.3 & 10.0 &   0.11 &  1.9 &10.2 &  0.08\\
210.00 & --0.75 &  35.00 &         & 4.60 &   0.280 &kout  & mab97 & 4.8 & 12.9 &   0.30 & 4.1 & 11.7 &   0.22 &  3.5 &11.7 &  0.16\\
210.13 & --1.75 &  39.80 &         & 5.50 &   0.300 &kout  & mab97 & 5.7 & 13.7 &   0.32 & 4.9 & 12.5 &   0.24 &  4.2 &12.4 &  0.17\\
210.13 & --1.50 &  37.80 &         & 5.10 &   0.180 &kout  & mab97 & 5.3 & 13.4 &   0.19 & 4.6 & 12.2 &   0.15 &  3.9 &12.1 &  0.11\\
210.38 &  0.00 &  35.90 &         & 4.70 &   1.010 &kout  & mab97 & 4.9 & 12.9 &   1.10 & 4.2 & 11.8 &   0.81 &  3.6 &11.7 &  0.59\\
211.04 &  1.18 &  54.29 &         & 8.88 &   0.140 &kout  &  bw94 & 9.1 & 17.0 &   0.15 & 7.9 & 15.3 &   0.11 &  6.6 &14.6 &  0.08\\
211.25 & --1.75 &  42.20 &         & 5.80 &   0.350 &kout  & mab97 & 6.0 & 14.0 &   0.37 & 5.2 & 12.7 &   0.28 &  4.4 &12.5 &  0.20\\
211.25 & --1.00 &  43.80 &         & 6.10 &   0.180 &kout  & mab97 & 6.3 & 14.3 &   0.19 & 5.5 & 13.0 &   0.15 &  4.7 &12.7 &  0.11\\
211.25 & --0.50 &  21.70 &         & 2.40 &   0.090 &kout  & mab97 & 2.5 & 10.7 &   0.10 & 2.1 &  9.9 &   0.07 &  1.8 &10.1 &  0.05\\
211.50 &  1.00 &  45.40 &         & 6.40 &   1.190 &kout  & mab97 & 6.6 & 14.6 &   1.27 & 5.7 & 13.2 &   0.94 &  4.9 &12.9 &  0.70\\
212.25 & --1.00 &  27.50 &         & 3.10 &   0.150 &kout  & mab97 & 3.3 & 11.4 &   0.17 & 2.8 & 10.4 &   0.12 &  2.4 &10.6 &  0.09\\
212.25 & --1.00 &  44.30 &         & 6.00 &   1.110 &kout  & mab97 & 6.2 & 14.1 &   1.19 & 5.3 & 12.8 &   0.87 &  4.6 &12.6 &  0.65\\
213.00 &  1.25 &  42.90 &         & 5.50 &   1.040 &kout  & mab97 & 5.8 & 13.7 &   1.16 & 5.0 & 12.5 &   0.86 &  4.2 &12.3 &  0.61\\
213.25 & --2.00 &  37.10 &         & 4.50 &   0.210 &kout  & mab97 & 4.7 & 12.7 &   0.23 & 4.0 & 11.6 &   0.17 &  3.5 &11.6 &  0.13\\
214.00 &  0.75 &  44.10 &         & 5.60 &   0.940 &kout  & mab97 & 5.8 & 13.7 &   1.01 & 5.0 & 12.5 &   0.75 &  4.3 &12.3 &  0.55\\
214.50 & --1.75 &  27.60 &         & 2.90 &   0.390 &kout  & mab97 & 3.1 & 11.2 &   0.45 & 2.6 & 10.3 &   0.31 &  2.2 &10.4 &  0.22\\
214.50 &  1.00 &  46.00 &         & 5.90 &   1.130 &kout  & mab97 & 6.1 & 13.9 &   1.21 & 5.2 & 12.7 &   0.88 &  4.5 &12.4 &  0.66\\
215.00 &  1.00 &  47.60 &         & 6.10 &   2.480 &kout  & mab97 & 6.3 & 14.1 &   2.65 & 5.4 & 12.8 &   1.94 &  4.6 &12.6 &  1.41\\
215.25 & --0.25 &  29.20 &         & 3.10 &   0.400 &kout  & mab97 & 3.3 & 11.3 &   0.45 & 2.8 & 10.4 &   0.33 &  2.4 &10.5 &  0.24\\
215.50 & --2.25 &  43.40 &         & 5.20 &   0.100 &kout  & mab97 & 5.4 & 13.3 &   0.11 & 4.7 & 12.1 &   0.08 &  4.1 &12.0 &  0.06\\
215.50 &  0.25 &  48.70 &         & 6.20 &   0.520 &kout  & mab97 & 6.4 & 14.2 &   0.55 & 5.5 & 12.9 &   0.41 &  4.7 &12.7 &  0.30\\
215.75 &  1.25 &  50.70 &         & 6.50 &   0.720 &kout  & mab97 & 6.8 & 14.5 &   0.79 & 5.9 & 13.2 &   0.59 &  5.0 &12.9 &  0.43\\
216.00 &  0.50 &  45.70 &         & 5.50 &   0.180 &kout  & mab97 & 5.8 & 13.6 &   0.20 & 5.0 & 12.4 &   0.15 &  4.2 &12.2 &  0.10\\
216.25 &  0.00 &  28.20 &         & 2.90 &   0.100 &kout  & mab97 & 3.1 & 11.1 &   0.11 & 2.6 & 10.2 &   0.08 &  2.2 &10.3 &  0.06\\
216.50 &  0.50 &  46.70 &         & 5.70 &   0.130 &kout  & mab97 & 5.9 & 13.7 &   0.14 & 5.1 & 12.4 &   0.10 &  4.3 &12.3 &  0.07\\
216.50 &  1.25 &  48.00 &         & 5.90 &   0.580 &kout  & mab97 & 6.1 & 13.9 &   0.62 & 5.3 & 12.6 &   0.47 &  4.5 &12.4 &  0.34\\
216.50 & --2.50 &  22.80 &         & 2.14 &  12.303 &kout  & sod91 & 2.4 & 10.5 &  15.47 & 2.0 &  9.7 &  10.75 &  1.7 & 9.9 &  7.76\\
216.75 & --1.00 &  31.30 &         & 3.30 &   0.140 &kout  & mab97 & 3.4 & 11.4 &   0.15 & 2.9 & 10.5 &   0.11 &  2.5 &10.6 &  0.08\\
217.00 &  0.50 &  50.90 &         & 6.30 &   1.340 &kout  & mab97 & 6.6 & 14.3 &   1.47 & 5.7 & 13.0 &   1.10 &  4.8 &12.7 &  0.78\\
217.13 & --0.13 &  53.80 &         & 6.90 &   0.220 &kout  & mab97 & 7.1 & 14.8 &   0.23 & 6.2 & 13.4 &   0.18 &  5.3 &13.1 &  0.13\\
217.25 &  0.00 &  26.70 &         & 2.70 &   0.700 &kout  & mab97 & 2.8 & 10.9 &   0.75 & 2.4 & 10.0 &   0.55 &  2.0 &10.2 &  0.38\\
\hline
217.25 &  0.00 &  27.70 &         & 2.80 &   0.360 &kout  & mab97 & 3.0 & 11.0 &   0.41 & 2.5 & 10.1 &   0.29 &  2.1 &10.3 &  0.20\\
217.38 & --1.38 &  50.50 &         & 6.20 &   0.280 &kout  & mab97 & 6.4 & 14.2 &   0.30 & 5.5 & 12.9 &   0.22 &  4.8 &12.6 &  0.17\\
217.38 &  0.25 &  50.50 &         & 6.20 &   1.550 &kout  & mab97 & 6.4 & 14.1 &   1.65 & 5.5 & 12.9 &   1.22 &  4.8 &12.6 &  0.93\\
218.00 & --0.25 &  27.70 &         & 2.70 &   0.630 &kout  & mab97 & 2.9 & 11.0 &   0.73 & 2.5 & 10.1 &   0.54 &  2.1 &10.2 &  0.38\\
218.00 &  0.25 &  22.80 &         & 2.20 &   0.200 &kout  & mab97 & 2.4 & 10.5 &   0.24 & 2.0 &  9.6 &   0.17 &  1.6 & 9.8 &  0.11\\
218.25 & --2.00 &  26.30 &         & 2.60 &   0.360 &kout  & mab97 & 2.7 & 10.8 &   0.39 & 2.3 &  9.9 &   0.28 &  2.0 &10.1 &  0.21\\
218.25 & --2.00 &  32.30 &         & 3.30 &   0.720 &kout  & mab97 & 3.5 & 11.4 &   0.81 & 3.0 & 10.5 &   0.60 &  2.5 &10.6 &  0.41\\
218.75 & --0.50 &  56.10 &         & 7.00 &   0.170 &kout  & mab97 & 7.3 & 14.9 &   0.18 & 6.3 & 13.5 &   0.14 &  5.4 &13.1 &  0.10\\
218.75 & --0.50 &  59.60 &         & 7.80 &   0.220 &kout  & mab97 & 8.0 & 15.6 &   0.23 & 6.9 & 14.1 &   0.17 &  5.9 &13.6 &  0.13\\
219.00 & --1.88 &  31.30 &         & 3.10 &   0.290 &kout  & mab97 & 3.3 & 11.3 &   0.33 & 2.8 & 10.3 &   0.24 &  2.4 &10.5 &  0.17\\
219.75 & --2.00 &  30.90 &         & 3.00 &   0.210 &kout  & mab97 & 3.2 & 11.2 &   0.24 & 2.7 & 10.2 &   0.17 &  2.3 &10.4 &  0.12\\
220.25 & --2.00 &  33.80 &         & 3.40 &   0.310 &kout  & mab97 & 3.6 & 11.5 &   0.35 & 3.0 & 10.5 &   0.24 &  2.6 &10.6 &  0.18\\
221.75 &  0.25 &  63.80 &         & 8.00 &   0.200 &kout  & mab97 & 8.2 & 15.6 &   0.21 & 7.2 & 14.2 &   0.16 &  6.1 &13.7 &  0.12\\
222.50 & --0.25 &  51.30 &         & 5.60 &   0.160 &kout  & mab97 & 5.9 & 13.4 &   0.18 & 5.1 & 12.2 &   0.13 &  4.3 &12.1 &  0.09\\
222.75 & --2.00 &  39.70 &         & 4.00 &   0.290 &kout  & mab97 & 4.2 & 11.9 &   0.32 & 3.6 & 10.9 &   0.23 &  3.1 &11.0 &  0.17\\
223.75 & --2.00 &  63.20 &         & 7.50 &   0.190 &kout  & mab97 & 7.8 & 15.1 &   0.21 & 6.7 & 13.7 &   0.15 &  5.8 &13.3 &  0.11\\
224.00 & --2.00 &  17.60 &         & 1.47 &   4.677 &kout  & sod91 & 1.7 &  9.8 &   6.26 & 1.4 &  9.0 &   4.24 &  1.1 & 9.3 &  2.62\\
228.00 & --1.00 &  36.50 &         & 3.40 &   0.110 &kout  & mab97 & 3.6 & 11.2 &   0.12 & 3.1 & 10.3 &   0.09 &  2.6 &10.4 &  0.06\\
229.75 & --2.50 &  41.30 &         & 3.90 &   0.120 &kout  & mab97 & 4.1 & 11.6 &   0.13 & 3.5 & 10.6 &   0.10 &  3.0 &10.7 &  0.07\\
230.00 & --1.00 &  39.20 &         & 3.70 &   0.230 &kout  & mab97 & 3.9 & 11.4 &   0.26 & 3.3 & 10.4 &   0.18 &  2.8 &10.5 &  0.13\\
230.25 & --2.75 &  42.00 &         & 4.00 &   0.380 &kout  & mab97 & 4.2 & 11.6 &   0.42 & 3.6 & 10.6 &   0.31 &  3.1 &10.7 &  0.23\\
231.13 &  0.00 &  43.40 &         & 4.10 &   0.210 &kout  & mab97 & 4.3 & 11.7 &   0.23 & 3.7 & 10.7 &   0.17 &  3.2 &10.8 &  0.13\\
231.50 & --1.00 &  43.10 &         & 4.10 &   0.910 &kout  & mab97 & 4.3 & 11.6 &   1.00 & 3.7 & 10.7 &   0.74 &  3.2 &10.7 &  0.55\\
232.00 &  0.13 &  57.20 &         & 5.80 &   0.360 &kout  & mab97 & 6.0 & 13.1 &   0.39 & 5.2 & 11.9 &   0.29 &  4.5 &11.8 &  0.22\\
232.00 & --2.50 &  42.30 &         & 3.81 &  14.454 &kout  & sod91 & 4.2 & 11.6 &  17.56 & 3.6 & 10.6 &  12.90 &  3.1 &10.7 &  9.57\\
232.25 & --1.00 &  44.60 &         & 4.20 &   0.370 &kout  & mab97 & 4.4 & 11.7 &   0.41 & 3.8 & 10.8 &   0.30 &  3.3 &10.8 &  0.23\\
232.75 & --3.25 &  40.60 &         & 3.80 &   0.790 &kout  & mab97 & 4.0 & 11.4 &   0.88 & 3.4 & 10.4 &   0.63 &  2.9 &10.5 &  0.46\\
233.00 & --1.00 &  47.00 &         & 4.50 &   0.410 &kout  & mab97 & 4.7 & 11.9 &   0.45 & 4.0 & 10.9 &   0.32 &  3.5 &11.0 &  0.25\\
233.00 & --0.75 &  45.60 &         & 4.30 &   1.430 &kout  & mab97 & 4.5 & 11.8 &   1.57 & 3.9 & 10.8 &   1.18 &  3.4 &10.9 &  0.89\\
233.00 &  0.00 &  18.90 &         & 1.57 &   1.549 &kout  & sod91 & 1.8 &  9.7 &   2.04 & 1.5 &  9.0 &   1.41 &  1.1 & 9.2 &  0.76\\
233.25 & --1.75 &  51.60 &         & 5.00 &   0.160 &kout  & mab97 & 5.3 & 12.4 &   0.18 & 4.5 & 11.3 &   0.13 &  3.9 &11.3 &  0.10\\
234.00 & --0.25 &  43.70 &         & 4.10 &   1.930 &kout  & mab97 & 4.3 & 11.6 &   2.12 & 3.7 & 10.6 &   1.57 &  3.2 &10.7 &  1.18\\
234.25 & --0.88 &  76.30 &         & 8.60 &   0.930 &kout  & mab97 & 8.8 & 15.4 &   0.97 & 7.7 & 14.0 &   0.75 &  6.7 &13.5 &  0.56\\
234.50 & --0.25 &  43.40 &         & 4.10 &   1.540 &kout  & mab97 & 4.3 & 11.5 &   1.69 & 3.7 & 10.6 &   1.25 &  3.2 &10.6 &  0.94\\
235.00 &  0.50 &  47.90 &         & 4.60 &   0.160 &kout  & mab97 & 4.8 & 11.9 &   0.17 & 4.1 & 10.9 &   0.13 &  3.6 &10.9 &  0.10\\
235.38 & --1.75 &  80.80 &         & 9.30 &   0.950 &kout  & mab97 & 9.5 & 16.0 &   0.99 & 8.3 & 14.4 &   0.76 &  7.2 &13.9 &  0.57\\
235.50 &  0.75 &  48.30 &         & 4.60 &   0.160 &kout  & mab97 & 4.8 & 11.9 &   0.17 & 4.2 & 10.9 &   0.13 &  3.6 &10.9 &  0.10\\
235.75 & --1.50 &  52.60 &         & 5.10 &   0.190 &kout  & mab97 & 5.3 & 12.3 &   0.21 & 4.6 & 11.3 &   0.15 &  4.0 &11.3 &  0.12\\
236.38 & --1.75 &  77.60 &         & 8.60 &   1.230 &kout  & mab97 & 8.9 & 15.3 &   1.32 & 7.7 & 13.9 &   0.99 &  6.7 &13.4 &  0.75\\
236.38 & --1.50 &  29.20 &         & 2.60 &   0.200 &kout  & mab97 & 2.8 & 10.3 &   0.23 & 2.4 &  9.5 &   0.17 &  2.0 & 9.7 &  0.12\\
236.75 & --0.50 &  56.60 &         & 5.60 &   0.130 &kout  & mab97 & 5.8 & 12.7 &   0.14 & 5.0 & 11.5 &   0.10 &  4.4 &11.5 &  0.08\\
237.00 & --2.00 &  23.20 &         & 2.10 &   0.190 &kout  & mab97 & 2.3 &  9.9 &   0.23 & 1.9 &  9.1 &   0.16 &  1.5 & 9.4 &  0.10\\
237.13 & --2.38 &  57.10 &         & 5.60 &   0.260 &kout  & mab97 & 5.9 & 12.7 &   0.29 & 5.1 & 11.6 &   0.22 &  4.5 &11.5 &  0.17\\
237.25 & --1.25 &  74.40 &         & 8.00 &   1.050 &kout  & mab97 & 8.3 & 14.7 &   1.13 & 7.2 & 13.4 &   0.85 &  6.3 &13.0 &  0.65\\
237.50 & --0.25 &  55.90 &         & 5.50 &   0.180 &kout  & mab97 & 5.7 & 12.5 &   0.19 & 4.9 & 11.4 &   0.14 &  4.3 &11.4 &  0.11\\
237.75 & --1.00 &  22.40 &         & 2.00 &   0.260 &kout  & mab97 & 2.2 &  9.9 &   0.31 & 1.8 &  9.1 &   0.21 &  1.4 & 9.3 &  0.13\\
238.38 & --3.25 &  65.20 &         & 6.70 &   0.490 &kout  & mab97 & 6.9 & 13.5 &   0.52 & 6.0 & 12.3 &   0.39 &  5.3 &12.1 &  0.31\\
239.13 & --3.50 &  23.00 &         & 2.10 &   0.190 &kout  & mab97 & 2.3 &  9.9 &   0.23 & 1.9 &  9.1 &   0.16 &  1.5 & 9.4 &  0.10\\
239.13 & --1.88 &  83.00 &         & 9.40 &   2.460 &kout  & mab97 & 9.6 & 15.8 &   2.57 & 8.4 & 14.3 &   1.96 &  7.3 &13.8 &  1.48\\
239.50 & --1.88 &  79.90 &         & 8.80 &   0.410 &kout  & mab97 & 9.1 & 15.3 &   0.44 & 7.9 & 13.8 &   0.33 &  6.9 &13.4 &  0.25\\
240.25 & --1.75 &  23.90 &         & 2.20 &   0.180 &kout  & mab97 & 2.4 &  9.9 &   0.21 & 2.0 &  9.1 &   0.15 &  1.6 & 9.4 &  0.10\\
240.25 &  0.00 &  68.10 &         & 7.00 &   0.340 &kout  & mab97 & 7.3 & 13.7 &   0.37 & 6.3 & 12.4 &   0.28 &  5.5 &12.2 &  0.21\\
240.25 &  0.50 &  68.70 &         & 7.10 &   0.230 &kout  & mab97 & 7.4 & 13.7 &   0.25 & 6.4 & 12.5 &   0.19 &  5.6 &12.3 &  0.14\\
241.00 & --1.25 &  66.10 &         & 6.80 &   2.760 &kout  & mab97 & 7.0 & 13.4 &   2.92 & 6.1 & 12.2 &   2.22 &  5.3 &12.0 &  1.68\\
241.75 & --0.63 &  68.40 &         & 7.10 &   3.350 &kout  & mab97 & 7.3 & 13.6 &   3.54 & 6.4 & 12.4 &   2.72 &  5.6 &12.2 &  2.08\\
242.00 & --0.88 &  66.00 &         & 6.70 &   1.720 &kout  & mab97 & 7.0 & 13.3 &   1.88 & 6.1 & 12.1 &   1.43 &  5.3 &12.0 &  1.08\\
242.25 & --2.75 &  21.50 &         & 2.00 &   0.190 &kout  & mab97 & 2.2 &  9.7 &   0.23 & 1.8 &  9.0 &   0.15 &  1.4 & 9.3 &  0.09\\
242.25 & --2.75 &  26.20 &         & 2.40 &   0.320 &kout  & mab97 & 2.7 & 10.0 &   0.41 & 2.2 &  9.2 &   0.27 &  1.8 & 9.5 &  0.18\\
242.25 &  0.13 &  69.80 &         & 7.20 &   1.720 &kout  & mab97 & 7.5 & 13.7 &   1.87 & 6.5 & 12.5 &   1.40 &  5.7 &12.3 &  1.08\\
242.63 &  0.13 &  71.90 &         & 7.50 &   1.210 &kout  & mab97 & 7.8 & 13.9 &   1.31 & 6.8 & 12.7 &   0.99 &  5.9 &12.4 &  0.75\\
242.88 & --0.50 &  64.30 &         & 6.50 &   1.580 &kout  & mab97 & 6.8 & 13.1 &   1.73 & 5.9 & 11.9 &   1.30 &  5.2 &11.8 &  1.01\\
242.88 &  0.00 &  55.80 &         & 5.50 &   0.500 &kout  & mab97 & 5.7 & 12.2 &   0.54 & 5.0 & 11.2 &   0.41 &  4.4 &11.2 &  0.32\\
243.00 & --3.00 &  22.20 &         & 2.10 &   0.170 &kout  & mab97 & 2.3 &  9.8 &   0.20 & 1.9 &  9.0 &   0.14 &  1.5 & 9.3 &  0.09\\
243.25 & --0.75 &  66.90 &         & 6.90 &   0.760 &kout  & mab97 & 7.1 & 13.3 &   0.80 & 6.2 & 12.1 &   0.61 &  5.4 &12.0 &  0.47\\
244.25 & --2.50 &  70.40 &         & 7.30 &   0.130 &kout  & mab97 & 7.6 & 13.6 &   0.14 & 6.6 & 12.4 &   0.11 &  5.8 &12.2 &  0.08\\
\hline
244.50 & --2.00 &  92.60 &         &10.80 &   0.750 &kout  & mab97 &11.1 & 16.6 &   0.79 & 9.7 & 15.0 &   0.61 &  8.4 &14.3 &  0.45\\
244.75 & --0.25 &  62.50 &         & 6.30 &   0.190 &kout  & mab97 & 6.6 & 12.8 &   0.21 & 5.7 & 11.7 &   0.16 &  5.0 &11.6 &  0.12\\
246.75 & --0.50 &  21.40 &         & 2.10 &   0.620 &kout  & mab97 & 2.4 &  9.7 &   0.81 & 1.9 &  8.9 &   0.51 &  1.5 & 9.2 &  0.32\\
247.75 & --0.75 &  29.90 &         & 3.00 &   0.090 &kout  & mab97 & 3.2 & 10.2 &   0.10 & 2.7 &  9.4 &   0.07 &  2.3 & 9.6 &  0.05\\
248.00 &  0.50 &  27.00 &         & 2.70 &   0.230 &kout  & mab97 & 3.0 & 10.0 &   0.28 & 2.5 &  9.2 &   0.20 &  2.0 & 9.4 &  0.13\\
249.50 & --3.88 &  89.90 &         &10.20 &   0.480 &kout  & mab97 &10.6 & 15.7 &   0.52 & 9.3 & 14.2 &   0.40 &  8.2 &13.7 &  0.31\\
249.88 & --3.13 &  98.00 &         &11.60 &   1.500 &kout  & mab97 &12.0 & 16.9 &   1.61 &10.5 & 15.2 &   1.23 &  9.2 &14.5 &  0.94\\
250.38 & --1.00 &  66.40 &         & 7.00 &   0.200 &kout  & mab97 & 7.3 & 12.9 &   0.22 & 6.4 & 11.8 &   0.17 &  5.7 &11.7 &  0.13\\
250.63 & --3.63 &  89.60 &         &10.20 &   0.320 &kout  & mab97 &10.5 & 15.6 &   0.34 & 9.2 & 14.1 &   0.26 &  8.2 &13.6 &  0.21\\
251.25 & --2.00 &  54.80 &         & 5.70 &   0.260 &kout  & mab97 & 6.0 & 11.9 &   0.29 & 5.2 & 10.9 &   0.22 &  4.6 &10.9 &  0.17\\
252.00 & --3.50 &   8.50 &         & 1.00 &   0.093 &kout  & sod91 & 1.4 &  9.0 &   0.18 & 0.9 &  8.3 &   0.08 &  0.5 & 8.7 &  0.02\\
252.00 & --3.50 &  15.00 &         & 1.68 &   0.891 &kout  & sod91 & 2.0 &  9.3 &   1.26 & 1.6 &  8.6 &   0.81 &  1.1 & 8.9 &  0.38\\
252.08 & --4.24 &  95.14 &         &11.22 &   0.100 &kout  &  bw94 &11.5 & 16.3 &   0.11 &10.1 & 14.7 &   0.08 &  8.9 &14.1 &  0.06\\
253.00 & --2.93 &  99.60 &    0.91 &11.50 &   0.111 &kout  &nomf05 &12.3 & 16.8 &   0.13 &10.8 & 15.2 &   0.10 &  9.5 &14.5 &  0.08\\
253.00 & --1.50 &   5.90 &         & 0.74 &   0.263 &kout  & sod91 & 1.1 &  8.9 &   0.58 & 0.7 &  8.2 &   0.24 &  0.2 & 8.6 &  0.02\\
253.23 & --2.73 &  98.90 &    0.99 &11.30 &   0.121 &kout  &nomf05 &12.1 & 16.7 &   0.14 &10.7 & 15.1 &   0.11 &  9.4 &14.4 &  0.08\\
253.53 & --3.17 & 107.20 &    0.94 &12.80 &   0.114 &kout  &nomf05 &13.7 & 18.1 &   0.13 &12.1 & 16.3 &   0.10 & 10.6 &15.3 &  0.08\\
253.63 &  0.00 &  34.70 &         & 3.80 &   2.640 &kout  & mab97 & 4.1 & 10.4 &   3.07 & 3.5 &  9.6 &   2.24 &  3.0 & 9.8 &  1.65\\
253.75 &  0.00 &  34.60 &         & 3.80 &   1.690 &kout  & mab97 & 4.1 & 10.4 &   1.97 & 3.5 &  9.6 &   1.43 &  3.0 & 9.8 &  1.05\\
254.25 & --2.25 &  36.10 &         & 4.00 &   0.350 &kout  & mab97 & 4.3 & 10.5 &   0.40 & 3.6 &  9.6 &   0.28 &  3.2 & 9.8 &  0.22\\
254.25 & --1.00 &  76.60 &         & 8.50 &   0.330 &kout  & mab97 & 8.8 & 13.8 &   0.35 & 7.7 & 12.5 &   0.27 &  6.9 &12.3 &  0.22\\
255.25 & --1.00 &  80.10 &         & 9.00 &   1.180 &kout  & mab97 & 9.3 & 14.1 &   1.26 & 8.2 & 12.8 &   0.98 &  7.3 &12.6 &  0.78\\
255.38 & --0.75 &  86.00 &         & 9.90 &   1.440 &kout  & mab97 &10.1 & 14.8 &   1.50 & 8.9 & 13.4 &   1.16 &  8.0 &13.0 &  0.94\\
255.47 & --3.93 & 108.70 &    2.85 &13.20 &   0.348 &kout  &nomf05 &14.1 & 18.2 &   0.40 &12.4 & 16.4 &   0.31 & 10.9 &15.4 &  0.24\\
255.75 & --1.00 &  35.70 &         & 4.10 &   0.150 &kout  & mab97 & 4.3 & 10.4 &   0.16 & 3.7 &  9.6 &   0.12 &  3.2 & 9.8 &  0.09\\
256.00 & --2.50 &   9.80 &         & 1.31 &   1.738 &kout  & sod91 & 1.7 &  9.1 &   2.93 & 1.2 &  8.4 &   1.46 &  0.7 & 8.7 &  0.50\\
256.12 & --1.50 &  87.50 &         &10.10 &   0.950 &kout  & mab97 &10.4 & 14.9 &   1.01 & 9.2 & 13.5 &   0.79 &  8.2 &13.2 &  0.63\\
256.72 & --2.00 &  44.40 &         & 5.00 &   0.280 &kout  & mab97 & 5.3 & 11.0 &   0.31 & 4.6 & 10.1 &   0.24 &  4.1 &10.2 &  0.19\\
257.00 & --0.50 &  53.00 &         & 6.00 &   0.810 &kout  & mab97 & 6.2 & 11.6 &   0.86 & 5.4 & 10.6 &   0.66 &  4.8 &10.7 &  0.52\\
257.37 & --1.10 &  90.90 &    1.55 &10.30 &   0.190 &kout  &nomf05 &11.0 & 15.3 &   0.22 & 9.7 & 13.9 &   0.17 &  8.7 &13.4 &  0.14\\
257.50 & --2.38 &  93.10 &         &11.10 &   1.100 &kout  & mab97 &11.4 & 15.6 &   1.16 &10.0 & 14.1 &   0.89 &  9.0 &13.6 &  0.72\\
257.50 & --2.00 &  41.50 &         & 4.80 &   0.380 &kout  & mab97 & 5.1 & 10.8 &   0.43 & 4.4 &  9.9 &   0.32 &  3.9 &10.1 &  0.25\\
257.50 & --0.25 &  55.50 &         & 6.30 &   0.190 &kout  & mab97 & 6.5 & 11.8 &   0.20 & 5.7 & 10.8 &   0.16 &  5.1 &10.8 &  0.12\\
257.53 & --2.33 &  94.80 &    6.84 &10.90 &   0.836 &kout  &nomf05 &11.6 & 15.8 &   0.95 &10.3 & 14.3 &   0.75 &  9.2 &13.8 &  0.60\\
257.57 & --2.07 &  91.30 &    5.02 &10.40 &   0.613 &kout  &nomf05 &11.1 & 15.4 &   0.70 & 9.8 & 13.9 &   0.54 &  8.8 &13.4 &  0.44\\
257.83 & --2.63 &  91.10 &    4.57 &10.40 &   0.558 &kout  &nomf05 &11.1 & 15.3 &   0.64 & 9.8 & 13.9 &   0.50 &  8.8 &13.4 &  0.40\\
258.40 & --2.93 & 100.90 &    1.91 &11.90 &   0.233 &kout  &nomf05 &12.7 & 16.7 &   0.27 &11.2 & 15.0 &   0.21 & 10.0 &14.4 &  0.16\\
258.50 & --1.50 &  67.20 &         & 7.70 &   0.530 &kout  & mab97 & 7.9 & 12.7 &   0.56 & 7.0 & 11.6 &   0.44 &  6.3 &11.6 &  0.35\\
258.50 & --1.00 &  81.20 &         & 9.40 &   0.960 &kout  & mab97 & 9.7 & 14.1 &   1.02 & 8.6 & 12.8 &   0.80 &  7.7 &12.6 &  0.64\\
258.60 & --3.87 & 105.10 &    0.87 &12.70 &   0.107 &kout  &nomf05 &13.5 & 17.3 &   0.12 &11.9 & 15.6 &   0.09 & 10.6 &14.8 &  0.07\\
258.93 & --1.03 &  90.60 &    0.81 &10.40 &   0.099 &kout  &nomf05 &11.1 & 15.2 &   0.11 & 9.8 & 13.8 &   0.09 &  8.8 &13.3 &  0.07\\
259.00 & --1.75 &  58.50 &         & 6.70 &   1.290 &kout  & mab97 & 7.0 & 12.0 &   1.41 & 6.1 & 11.0 &   1.07 &  5.5 &11.0 &  0.87\\
259.25 & --2.50 &  54.50 &         & 6.30 &   1.160 &kout  & mab97 & 6.6 & 11.7 &   1.27 & 5.8 & 10.7 &   0.98 &  5.2 &10.8 &  0.79\\
259.37 & --1.90 &  95.80 &    1.28 &11.20 &   0.156 &kout  &nomf05 &11.9 & 15.9 &   0.18 &10.5 & 14.4 &   0.14 &  9.4 &13.8 &  0.11\\
259.75 & --1.25 &  62.90 &         & 7.30 &   0.840 &kout  & mab97 & 7.6 & 12.3 &   0.91 & 6.6 & 11.3 &   0.69 &  6.0 &11.2 &  0.57\\
259.83 & --1.57 &  94.20 &    1.00 &11.00 &   0.122 &kout  &nomf05 &11.7 & 15.6 &   0.14 &10.3 & 14.2 &   0.11 &  9.3 &13.6 &  0.09\\
260.00 &  0.00 &  60.70 &         & 7.10 &   0.680 &kout  & mab97 & 7.3 & 12.2 &   0.72 & 6.4 & 11.1 &   0.55 &  5.8 &11.1 &  0.45\\
260.07 & --1.53 &  91.70 &    2.05 &10.70 &   0.251 &kout  &nomf05 &11.3 & 15.3 &   0.28 &10.0 & 13.9 &   0.22 &  9.0 &13.4 &  0.18\\
260.50 & --0.75 &  57.20 &         & 6.70 &   0.860 &kout  & mab97 & 7.0 & 11.9 &   0.94 & 6.1 & 10.9 &   0.71 &  5.6 &10.9 &  0.60\\
260.63 &  0.38 &  55.30 &         & 6.50 &   0.950 &kout  & mab97 & 6.8 & 11.7 &   1.04 & 6.0 & 10.7 &   0.81 &  5.4 &10.8 &  0.66\\
260.75 & --2.50 &  64.10 &         & 7.50 &   0.390 &kout  & mab97 & 7.8 & 12.4 &   0.42 & 6.9 & 11.4 &   0.33 &  6.2 &11.3 &  0.27\\
261.00 & --0.75 &  70.00 &         & 8.30 &   1.510 &kout  & mab97 & 8.5 & 12.9 &   1.58 & 7.5 & 11.8 &   1.23 &  6.8 &11.7 &  1.01\\
261.00 & --2.50 &  13.70 &         & 2.10 &   2.951 &kout  & sod91 & 2.5 &  9.2 &   4.18 & 2.0 &  8.5 &   2.68 &  1.4 & 8.8 &  1.31\\
261.27 & --5.03 & 102.20 &    2.03 &12.40 &   0.248 &kout  &nomf05 &13.2 & 16.8 &   0.28 &11.7 & 15.1 &   0.22 & 10.4 &14.4 &  0.17\\
261.40 & --2.00 & 102.90 &    1.94 &12.50 &   0.237 &kout  &nomf05 &13.3 & 16.8 &   0.27 &11.8 & 15.2 &   0.21 & 10.5 &14.5 &  0.17\\
261.50 & --1.25 &  55.40 &         & 6.60 &   0.220 &kout  & mab97 & 6.9 & 11.7 &   0.24 & 6.1 & 10.7 &   0.19 &  5.5 &10.8 &  0.15\\
261.80 & --3.03 & 103.00 &    1.20 &12.50 &   0.147 &kout  &nomf05 &13.4 & 16.8 &   0.17 &11.8 & 15.2 &   0.13 & 10.6 &14.5 &  0.11\\
261.83 & --1.47 & 101.40 &    1.12 &12.30 &   0.137 &kout  &nomf05 &13.1 & 16.6 &   0.16 &11.6 & 15.0 &   0.12 & 10.3 &14.3 &  0.10\\
262.17 & --2.20 &  98.60 &    0.83 &11.90 &   0.102 &kout  &nomf05 &12.6 & 16.2 &   0.11 &11.2 & 14.6 &   0.09 & 10.0 &14.0 &  0.07\\
262.25 & --2.00 &  59.70 &         & 7.20 &   0.780 &kout  & mab97 & 7.5 & 12.0 &   0.85 & 6.6 & 11.0 &   0.66 &  6.0 &11.0 &  0.54\\
262.75 & --0.50 &  59.90 &         & 7.30 &   0.490 &kout  & mab97 & 7.5 & 12.1 &   0.52 & 6.6 & 11.0 &   0.40 &  6.0 &11.0 &  0.33\\
263.00 & --2.50 &  62.10 &         & 7.50 &   0.230 &kout  & mab97 & 7.8 & 12.2 &   0.25 & 6.9 & 11.2 &   0.19 &  6.3 &11.2 &  0.16\\
263.60 & --4.00 & 111.00 &    2.51 &14.10 &   0.307 &kout  &nomf05 &15.1 & 18.1 &   0.35 &13.3 & 16.3 &   0.27 & 11.8 &15.3 &  0.22\\
264.00 & --1.50 &  29.40 &         & 4.20 &   0.800 &kout  & mab97 & 4.5 & 10.0 &   0.92 & 3.9 &  9.2 &   0.69 &  3.4 & 9.5 &  0.52\\
\hline
265.00 & --1.50 &  29.40 &         & 4.30 &   0.480 &kout  & mab97 & 4.6 & 10.0 &   0.55 & 4.0 &  9.2 &   0.42 &  3.5 & 9.5 &  0.32\\
265.00 &  0.00 &  47.10 &         & 6.20 &   0.150 &kout  & mab97 & 6.4 & 11.1 &   0.16 & 5.6 & 10.2 &   0.12 &  5.1 &10.3 &  0.10\\
265.25 & --1.25 &  65.90 &         & 8.30 &   0.360 &kout  & mab97 & 8.5 & 12.5 &   0.38 & 7.5 & 11.4 &   0.29 &  6.9 &11.4 &  0.25\\
265.50 & --1.50 &  27.00 &         & 4.10 &   0.150 &kout  & mab97 & 4.4 &  9.9 &   0.17 & 3.8 &  9.1 &   0.13 &  3.3 & 9.4 &  0.10\\
265.53 & --2.00 &  90.80 &    4.20 &11.10 &   0.513 &kout  &nomf05 &11.8 & 15.1 &   0.58 &10.5 & 13.6 &   0.46 &  9.5 &13.2 &  0.38\\
266.50 & --3.00 &  65.50 &         & 8.30 &   0.340 &kout  & mab97 & 8.6 & 12.5 &   0.37 & 7.7 & 11.4 &   0.29 &  7.0 &11.4 &  0.24\\
266.50 & --2.00 &  24.20 &         & 4.00 &   0.110 &kout  & mab97 & 4.3 &  9.7 &   0.13 & 3.6 &  9.0 &   0.09 &  3.2 & 9.2 &  0.07\\
266.75 & --2.00 &  57.60 &         & 7.50 &   0.280 &kout  & mab97 & 7.8 & 11.8 &   0.30 & 6.9 & 10.8 &   0.24 &  6.3 &10.9 &  0.20\\
266.75 & --0.75 &  31.20 &         & 4.70 &   0.180 &kout  & mab97 & 5.0 & 10.1 &   0.20 & 4.4 &  9.3 &   0.16 &  3.9 & 9.5 &  0.12\\
267.00 & --0.75 &  19.20 &         & 3.50 &   0.320 &kout  & mab97 & 3.8 &  9.5 &   0.38 & 3.2 &  8.8 &   0.27 &  2.7 & 9.0 &  0.19\\
267.25 & --2.00 &  18.10 &         & 3.40 &   0.160 &kout  & mab97 & 3.7 &  9.4 &   0.19 & 3.1 &  8.7 &   0.13 &  2.6 & 9.0 &  0.09\\
268.50 & --2.50 &  62.60 &         & 8.30 &   1.120 &kout  & mab97 & 8.6 & 12.2 &   1.20 & 7.6 & 11.2 &   0.94 &  7.0 &11.2 &  0.80\\
269.00 & --1.25 &  62.10 &         & 8.30 &   0.720 &kout  & mab97 & 8.6 & 12.2 &   0.77 & 7.6 & 11.1 &   0.60 &  7.0 &11.1 &  0.51\\
269.00 & --0.50 &  42.50 &         & 6.20 &   1.070 &kout  & mab97 & 6.5 & 10.8 &   1.18 & 5.7 &  9.9 &   0.90 &  5.2 &10.1 &  0.75\\
269.00 & --0.50 &  46.60 &         & 6.70 &   0.480 &kout  & mab97 & 6.9 & 11.1 &   0.51 & 6.1 & 10.1 &   0.40 &  5.6 &10.3 &  0.34\\
269.25 & --1.75 &  79.10 &         &10.40 &   0.910 &kout  & mab97 &10.7 & 13.7 &   0.96 & 9.5 & 12.5 &   0.76 &  8.7 &12.3 &  0.64\\
269.88 & --0.13 &  74.90 &         &10.00 &   1.870 &kout  & mab97 &10.2 & 13.3 &   1.95 & 9.1 & 12.1 &   1.55 &  8.4 &11.9 &  1.32\\
270.90 & --0.50 &  52.00 &         & 6.80 &   2.200 &kout  &gcbt88 & 7.8 & 11.4 &   2.89 & 6.9 & 10.5 &   2.27 &  6.4 &10.5 &  1.95\\
279.90 & --1.60 &  35.00 &         & 7.10 &   1.000 &kout  &gcbt88 & 7.6 & 10.4 &   1.15 & 6.8 &  9.5 &   0.92 &  6.4 & 9.7 &  0.81\\
281.40 & --1.10 &  --5.00 &         & 3.20 &  22.500 &stel  &gcbt88 &     &      &  22.50 &     &      &  22.50 &      &     & 22.50\\
282.00 & --0.80 &  17.00 &         & 6.10 &   2.700 &kout  &gcbt88 & 6.2 &  9.4 &   2.79 & 5.4 &  8.7 &   2.12 &  5.1 & 8.9 &  1.89\\
282.90 &  1.30 & --19.00 &         & 2.20 &   0.700 &ktan  &gcbt88 & 1.9 &  8.3 &   0.52 & 1.8 &  7.8 &   0.47 &  1.9 & 8.2 &  0.52\\
282.90 & --0.70 &  --5.00 &         & 3.20 &   2.800 &kfar  &gcbt88 & 3.4 &  8.4 &   3.16 & 2.3 &  7.8 &   1.45 &  1.9 & 8.3 &  0.99\\
283.80 &  0.00 &  --5.00 &         & 4.00 &   6.300 &stel  &gcbt88 &     &      &   6.30 &     &      &   6.30 &      &     &  6.30\\
284.50 & --0.20 &  12.00 &         & 6.60 &  20.300 &kout  &gcbt88 & 6.2 &  9.2 &  17.91 & 5.4 &  8.5 &  13.59 &  5.1 & 8.8 & 12.12\\
285.30 &  0.00 &   0.00 &         & 5.30 &   4.500 &kfar  &gcbt88 & 4.9 &  8.6 &   3.85 & 4.2 &  8.0 &   2.83 &  3.7 & 8.3 &  2.19\\
286.00 &  0.00 & --20.20 &         & 2.34 &  19.498 &kout  & sod91 &     &      &        &     &      &        &  2.3 & 8.2 & 18.84\\
286.40 & --0.30 &  14.00 &         & 7.50 &   8.300 &kout  &gcbt88 & 6.8 &  9.3 &   6.82 & 6.1 &  8.6 &   5.49 &  5.8 & 8.8 &  4.96\\
287.50 & --0.50 & --19.00 &         & 2.70 &   6.700 &stel  &gcbt88 &     &      &   6.70 &     &      &   6.70 &      &     &  6.70\\
287.53 &  0.97 &  --4.20 &    3.34 & 2.60 &   0.117 &stel  &dkm+08 &     &      &   0.12 &     &      &   0.12 &      &     &  0.12\\
287.60 &  8.00 & --27.80 &    3.29 & 2.60 &   0.115 &stel  &dkm+08 &     &      &   0.12 &     &      &   0.12 &      &     &  0.12\\
288.60 &  1.50 & --22.00 &         & 3.20 &   3.500 &ktan  &gcbt88 & 2.7 &  8.1 &   2.49 & 2.6 &  7.6 &   2.31 &  2.7 & 8.0 &  2.49\\
288.80 &  1.07 & --16.90 &   56.05 & 2.60 &   1.960 &stel  &dkm+08 &     &      &   1.96 &     &      &   1.96 &      &     &  1.96\\
289.00 & --0.50 &  20.20 &         & 7.67 &  85.114 &kout  & sod91 & 8.0 &  9.6 &  92.60 & 7.2 &  8.9 &  75.00 &  7.0 & 9.1 & 70.89\\
289.30 & --0.60 &  22.00 &         & 7.90 &  31.200 &stel  &gcbt88 &     &      &  31.20 &     &      &  31.20 &      &     & 31.20\\
290.20 & --0.20 &  --1.00 &         & 6.80 &   2.700 &kfar  &gcbt88 & 6.1 &  8.6 &   2.17 & 5.4 &  8.0 &   1.70 &  5.1 & 8.3 &  1.52\\
290.60 & --0.20 &  15.00 &         & 8.70 &   4.300 &kout  &gcbt88 & 7.9 &  9.3 &   3.55 & 7.1 &  8.6 &   2.86 &  6.9 & 8.9 &  2.70\\
291.50 & --1.50 & --25.40 &         & 3.12 &  36.308 &kout  & sod91 &     &      &        &     &      &        &  3.1 & 7.9 & 35.84\\
291.60 & --0.40 &  15.00 &         & 7.20 &   4.100 &stel  &gcbt88 &     &      &   4.10 &     &      &   4.10 &      &     &  4.10\\
291.60 & --0.50 &  29.00 &         &10.30 &  12.500 &kout  &gcbt88 & 9.4 & 10.1 &  10.41 & 8.6 &  9.3 &   8.71 &  8.4 & 9.5 &  8.31\\
292.60 & --0.30 &   4.00 &         & 8.10 &  12.000 &kfar  &gcbt88 & 7.3 &  8.8 &   9.75 & 6.5 &  8.2 &   7.73 &  6.4 & 8.4 &  7.49\\
293.30 & --1.40 & --25.00 &         & 2.40 &   7.100 &stel  &gcbt88 &     &      &   7.10 &     &      &   7.10 &      &     &  7.10\\
294.00 & --0.90 &  32.00 &         &11.30 &  12.200 &kout  &gcbt88 &10.3 & 10.3 &  10.14 & 9.3 &  9.5 &   8.26 &  9.2 & 9.7 &  8.09\\
294.40 & --0.70 &  --3.00 &         & 8.00 &  12.200 &kfar  &gcbt88 & 7.0 &  8.5 &   9.34 & 6.3 &  7.9 &   7.57 &  6.2 & 8.2 &  7.33\\
295.10 & --0.80 &  26.00 &         &11.00 &   4.500 &kout  &gcbt88 &10.0 & 10.0 &   3.72 & 9.1 &  9.2 &   3.08 &  9.0 & 9.4 &  3.01\\
297.40 & --0.50 &  22.00 &         &11.20 &  32.100 &kout  &gcbt88 &10.1 &  9.8 &  26.10 & 9.2 &  9.0 &  21.66 &  9.2 & 9.2 & 21.66\\
297.50 & --0.50 &  25.40 &         &10.14 & 218.776 &kout  & sod91 &10.5 & 10.0 & 234.59 & 9.5 &  9.2 & 192.03 &  9.5 & 9.4 &192.03\\
298.60 &  0.10 & --35.00 &         & 4.70 &  26.500 &ktan  &gcbt88 & 4.1 &  7.5 &  20.17 & 3.8 &  7.0 &  17.32 &  4.0 & 7.4 & 19.19\\
298.80 & --0.20 &  25.00 &         &12.00 &  30.800 &kout  &gcbt88 &10.7 & 10.0 &  24.49 & 9.8 &  9.2 &  20.54 &  9.8 & 9.4 & 20.54\\
299.40 & --0.10 &  --6.00 &         & 9.40 &   4.600 &kfar  &gcbt88 & 8.0 &  8.4 &   3.33 & 7.3 &  7.8 &   2.77 &  7.3 & 8.1 &  2.77\\
299.50 & --0.50 & --39.70 &         & 4.19 &  57.544 &kout  & sod91 &     &      &        &     &      &        &  4.1 & 7.4 & 55.10\\
300.30 & --0.20 &  31.00 &         &12.90 &   7.000 &kout  &gcbt88 &11.6 & 10.4 &   5.66 &10.6 &  9.6 &   4.73 & 10.6 & 9.7 &  4.73\\
300.60 & --0.20 &  10.00 &         &11.10 &   4.200 &kout  &gcbt88 & 9.8 &  9.1 &   3.27 & 9.0 &  8.4 &   2.76 &  9.0 & 8.7 &  2.76\\
301.80 &  0.00 &  24.00 &         &12.70 &   4.200 &kout  &gcbt88 &11.3 & 10.0 &   3.33 &10.4 &  9.2 &   2.82 & 10.4 & 9.3 &  2.82\\
302.30 & --0.70 &  32.00 &         &13.60 &  29.400 &kout  &gcbt88 &12.2 & 10.5 &  23.66 &11.2 &  9.7 &  19.94 & 11.2 & 9.8 & 19.94\\
303.90 & --0.40 &  29.00 &         &13.70 &  46.200 &kout  &gcbt88 &12.3 & 10.3 &  37.24 &11.3 &  9.5 &  31.43 & 11.3 & 9.6 & 31.43\\
306.90 & --0.50 &  25.00 &         &14.20 &   5.600 &kout  &gcbt88 &12.6 & 10.1 &   4.41 &11.6 &  9.3 &   3.74 & 11.7 & 9.5 &  3.80\\
308.00 & --0.70 &  32.00 &         &15.10 &  21.000 &kout  &gcbt88 &13.5 & 10.7 &  16.79 &12.4 &  9.8 &  14.16 & 12.5 & 9.9 & 14.39\\
\hline
311.30 & --0.30 &  27.00 &         &15.50 & 109.200 &kout  &gcbt88 &13.8 & 10.4 &  86.56 &12.7 &  9.6 &  73.31 & 12.8 & 9.6 & 74.47\\
313.20 & --0.30 &  42.00 &         &17.60 &  18.200 &kout  &gcbt88 &15.9 & 11.8 &  14.85 &14.6 & 10.8 &  12.52 & 14.6 &10.7 & 12.52\\
314.00 & --0.10 &  28.00 &         &16.40 &   4.200 &kout  &gcbt88 &14.5 & 10.6 &   3.28 &13.4 &  9.7 &   2.80 & 13.5 & 9.8 &  2.85\\
315.30 & --0.30 &  14.00 &         &15.40 &  11.200 &kout  &gcbt88 &13.5 &  9.5 &   8.61 &12.4 &  8.8 &   7.26 & 12.7 & 8.9 &  7.62\\
318.00 & --0.30 &  30.00 &         &17.60 &   2.800 &kout  &gcbt88 &15.7 & 10.9 &   2.23 &14.4 & 10.0 &   1.87 & 14.6 &10.0 &  1.93\\
320.50 & --0.50 &  26.00 &         &17.80 &  12.600 &kout  &gcbt88 &15.8 & 10.7 &   9.93 &14.6 &  9.8 &   8.48 & 14.8 & 9.8 &  8.71\\
325.30 & --0.10 &  28.00 &         &19.30 &  19.600 &kout  &gcbt88 &17.1 & 11.2 &  15.39 &15.8 & 10.3 &  13.14 & 16.0 &10.2 & 13.47\\
328.50 & --0.10 &  30.00 &         &20.40 &   5.600 &kout  &gcbt88 &18.2 & 11.8 &   4.46 &16.8 & 10.8 &   3.80 & 17.0 &10.7 &  3.89\\
335.50 & --0.50 &  27.00 &         &22.00 &   0.600 &kout  &gcbt88 &19.7 & 12.4 &   0.48 &18.1 & 11.4 &   0.41 & 18.3 &11.1 &  0.42\\
\hline
\end{tabular}
\medskip
References: bw94: \citet{bw94}; css90: \citet{css90}; dect86: \citet{dect86}; dbt90: \citet{dbt90}; dgt94: \citet{dgt94}; dlp+96: \citet{dlp+96}; dkm+08: \citet{dkm+08}; gcbt88: \citet{gcbt88}; hcs01:
\citet{hcs01}; mk88: \citet{mk88}; mab97: \citet{mab97}; nomf05: \citet{nomf05}; sod91: \citet{sod91}; srby87: \citet{srby87}; tllw07: \citet{tllw07}.
\end{table}